

Engaging Students in Sustainable Science Education

Edited by Larry J. Grabau Printed Edition of the Special Issue Published in *Sustainability*



www.mdpi.com/journal/sustainability

Engaging Students in Sustainable Science Education

Engaging Students in Sustainable Science Education

Editor

Larry J. Grabau

MDPI • Basel • Beijing • Wuhan • Barcelona • Belgrade • Manchester • Tokyo • Cluj • Tianjin



Editor Larry J. Grabau University of Kentucky USA

Editorial Office MDPI St. Alban-Anlage 66 4052 Basel, Switzerland

This is a reprint of articles from the Special Issue published online in the open access journal *Sustainability* (ISSN 2071-1050) (available at: https://www.mdpi.com/journal/sustainability/ special_issues/Engaging_Students_Sustainable_Education).

For citation purposes, cite each article independently as indicated on the article page online and as indicated below:

LastName, A.A.; LastName, B.B.; LastName, C.C. Article Title. *Journal Name* Year, *Volume Number*, Page Range.

ISBN 978-3-0365-5191-3 (Hbk) ISBN 978-3-0365-5192-0 (PDF)

© 2022 by the authors. Articles in this book are Open Access and distributed under the Creative Commons Attribution (CC BY) license, which allows users to download, copy and build upon published articles, as long as the author and publisher are properly credited, which ensures maximum dissemination and a wider impact of our publications.

The book as a whole is distributed by MDPI under the terms and conditions of the Creative Commons license CC BY-NC-ND.

Contents

About the Editor
Larry J. Grabau Engaging Students in Sustainable Science Education Reprinted from: Sustainability 2022, 14, 6127, doi:10.3390/su14106127 1
Chih-Hao Lin and Yao-Yun ChangA Progressive Three-Stage Teaching Method Using Interactive Classroom Activities to ImproveLearning Motivation in Computer Networking CoursesReprinted from: Sustainability 2022, 14, 5191, doi:10.3390/su140951913
Josep Petchamé, Ignasi Iriondo, Xavi Canaleta, David Riu and Silvia NecchiEngaging ICT Engineering Undergraduates in a Management Subject through First Day of ClassActivities: An Empirical StudyReprinted from: Sustainability 2021, 13, 7440, doi:10.3390/su1313744021
Ruba Ali, Jolly Bhadra, Nitha Siby, Zubair Ahmad and Noora Jabor Al-Thani A STEM Model to Engage Students in Sustainable Science Education through Sports: A Case Study in Qatar Reprinted from: <i>Sustainability</i> 2021 , <i>13</i> , 3483, doi:10.3390/su13063483
 Larry J. Grabau, Jari Lavonen and Kalle Juuti Finland, A Package Deal: Disciplinary Climate in Science Classes, Science Dispositions and Science Literacy Reprinted from: <i>Sustainability</i> 2021, <i>13</i>, 13857, doi:10.3390/su132413857
Kanokwan Kiwfo, Chonnipa Yeerum, Piyanat Issarangkura Na Ayutthaya,Kullapon Kesonkan, Siripat Suteerapataranon, Piyatida Panitsupakamol,Dujrudee Chinwong, Pathinan Paengnakorn, Surarong Chinwong,Narong Kotchabhakdi, Chalermpong Saenjum, Monnapat Vongboot and Kate GrudpanSustainable Education with Local-Wisdom Based Natural Reagent for Green Chemical Analysiswith a Smart Device: Experiences in ThailandReprinted from: Sustainability 2021, 13, 11147, doi:10.3390/su13201114783
Chonnipa Yeerum, Piyanat Issarangkura Na Ayutthaya, Kullapon Kesonkan, Kanokwan Kiwfo, Siripat Suteerapataranon, Piyatida Panitsupakamol, Pathinan Paengnakorn, Dujrudee Chinwong, Surarong Chinwong, Chalermpong Saenjum, Monnapat Vongboot and Kate Grudpan Lab-at-Home: Hands-On Green Analytical Chemistry Laboratory New Normal Experimentation Reprinted from: <i>Sustainability</i> 2022, <i>14</i> , 3314, doi:10.3390/su14063314
Nieves Aquino-Llinares and Pilar Moreno-Navarro Engaging Environmental Sciences Students in Statistics through an Inclusive Experience in a Spanish University Reprinted from: <i>Sustainability</i> 2022 , <i>14</i> , 8352, doi:10.3390/su14148352
Hanns de la Fuente-Mella, Ricardo Campos-Espinoza, Nelson Lay-Raby, Omar Lamelés-Corvalán, Mario Pino-Moya and Reynier Ramírez-Molina Multinomial Cross-Sectional Regression Models to Estimate and Predict the Determinants of Academic Performance: The Case of Auditor Accountant of the Pontifical Catholic University of Valparaíso Reprinted from: <i>Sustainability</i> 2022, <i>14</i> , 9232, doi:10.3390/su14159232

About the Editor

Larry J. Grabau

Larry J. Grabau is a Professor of Plant and Soil Sciences. Larry's career has revolved around scientific instruction in classrooms, laboratories, in the field, and through studies abroad. He recently earned a second doctorate in educational psychology, and has a new focus on scientific education with emphasis on sustainability for diverse students, instructors and institutions.





Engaging Students in Sustainable Science Education

Larry J. Grabau

Department of Plant and Soil Science, University of Kentucky, Lexington, KY 40546, USA; larry.grabau@uky.edu

Students who find science to be enjoyable are more likely to engage in their science classes [1], and perform at higher levels on science exams [2]. Such students may also have higher levels of interest in science-related careers [3] and become informed citizens on science-related issues [4]. Rather than simply accepting that some students (in primary, secondary, and tertiary education) are naturally intrigued by science, some science practitioners and educators are now devising approaches to science concepts intended to stimulate such enjoyment and engagement [5]. Teaching support, hands-on activities, nature-of-science frameworks, and self-paced (yet mentored) online instruction are just a few of these approaches. In addition, educators are finding that some students feel excluded from science by virtue of their gender, ethnic or racial background, or socio-economic standing [6]; those educators are actively seeking ways to help such students find their identity in science fields (for example, by addressing science issues especially relevant to such possibly excluded students). Science need not be taught as a set of disjointed facts; instead, compelling science narratives can be invoked, drawing students into the story (and study) of science [7].

Six articles (see list of contributions below) comprise this Special Issue. Four articles focus on sustainable, engaging classrooms, emphasizing teachers' development of environments conducive to students' science learning. Meanwhile, the remaining pair of articles present sustainable, engaging classrooms, emphasizing sustainability of materials utilized for teaching and learning—creating a positive feedback loop of both instructional style and practical choices supporting sustainability in the learning context.

Four articles concentrate on teachers' management of science learning environments: Lin and Chang, Petchamé et al., Ali et al., and Grabau et al. Lin and Chang's research considered the possibility that Generation Z students may well have distinct learning preferences, and thus respond to re-considered teaching and learning environments. Their work involved the use of a progressive set of instructional approaches, moving first through a traditional lecture, then to flipped classroom methods, then to well-designed teamwork projects. Student participants were supportive of this novel approach to classroom instruction. Petchamé et al. faced a unique difficulty-engineering students, enrolled in a business management course, faced a reasonable lack of initial engagement in a course that would naturally be outside their skill set and perhaps even outside their interest. These authors developed and tested a series of interactive activities (for example, reciprocal interviews) for the first two days of class, with the goal of enhanced early engagement as well as full-semester learning. Ali et al. invoked a STEM learning model to assist secondary students to grasp concepts in materials science. Their unique approach focused on sports applications of materials; their work included an assessment of students' resultant attitudes toward STEM fields. Grabau et al. considered the relationships between teaching climates in science classroom, finding that disciplinary climate (characterized by a "get-down-tobusiness" science teaching and learning model) to benefit not only student dispositions toward science but also the students' performance on standardized exams.

Meanwhile, the remaining pair of articles (Kiwfo et al. and Yeerum et al.) helped students contextualize their learning about sustainability in chemistry classrooms by seeking out "green" materials to use, particularly during the season of the COVID-19 pandemic.

Citation: Grabau, L.J. Engaging Students in Sustainable Science Education. *Sustainability* 2022, 14, 6127. https://doi.org/10.3390/ su14106127

Received: 11 May 2022 Accepted: 16 May 2022 Published: 18 May 2022

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2022 by the author. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/).

1

While both of these articles' teaching/research practice took place in Thailand, the approaches were somewhat different. Kiwfo et al. used locally available guava leaves to assay for iron in the classroom context—data quality was good, reagent expense was markedly reduced and students were challenged, by this very experience, to learn to think of "doing science" in more sustainable ways. Yeerum et al., faced with stay-at-home orders for many of their students, devised a mail-out kit to help their confined students to learn skills in analytical chemistry. The at-home work of students was complemented by a subsequent synchronous online learning session.

Taken together, the articles comprising this Special Issue support student learning in science-related fields through enhanced and innovative instructional approaches, along with creative techniques to help students learn difficult subjects even when facing resource constraints.

Funding: This research received no external funding.

Conflicts of Interest: The author declares no conflict of interest.

References

- Lau, K.-C.; Ho, S.-C.E. Attitudes towards science, teaching practices, and science performance in PISA 2015: Multilevel analysis
 of the Chinese and Western top performers. *Res. Sci. Educ.* 2020, 52, 415–426. [CrossRef]
- Long, H.; Gao, S.; Yang, L.; Chen, J. Do teaching practices and enjoyment of science matter to science achievement? *Psychol. Schs.* 2022, 59, 334–355. [CrossRef]
- Wang, H.-H.; Lin, H.-S.; Chen, Y.-C.; Pan, Y.-T.; Hong, Z.-R. Modelling relationships among students' inquiry-related learning activities, enjoyment of learning, and their intended choice of a future STEM career. Int. J. Sci. Educ. 2021, 43, 157–178. [CrossRef]
- 4. Lin, H.-S.; Lawrenz, F.; Lin, S.-F.; Hong, Z.-R. Relationships among affective factors and preferred engagement in science-related activities. *Public Underst. Sci.* 2012, 22, 941–954. [CrossRef] [PubMed]
- 5. Hampden-Thompson, G.; Bennett, J. Science teaching and learning activities and students' engagement in science. *Int. J. Sci. Educ.* 2013, *35*, 1325–1343. [CrossRef]
- Bøe, M.V.; Henriksen, E.K. Love it or leave it: Norwegian students' motivations and expectations for postcompulsory physics. Sci. Educ. 2013, 97, 550–573. [CrossRef]
- Williams, C.T.; Rudge, D.W. Effects of historical story telling on student understanding of the nature of science. Sci. Educ. 2019, 28, 1105–1133. [CrossRef]



Article



A Progressive Three-Stage Teaching Method Using Interactive Classroom Activities to Improve Learning Motivation in Computer Networking Courses

Chih-Hao Lin * and Yao-Yun Chang

Department of Information Management, Chung Yuan Christian University, Taoyuan 320314, Taiwan; s10840129@cycu.edu.tw

* Correspondence: linch@cycu.edu.tw; Tel.: +886-3-265-5410

Abstract: Generation Z students have their learning preferences. They like to learn independently, advocate for what they believe in, and work hard to achieve their goals. However, there are significant gaps between Generation Z students' expectations for learning and prior experiences, especially for three domains of motivation in online learning environments: relatability, affirmation, and opportunity. This study aims at exploring the effectiveness of a progressive teaching method designed for Generation Z students in computer networking courses. This study proposes a progressive three-stage teaching method that gradually implements traditional lecture, individual flipped learning, and cooperative flipped learning methods over a semester. The design principle of this study differs from most existing studies that focus on the effectiveness of specific teaching methods. This study encourages each student to learn sequentially through three teaching stages. The purpose of this study is to investigate the changes in students' learning experiences, particularly in terms of learning comprehension and learning motivation. The research results show that the proposed progressive teaching method can improve students' understanding of computer networking courses and enhance their learning motivation. Participants agreed that the proposed progressive pedagogy can improve their teamwork skills and provide a different learning experience in the computer networking courses.

Keywords: progressive pedagogy; three teaching stages; individual learning; cooperative learning; mixed-method study

1. Introduction

The current generation of college students, born between 1995 and 2010, is known as Generation Z (Gen-Z) [1]. They enjoy consuming information from a variety of digital sources to develop their learning plans [2]. Being tech-savvy also helps Gen-Z students become comfortable with learning independently, where they can set their own pace and practice what they are learning in a personal environment [3]. That does not mean that Gen-Z students give up on collaborative group work with classmates. Instead, they are most motivated by relationships. Gen-Z students enjoy using video sharing, texting, and engaging with social media platforms to build relationships. They cherish the opportunity to interact with their peers and teachers and prefer to collaborate with others after their individual learning [3]. As a result, Gen-Z students are not only self-learners, they also prefer self-paced learning, which means that flipped learning (FL) would work well for them [4].

The traditional lecture (TL) teaching method is a form of teacher-centered pedagogy, which only provides a limited learning environment for students to practice textbook examples. Just as Gen-Z students are motivated by engaging with their identified passions, failure to engage and be recognized in the learning process can also be a lack of motivation for this generation [5]. They like to complete self-study before being asked to do it in class

Citation: Lin, C.-H.; Chang, Y.-Y. A Progressive Three-Stage Teaching Method Using Interactive Classroom Activities to Improve Learning Motivation in Computer Networking Courses. *Sustainability* **2022**, *14*, 5191. https://doi.org/10.3390/su14095191

Academic Editor: Larry J. Grabau

Received: 31 March 2022 Accepted: 22 April 2022 Published: 25 April 2022

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/).

3

or work with a team, so Gen-Z students can be motivated through experiential learning and teamwork [6].

This study proposes a progressive three-stage teaching approach that guides students to participate in a process ranging from (1) traditional lecture (TL) to (2) individual flipped learning (IFL) to (3) cooperative flipped learning (CFL) methods within a semester. According to the surface, shallow, and deep knowledge levels of teaching content, the proposed progressive three-stage teaching method contains the following: (1) the TL stage serves as the basis for knowledge memory, (2) the IFL stage is used to create a self-learning environment, and (3) the CFL stage is adopted to carry out collaborative group work with peers.

The design principle of this study differs from those of most studies that use groups of students to compare the effectiveness of a particular teaching method with traditional ones. The progressive pedagogy encourages each student to learn through the three teaching stages in sequence and collect his/her subjective feelings about the differences in the learning experience of each stage. This research is based on the belief that knowledge is constructed in an ongoing manner by learners as they engage and give meaning to an activity, experience, or interaction. This research focuses on the acquisition of meaning through students' subjective experiences and perspectives in the progressive teaching environment. The purpose of this study is to investigate the changes in students' learning experiences, especially in terms of learning comprehension and learning motivation, after they sequentially participate in three different teaching stages. This research wants to find a teaching process that allows students to explore their own learning characteristics and find a learning style that is suitable for their lifelong learning.

The remaining parts are organized as follows. Section 2 presents the learning characteristics of Generation Z and the background of different learning methods. Section 3 describes the research questions, participants, research methods, instructional design, progressive teaching approach, and research procedure. Section 4 analyzes the questionnaire data, presents the results of student feedback, and summarizes findings on student engagement. Section 5 concludes the work and summarizes the research contributions.

2. Literature Review

In 2014, the marketing firm Sparks & Honey released a report promoting the name Generation Z [1]. Although many studies have slightly different ranges of birth years for Gen-Z, to make the findings of this study consistent with most studies, the birth years of Gen-Z participants fell within the range of 1995 to 2010. Exposed to digital devices from an early age, Gen-Z has access to more information than any other generation, resulting in different learning preferences such as instant inquiry, video learning, and learning by doing [3]. To match the tech-savvy characteristics of Gen-Z students, information and communication technologies (ICT) should be widely incorporated into learning environments to provide Gen-Z students with modular learning content and diversified knowledge presentations [7].

2.1. Motivation of Generation Z Learners

Research findings by Dusseau [8] show that participants have significant gaps between desired connections and previous experiences for three domains of motivation in online learning environments: relatability, affirmation, and opportunity. That is, while Gen-Z is accustomed to individual online learning, the isolation of online learning can hinder the motivation of Gen-Z students. Therefore, the way teachers establish face-toface interaction and communication with Gen-Z learners will initiate this motivational improvement process.

First, face-to-face settings are great as a motivator to increase the motivation of relatability. Gen-Z students find it easier to engage with teachers through small talk in FL-based teaching methods. Students typically adopt formal and structured behaviors during online learning, and more casual and immediate behaviors in face-to-face interactions [8]. Therefore, well-defined interactive classroom activities will motivate students to learn.

Lo and Hew [9] recommended that teachers should establish an online learning platform to provide an individual learning environment with self-evaluation mechanisms and learning-by-doing activities. FL is a means to create an interactive teaching environment under time constraints [10]. Classroom time was reserved for students to engage in experiential learning activities in teams [11]. Therefore, this study establishes an FL environment by creating an online learning platform, including instructional videos and learning materials, to help students preview the teaching content by themselves.

Second, affirmation of ability, whether of internal or external origin, acts as a major motivator. It is used to increase their motivation to learn by affirming their own potential [8]. Such motivation is a visualization of the goals in front of them, so experiential learning in FL must allow students to see what is coming next and affirm who they will become. Through peer review, professional production, and academic achievement, Gen-Z can gain the voice and feedback of their peers.

According to Foldnes [12], one of the main variables affecting FL success is the way classroom activities are organized. Teachers take advantage of online learning platforms to provide students with the means to learn by themselves or remotely before classes. Then, teachers can use classroom time to carry out experiential learning and problem-solving activities [9,13]. Compared to TL, FL is more flexible, allowing for the creation of more diverse learning environments and a wider range of teaching materials [14].

Third, the most interesting part of the opportunistic motivation is when the participants discuss the stories of group cooperation. While group projects tend to bring a lot of frustration, it is often the most valuable memory after the class. This cooperative learning opportunity usually makes what is taught in the class relevant, receives affirmation from peers, and organizes a team to share ideas. This opportunity motivation for teamwork cannot be replaced by any other assessment [8].

Therefore, this study constructs an FL environment with different types of interactive classroom activities to explore the learning preferences of Gen-Z students.

2.2. Individual Flipped Learning

The research findings of Seemiller and Grace [3] indicate that Gen-Z students are more inclined to learn individually than interpersonally. Students enjoy exploring their own values first and then sharing the results with their peers [3]. This research carried out the following tasks for implementing an individual flipped learning (IFL) environment: (1) Conduct classroom activities that encourage students to solve problems with the support of teacher guidance, online information search, and critical thinking; and (2) Collect feedback from students to understand their learning progress and experience.

Findings from the Northeastern University's Innovation Survey indicated that Gen-Z students prefer a hands-on learning environment [6]. Gen-Z students value the practicality of what they learn in everyday life and are eager to apply it in a variety of settings [3]. In particular, problem-solving discussions can increase students' interest in learning [15]. In the IFL of this study, the online platform is used as a means of computer-mediated communication between teachers and classmates. Each student independently completes classroom assignments, watches instructional videos, searches for relevant information online, thinks about answers, and reports to the platform individually.

2.3. Cooperative Flipped Learning

Hashim [15] suggested that teachers create a cooperative learning environment where students can realize their creations through teamwork. The theoretical background of cooperative learning is based on Vygotsky's sociocultural constructivism and the zone of proximal development theory [16]. It encourages students to work with their peers and cooperatively apply their knowledge to solve problems. During cooperative learning, team members learn a certain subject through continuous interactions to gain professional knowl-

edge [17]. It can help students consolidate personal experiences and interact with team members to achieve shared learning goals, thereby improving learning effectiveness [18]. Kettunen, Kairisto-Mertanen, and Penttilä [19] also recommended that group-based learning can promote the social constructivist theory more than self-learning.

Many previous studies adopted the notion of cooperative flipped learning (CFL) [20–22]. They reported that CFL increased students' academic achievement and learning satisfaction compared to TL. Furthermore, research results from Lee, Jeon, and Hong [23] demonstrated that CFL positively affected motivation, but it negatively affected achievement. On the other hand, IFL showed the opposite results. Therefore, this study followed the suggestions of Lai [24] and created a CFL environment that encourages peer interaction and teamwork projects. Classroom activities at CFL divide students into small groups to perform problemsolving tasks by using online platforms as a means of communication between group members. The purpose of CFL is to enhance students' learning motivation and improve their problem-solving ability. Students are required to apply the theories of computer networks to real-life applications and complete their narrative work in creative projects.

2.4. Progressive Interactive Learning Experience

The case study results reported by Ng and Nicholas [25] show that students who follow a progressive instructional learning model can consistently complete all the required tasks of an online learning program. Although the effects of cooperative learning on classroom interactions have been studied [23,26,27], few studies have focused on students' perspectives, specifically after they sequentially engage in several different learning methods [25]. Therefore, this study divides the one-semester course into three parts and adopts three different learning methods, from TL to IFL and then CFL, to increase the diversity of learning styles.

3. Methodology

3.1. Research Questions

The overarching research questions to guide the investigation of this progressive teaching method are:

- (1) How do IFL and CFL affect Gen-Z students' understanding of learning?
- (2) How do IFL and CFL affect the learning motivation of Gen-Z students?
- (3) After students participate in different learning styles in sequence, how are their learning experiences different?

3.2. Setting and Participants

This research was conducted in computer networking courses offered by the College of Business at Chung Yuan Christian University (CYCU). The participants in this study were 100 freshmen in the Department of Information Management. The investigation lasted for eighteen weeks. Students who participated in this study used the same book, *Computer Networking: A Top-Down Approach*, by James F. Kurose and Keith W. Ross (2017, 7th edition).

The computer networking courses introduce the functions of the Internet protocol stack and the networking mechanism required for data transmission. The teaching goals were to guide students to (1) understand the basic concepts of data communications and develop their protocol design reasoning; (2) identify necessary networking properties and select an appropriate network; and (3) create a problem-solving network model.

The CYCU developed an online learning platform called i-learning to improve management efficiency for learning and teaching. This study used the i-learning platform to maximize the effect of computer-supported collaborative learning for students. The functions of the i-learning online platform include (1) online examination, discussion, homework, and survey; (2) cloud storage for uploading teaching materials, such as instructional videos, PowerPoint slides, or other teaching materials; and (3) a cooperative learning platform as an e-learning facilitator allowing students to interact online with the teacher and classmates anywhere and anytime.

3.3. Methods of the Research

This study explores the effects of individual and cooperative flipped learning on learning comprehension and learning motivation with those of traditional lectures using a mixed-methods design. This study adopted a mixed research method design. The first goal was to obtain quantitative data through questionnaires to understand students' learning experiences. The second goal was to obtain students' perspectives on the progressive teaching method through student feedback. Through meticulous research design and requirements analysis, a progressive teaching method has been proposed. In the flipped learning method, this study allows students to experience the changes in learning comprehension and learning motivation caused by gradually increasing the degree of interaction. The purpose of this research is to allow students to experience different learning methods, so that students can find suitable learning methods that can improve their overall learning efficiency. During the study, the teaching methods were gradually adopted to take different levels of interactive classroom activities. The TL method takes a lecture teaching, and the class time is simply unidirectional interaction from the teacher to the students. When using IFL, students must complete online learning before class, and then the classroom time is used to guide students to complete homework. Through the chance of two-way interaction between the teacher and students, the teacher can understand the individual learning conditions of students. During the execution of CFL, class time is used to conduct group projects and encourage cooperative learning between students. Students jointly complete the group projects through cooperative learning and classroom interaction.

3.4. Instructional Design of the Program

To maximize the effectiveness of this research, this study used the ADDIE (analysis, design, development, implementation, and evaluation) model of instructional design [28] as a framework for designing this program. We firstly analyze the learning preferences of the students, design teaching goals, and propose teaching strategies. Then, we develop the flipped learning environment on the i-learning platform and implement the proposed three-stage teaching method sequentially. Finally, we evaluate the effectiveness of the teaching methods and conclude the research results. The design process of this teaching program is shown in Figure 1, which is divided into three main steps: (1) research design, (2) teaching practice, and (3) performance evaluation.

- Step 1. Analyze learning preferences
 - Step 1-1. Identify teaching vision: Explore the trend of interactive pedagogy and analyze the knowledge characteristics of computer networking courses.
 - Step 1-2. Set teaching goals: According to the learning preferences of Gen-Z students, define the research goals and evaluation methods.
- Step 2. Design and develop teaching environment
 - Step 2-1. Design teaching methods: Decide to use flipped learning method with progressive interactions to enhance students' comprehension and learning motivation.
 - Step 2-2. Develop an online learning platform: Design instructional video and digital teaching materials for each teaching stage.
- Step 3. (Teaching Stage I) Implement the TL method
 - Step 3-1. Carry out traditional lecture: Deliver core content through conventional classroom teaching.
 - Step 3-2. Assess learning situation: Observe students' notes and find that they could not understand and apply knowledge.
- Step 4. (Teaching Stage II) Introduce the IFL method
 - Step 4-1. Implement flipped teaching: guide students to self-study online, summarize the key points of the course, and encourage students to share their experiences in online forums.

Step 4-2. Conduce classroom activities: Ask questions in class and prompt students to recall learning and solve problems individually.

- Step 5. (Teaching Stage III) Execute the CFL method
 - Step 5-1. Arrange teamwork projects: Initiate digital storytelling projects in the classroom to encourage cooperative learning in groups.
 - Step 5-2. Organize computer–support interaction: Introduce cooperative learning by placing students in teams to solve problems, thereby enhancing learning motivation.
- Step 6. Evaluate research results
 - Step 6-1. Analyze learning feedback: Collect student feedback and questionnaire results to understand student preferences for different interactive learning methods.
 - Step 6-2. Conclude continuous improvement plans: Design improvement plans and summarize research findings to continuously improve learning outcomes

Research	Analyze Learning Preferences	 Identify teaching vision Set teaching goals
Design	Design and Develop Teaching Environment	Design teaching methodDevelop online learning platform
	(Teaching Stage I) Implement TL Method	Carry out traditional lectureAssess learning situation
Teaching Practice	(Teaching Stage II) Introduce IFL Method	Implement flipped teachingConduct classroom activities
	(Teaching Stage III) Execute CFL Method	Arrange teamwork projectsOrganize computer-support interaction
Performance Evaluation	Evaluate Research Results	Analyze learning feedbackConclude continuous improvement plans

Figure 1. The instructional design process of this study.

3.5. Progressive Teaching Approach

In this study, we followed the three-level learning pyramid of the pedagogy proposed by Bennet and Bennet [29] to categorize the content of computer networks into three levels: (1) surface-level—memorized knowledge, such as facts, data, concepts, and messages; (2) shallow level—social interactions, such as debates and group exchanges; and (3) deep level—practical experiences acquired through learning. This study investigates whether the proposed progressive teaching method affects students' subjective experience of learning comprehension and learning motivation. The characteristics of the three-stage teaching method are shown in Table 1.

Characteristics	Teaching Stage I	Teaching Stage II	Teaching Stage III
Knowledge Level	Surface	Shallow	Deep
Learning (internal	Awareness,	Causality,	Effortful practice,
reflection and	Memorizing,	Coherence,	Insights,
comprehension)	Understanding	Meaning-making	Lived experience
Learned Knowledge	Information,	Conscious,	Mostly unconscious,
	Conscious	Causality	Pattern detection
Exhibited Behavior	Remembering,	Explaining,	Creating,
	Communicating,	Anticipating,	Intuiting,
	Acting	Problem-solving	Predicting
Solving Problems	Simple Problems	Structured Problems	Unstructured Problems

Table 1. Characteristics of the three teaching stages of this study.

The three-stage teaching method proposed in this study is matched to three teaching activities at different knowledge levels. In Teaching Stage I, surface-level knowledge was presented to students. In this stage, the students acquired knowledge from the lecturer, memorized surface-level content, and solved simple problems. In Teaching Stage II, shallow-level knowledge was presented to the students. Students in Teaching Stage II learned about the causality and logic behind the knowledge content to solve structured problems. In Teaching Stage III, students organized groups to review deep-level knowledge, participated in problem-solving projects, and jointly integrated learning experiences into the learning-by-doing processes.

The learning styles and objectives for the proposed three-stage pedagogy are tabulated in Table 2. This research adopts different teaching methods at each stage. The purpose is to enable the same group of students to gradually experience different learning methods to truly experience the different learning effects. The proposed pedagogy expands individual learning to interactive networked learning and then to collaborative group-based learning.

Characteristics	Teaching Stage I	Teaching Stage II	Teaching Stage III
Pedagogy	Traditional lecture	Individual flipped learning	Cooperative flipped learning
Learning Style	Traditional learning	Individual learning	Cooperative learning
Learning Objectives	Describe concepts related to computer networking and solve problems in textbooks	Analyze data communication problems and propose networking mechanisms	Team up to study computer networking problems and cultivate problem-solving skills
Activities	Independently learn knowledge and do homework.	Watch instructional videos before class and solve protocol-related problems in class	Collaborate as a team on digital storytelling projects on computer networking issues
Teaching Purpose	Learn professional knowledge	Enhance learning comprehension	Enhance learning motivation

Table 2. Learning styles of the three teaching stages of this study.

In Teaching Stage I, the traditional lecture (TL) focuses on content internalization and personal practice. Students attend classroom lectures, review textbooks, complete homework, and take quizzes to effectively strengthen content memorization and foster theoretical reasoning skills. The purpose of teaching is to recite professional knowledge and answer textbook questions. Teaching Stage II adopted individual flipped learning (IFL), focusing on combining FL with individual classroom activities. IFL takes advantage of the spontaneity of online platforms to facilitate student learning [29]. The digital materials used in IFL included instructional videos and slides [30]. Students analyzed problems specific to data transmission and learned to apply suitable networking technologies to solve communication problems during in-class learning-by-doing activities The course also encouraged students to convert computer networking issues into mind maps and association diagrams, thereby enhancing students' comprehension

Teaching Stage III focused on group discussions and topical activities to encourage debate, reflection, and negotiation. Students applied their computer networking knowledge and participated in grouped problem-solving activities to realize cooperative flipped learning (CFL). These activities brought students from different fields together to collectively solve problems both in the classroom and online. Students can gain a better understanding of the course content by working in a team and learning from their teammates, thereby improving learning motivation.

3.6. Research Procedure

3.6.1. Teaching Stage I: Traditional Lecture

As shown in Table 3, the teacher provided traditional face-to-face teaching in Teaching Stage I. The students took notes during the lesson, and the teacher assigned homework as practice and scheduled tests to measure students' learning effectiveness. As shown in Figure 2, a student note records the processes of establishing and closing a TCP (transmission control protocol) connection. Observing the students' class notes can find that students just copied the content written by the teacher on the whiteboard. Students only recorded the surface of the teaching content and lacked the design meaning that can deepen their understanding. Students exhibited only a preliminary understanding of the course content. The student feedback also supported this inference.

Environment	Activities and Tasks	Time
	The teacher provides lectures in class	30–40 min
In class	Students take notes in class	5–10 min
(Classroom Instruction)	The teacher explains learning goals in class	1–3 min
	Q&A in class	Flexible
Out of class	Students do homework/taking online quizzes	2 h
(After class)	Students share learning feedback online	10 min

Table 3. Implementation of Teaching Stage I.

3.6.2. Teaching Stage II: Individual Flipped Learning

As tabulated in Table 4, the pedagogy used in Teaching Stage II is the IFL method. Students first engaged in self-learning on an online platform before class. The content of the instructional videos included knowledge delivery and case studies. The online teaching materials help students understand how to realize the service quality of different data communication through the cooperation of communication protocols. In class, students participated in problem-solving activities in the classroom. They were allowed to search for relevant information online to help them solve problems. Following the findings of Cevikbas and Kaiser [13], this study observes student notes as a means of understanding student learning. Figure 3 shows one example of student notes in this stage. Figure 3a shows that students annotated airline systems with airline functional layers to help understand the function of each protocol layer. Furthermore, Figure 3b shows that students can not only describe what they have learned, but also internalize the course content into a mind

map, which can help them improve their learning comprehension. That is, the pedagogy of Teaching Stage II improved students' understanding of the course content.



Figure 2. An example of student notes in Teaching Stage I.

Table 4. Implementation of Teaching Stage II.

Environment	Activities and Tasks	Time
	Students watch instructional videos online	15–20 min
Out of class (FL before class)	Students taking notes after studying online	5–10 min
(i E beloie enos)	Students share learning feedback online	10 min
	The teacher summarizes the teaching content in class	10–20 min
	Q&A in class	Flexible
In class (Individual classroom	Students do homework/taking quizzes/classroom activities under the guidance of the teacher	15–20 min
activities)	Students actively search for additional learning resources online to solve the classroom problems	Flexible
	The teacher preview the progress of the course next lesson	3–5 min



Figure 3. An example of student notes in Teaching Stage II: Using (**a**) a narrative script for an airline system to illustrate (**b**) the function of each layer of the Internet protocol stack.

Lo and Hew [9] asserted that the underlying risk of implementing a flipped-classroom course design is that students could lack a sense of responsibility for independent learning, resulting in students skipping online lectures and extracurricular activities and attending classroom lectures and activities instead. Therefore, we introduced a third teaching stage to encourage team interaction and enhance learning motivation.

3.6.3. Teaching Stage III: Cooperative Flipped Learning

To help students understand the course content from different perspectives while participating in team activities and engaging in discussions, we combined the FL with cooperative classroom activities in Teaching Stage III. Table 5 shows the implementation of this stage. First, students watched instructional videos before class. Then, the teacher raised real-world data communication problems during class and asked students to form teams to study the problems. After the class, students continued to work on assignments in teams. Workloads were allocated based on the different expertise of team members. The process allowed students to gain a deeper understanding of the course content while producing the final results.

The roles of the teacher are crucial in this stage. The teacher serves as a promotor, role model, evaluator, informer, and planner, promoting students to draw on what they have learned to solve problems [31]. In this study, the teacher guided students to create digital narrative works on computer networking issues. Through cooperative learning and storytelling skills, students can deepen their understanding of teaching content [32]. The effectiveness of this stage was measured based on group performance, learning logs, and survey results.

An example of student teamwork results is illustrated in Figure 4. The presentations were based on the transport layer of Internet protocol suite. The students adopted an anthropomorphic approach in transforming the TCP and user datagram protocol (UDP) into virtual characters and attempted to distinguish TCP and UDP from the perspective of reliable data transfer. Students used a comic style to represent TCP and UDP through the characteristics of two courier companies. Then, they used these two courier characters to make a comic strip. They compared the processing mechanisms of the two courier companies in connection establishment, network congestion, packet loss, and reliable data transmission to connect the comics with the course content. Students were very proud of their collaborative model, with some members analyzing course content, some scripting, and others creating comics. Feedback from students on group work indicated that they were very immersed in teamwork and found the activity had the effect of increasing interest and motivation in learning.

Environment	Activities and Tasks	Time
	Students watch instructional videos online	15–20 min
Out of class (FL before class)	Students taking notes after studying online	5–10 min
(I E before class)	Students share learning feedback online	10 min
	The teacher summarizes the main points of the instructional videos	10–15 min
	Q&A in class	Flexible
In class (Cooperative	The teacher introduces challenge problems and assigns them to students	5 min
classroom activities)	Students team up to study the assigned problems	10–15 min
	Students share what they have learned and discuss it with teammates to identify gaps in achieving targets	20–30 min
	Students brainstorm and determine group works	10–15 min
	Students actively search for additional learning resources online to solve assigned tasks	Flexible
Out of class (Online discussion after class)	Students share learning experiences and discuss with each other	Flexible
arter class)	Students propose their solutions to the ill-defined problems	Flexible

Table 5. Implementation of Teaching Stage III.



Figure 4. An example of student creation in Teaching Stage III: Introducing the two protocols of the transport layer in comic style.

4. Results

4.1. Questionnaire Survey

A questionnaire survey was administered after each teaching stage to collect quantitative data. All participants were similar in that they did not differ significantly by age and year of college. All students are taught by the same teacher. The items of the three questionnaires are tabulated in Table 6. The researcher used a five-point Likert-type response format (1 for the lowest impact and 5 for the most positive impact), where each student could choose to agree or disagree. Questionnaire A focused on Teaching Stage I. It aimed to measure the effects of TL on the students' learning effectiveness. Questionnaire B focused on Teaching Stage II. It aimed to compare the effects of TL and IFL on students' learning effectiveness. Questionnaire C focused on Teaching Stage III. It aimed to compare the effects of IFL and CFL on students' learning effectiveness.

Table 7 presents the results of three questionnaires (A, B, C), each with three items labeled (e.g., a1, a2, a3, etc.). A total of 100 students participated in the survey on the improvement of their learning experience. The table presents the frequency of student approval for each item. We can discuss the survey results separately in terms of learning comprehension, learning interest, and learning motivation, as follows.

- For learning comprehension shown in Table 8, the mode was 3 (39%) in Teaching Stage I, 3 (46%) in the second stage, and 4 (43%) in the third stage. This result shows that as the teaching mode changes from TL to IFL to CFL, the learning comprehension of most students gradually improves through individual and cooperative interaction. The variance of these three stages is reduced from 0.7011, 0.6236, to 0.51, and the number of students who scored 2 also dropped from 6%, 5%, to 1%. This shows that differences in students' comprehension levels can also be mitigated through different types of interactive activities.
- From the learning interest shown in Table 9, the modes of the three teaching stages were 4, and the number of people who scored 4 points had an increasing trend, from 38% to 45% and 42%, although the number of students who scored 5 points

gradually decreased in the three stages, mainly because the difficulty of the content gradually increased from surface level, shallow level, to deep level. However, the standard deviation of the overall learning interest in these three stages dropped from 0.8185, 0.7263, to 0.7228. This result shows that students are still able to maintain their interest in learning as the teaching mode changes from TL to IFL to CFL, although the difficulty of teaching content gradually increases. That is, the learning challenges posed by the difficulty of professional courses can be compensated by well-designed interactive activities.

For the learning motivation shown in Table 10, the mean was 4.15 in Teaching Stage I, 4.22 in Teaching Stage II, and 4.24 in Teaching Stage III. The percentage of students who scored 5 points gradually increased from 33%, 39%, to 40% in the three stages. These results show that the average value of students' learning motivation increases gradually. Moreover, the number of students scoring full marks gradually increased in the three-stage teaching method.

Table 6. Questionnaire survey items.

Questionnaire A: (Be Conducted after Teaching Stage I)

a1. In your opinion, how effective is attending classroom lectures on improving learning comprehension?

- a2. In your opinion, how effective is attending classroom lectures in increasing interest in learning?
- a3. In your opinion, how effective is attending classroom lectures in improving learning motivation?

Questionnaire B: (Be Conducted after Teaching Stage II)

- b1. In your opinion, how effective is previewing instructional videos and participating in classroom activities in improving learning comprehension?
- b2. In your opinion, how effective is previewing instructional videos and participating in classroom activities in increasing interest in learning?
- b3. In your opinion, how effective is previewing instructional videos and participating in classroom activities in improving learning motivation?

Questionnaire C: (Be Conducted after Teaching Stage III)

- c1. In your opinion, how effective is engaging in group discussions in improving learning comprehension?
- c2. In your opinion, how effective is engaging in group discussions in increasing interest in learning?
 - c3. In your opinion, how effective is engaging in group discussions in improving learning motivation?

Oursetienseine	τ.	Freque				iency	
Questionnaire	Item	1	2	3	4	5	Total
٨	a1	0	6	39	37	18	100
A (Teaching Stage I)	a2	0	2	33	38	27	100
(Teaching Stage I)	a3	0	0	18	49	33	100
P	b1	0	5	46	35	14	100
D (Teaching Stage II)	b2	0	0	35	45	20	100
(Teaching Stage II)	b3	0	0	17	44	39	100
C	c1	0	1	42	43	14	100
(Taaahing Stage III)	c2	0	0	41	42	17	100
(Teaching Stage III)	c3	0	0	16	44	40	100

Table 7. Results of the three questionnaires.

Tr	Frequency									Std.
Item	1	2	3	4	5	Total	Mode	ode Mean	variance	Deviation
a1	0	6	39	37	18	100	3	3.67	0.7011	0.8373
b1	0	5	46	35	14	100	3	3.58	0.6236	0.7897
c1	0	1	42	43	14	100	4	3.7	0.51	0.7141

Table 8. Results of the questionnaires on the degree of improvement in learning comprehension.

Table 9. Results of the questionnaires on the degree of increased interest in learning.

Trans	Frequency							M	¥7. •	Std.
Item	Item 1 2	2	3	4	5	Total	Mode	wiean	Variance	Deviation
a2	0	2	33	38	27	100	4	3.9	0.67	0.8185
b2	0	0	35	45	20	100	4	3.85	0.5275	0.7263
c2	0	0	41	42	17	100	4	3.76	0.5224	0.7228

Table 10. Results of the questionnaires on the degree of improvement in learning motivation.

T.	Frequency								X 7 •	Std.
Item	1	2	2 3 4 5 Total Mo	Mode	Mean	Variance	Deviation			
a3	0	0	18	49	33	100	4	4.15	0.4875	0.6982
b3	0	0	17	44	39	100	4	4.22	0.5116	0.7153
c3	0	0	16	44	40	100	4	4.24	0.5024	0.7088

Overall, students expressed that the proposed progressive pedagogy was better at enhancing learning effectiveness than TL. Based on the aforementioned points, we inferred that the three-stage pedagogy effectively improved students' learning comprehension and motivation.

4.2. Student Feedback and Perspectives

This study believes that motivation to learn is the meaning constructed by learners as they continue to engage in activities, experiences, or interactions. Therefore, the subjective experience and cognitive feelings of students in this progressive teaching environment are the focus of this study on the collection of student feedback and perspectives. It is particularly suitable for exploratory research to uncover the teaching experiences, course design, student feedback, and achievements of the proposed innovative pedagogy. This study collected the students' learning feedback and online discussions to illustrate the students' thoughts on participating in the course. These feedbacks are first-person descriptions of subjective experiences and corresponding feelings that students actively share. The learning motivation strategies described by these experiences can be integrated into lessonplanning strategies and will contribute to improving the effectiveness of the progressive teaching method.

The learning experiences of two students in Teaching Stage II are illustrated in Figure 5. In Figure 5a, the student expressed that the learning-by-doing activities helped him connect theory with everyday life. The assignments encouraged deeper contemplation, allowing him to find practical examples for the application of theories taught in the course. Another student compared the learning-by-doing activities with traditional assignments and expressed his learning experience in Figure 5b. He agreed that the graphical representation of theoretical transformations helped him better understand the course content. He said that the classroom activities required students to understand the content and communicate the content to others. This process helped him understand the content more deeply than TL.

The learning experiences of students in Teaching Stage III are illustrated in Figure 6. A student shared that, as shown in Figure 6a, in TL mode, he only concentrated on the

completeness of his notes instead of understanding the content of the class. However, after participating in the group work, he can review the content in more detail and listen to the opinions of other classmates. He said that group activities improved his academic performance. Another student shared, as shown in Figure 6b, saying that in traditional face-to-face lectures, he did not completely understand the concepts taught in class. After participating in the group work and discussing the content with his teammates, he was able to better understand the content of the course. He also expressed that giving examples helped him understand the content better and that teamwork is helpful for learning.

- (a) After the mid-term assignment, I realized that many protocols were similar to parts of everyday life. This made me think that maybe these protocols were designed by referring to lifestyles. Initially, I only focused on packet loss. However, the assignment forced me to continue searching for similarities. After finishing the homework, I found that there was more than one similarity. This was an interesting discovery. Associating protocols with everyday life makes studying significantly easier and more fun.
- (b) This assignment was very interesting. I was able to convert textbook content into a format that both creators and viewers alike. Although textbooks are still required to get a complete description, presenting textbook content as pictures or a story makes learning fundamental concepts much easier!

Figure 5. (a,b) Two examples of student feedback on the IFL in Teaching Stage II.

- (a) I was worried that I would not have enough time during class to take notes. I only focused on copying the content on the board without taking the time to understand the meaning. Later, I realized that I did not understand what I had taken down. Fortunately, the group assignment allowed me to review the content in my mind and listen to the views of my teammates. It helped me gain a better understanding of the content.
- (b) The group assignment helped me better understand the course content. The concepts introduced in class were confusing at first. However, through group discussion, I was able to hear different conceptual explanations, which clarified most of the initial confusion. Using analogies is a bit like concluding. This is not as simple as it seems. You must fully understand the content to clearly describe the content. When I integrated the content for my team, I reviewed the content to identify problems, which gave me a better understanding of the content.

Figure 6. (a,b) Two examples of student feedback on the CFL in Teaching Stage III.

4.3. Student Engagement

Gen-Z students in Taiwan are familiar with online social networking interactions but are cautious about face-to-face peer interactions. When using the IFL method in this study, we observed that most students are used to learning independently and completing tasks on their own. In the beginning, students only interacted with the teacher in class, and had less face-to-face interaction with their peers. Even with the encouragement of the teacher, most of the interaction between classmates is only on the online platform. They do not engage in course-related discussions until they watch other students' learning experiences posted online. When implementing the CFL method, the teacher required students to join groups and encouraged them to collaborate. Because students' face-to-face interaction in class increased their interest in learning, students not only actively engaged in face-to-face discussions in their spare time but also interacted through online social networks. These opportunities for peer interaction deepen students' engagement with the coursework and increase their interest in cooperative learning. As shown in Figure 7, students participated in face-to-face meetings after class to discuss solutions to complex problems.



Figure 7. Two photos show students creating a face-to-face learning atmosphere after class and actively participating in collaborative learning projects.

5. Conclusions

The learning characteristics of Gen-Z students are different from previous generations. Their learning preferences are self-learning, achievement-driven, a desire to be affirmed by their peers, and enjoyment of peer interaction. Research on motivation to learn online shows that Gen-Z wants support for relatability, affirmation, and opportunity. This study aims to explore a teaching method suitable for Gen-Z students to study computer networking courses in a college of business. The research process uses the ADDIE instructional design model to analyze learning preferences, design teaching methods, develop online instructional platforms, implement a progressive teaching method, evaluate teaching effectiveness, and collect student learning experiences.

This study proposes a progressive three-stage teaching method in which students are gradually guided to participate in traditional lectures, individual flipped learning, and cooperative flipped learning over a semester. This study places particular emphasis on making each student aware of the differences between the three learning styles and providing feedback on their learning experience. The purpose of this study is to investigate changes in students' learning experiences, focusing on changes in learning comprehension and learning motivation.

5.1. Research Findings

The progressive three-stage teaching method proposed in this study focuses on gradually improving the learning motivation of Gen-Z students, ranging from the face-to-face flipped learning that increases the motivation of relatability to the classroom interactive activities of individual flipped learning that can improve the motivation of affirmation, and finally the teamwork project of cooperative flipped learning that can improve the motivation of opportunity. The results of the learning experience questionnaire showed that the proposed progressive teaching method improved students' understanding of computer networking courses and enhanced their motivation to learn. Furthermore, student feedback on their subjective learning experiences showed that the proposed progressive classroom interaction activities not only recognized individual learning outcomes, but also promoted opportunities for teamwork.

5.2. Theoretical Contributions

The contribution of this research provides educators with a wealth of instructional planning details. This study developed a progressive three-stage teaching method based on learner motivation, and thus collected the subjective experiences of Gen-Z learners who participate in this pedagogy. The research design proposed in this study is similar to the within-subject study design, allowing each student to experience multiple learning approaches. Students' self-reports of subjective learning experiences during the learning process can be used to explore teaching methods that are suitable for students.

5.3. Practical Implications

For educators, different generations of students have different learning preferences. How to choose the most suitable teaching method for contemporary students is often the most important task for educators. Teachers often use a single teaching method with a rigorous assessment mechanism to evaluate the teaching effectiveness. However, when students learn new knowledge for the first time, they do not have enough experience to judge the effectiveness of teaching methods. Students will simply assume that particular courses tend to use the designated pedagogy. They cannot choose a learning style that is suitable for a particular course. Therefore, the progressive pedagogy proposed in this study allows students to experience the advantages and disadvantages of different learning methods, to be able to decide the learning method that is suitable for them.

5.4. Limitations and Further Research

In terms of recommendations for further research, this study proposes a progressive pedagogy targeting the motivations of Gen-Z students currently enrolled in higher education, but the type of courses and the sample size of students are limited. Further development of within-subject study design should be undertaken in the future to better understand whether the proposed progressive teaching method is applicable to various courses and to explore the degree of overall impact on students' learning experience.

Author Contributions: Conceptualization, C.-H.L. and Y.-Y.C.; data curation, C.-H.L. and Y.-Y.C.; formal analysis, C.-H.L.; funding acquisition, C.-H.L.; investigation, C.-H.L.; methodology, C.-H.L.; project administration, C.-H.L.; resources, C.-H.L.; supervision, C.-H.L.; validation, C.-H.L.; visualization, C.-H.L. and Y.-Y.C.; writing—original draft, C.-H.L. and Y.-Y.C.; writing—review and editing, C.-H.L. All authors have read and agreed to the published version of the manuscript.

Funding: This research was funded in part by the Ministry of Education, Taiwan, R.O.C., under the grant number PMS1100258.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: Not applicable.

Conflicts of Interest: The authors declare no conflict of interest.

References

- Sparks & Honey. Meet Gen Z: Forget Everything You Learned about Millennials. 2014. Available online: www.slideshare.net/ sparksandhoney/generation-z-final-june-17 (accessed on 18 April 2022).
- Engelbrecht, J.; Llinares, S.; Borba, M.C. Transformation of the mathematics classroom with the internet. ZDM Math. Educ. 2020, 52, 825–841. [CrossRef] [PubMed]
- 3. Seemiller, C.; Grace, M. Generation z: Educating and engaging the next generation of students. Campus 2017, 22, 21–26. [CrossRef]
- Seemiller, C. Motivation, learning, and communication preferences of generation Z students. EHearsay Electron. J. Ohio Speech-Lang. Hear. Assoc. 2017, 7, 4–9.
- Demir, B.; Sönmez, G. Generation Z students' expectations from English language instruction. J. Lang. Linguist. Stud. 2021, 17, 683–701. [CrossRef]

- Donaghy, R. Innovation Imperative: Meet Generation Z. Northeastern University Innovation Survey. 18 November 2014. Available online: www.northeastern.edu/news/2014/11/innovation-imperative-meet-generation-z/ (accessed on 31 March 2022).
- Szymkowiak, A.; Melović, B.; Dabić, M.; Jeganathan, K.; Kundi, G.S. Information technology and Gen Z: The role of teachers, the internet, and technology in the education of young people. *Technol. Soc.* 2021, 65, 101565. [CrossRef]
- Dusseau, M. Motivation and Learning for the Generation Z Learner in Higher Education. Ph.D. Dissertation, Capella University, Minneapolis, MN, USA, June 2019.
- Lo, C.K.; Hew, K.F. A critical review of flipped classroom challenges in K-12 education: Possible solutions and recommendations for future research. *Res. Pract. Technol. Enhanc. Learn.* 2017, 12, 1–22. [CrossRef]
- Hwang, G.J.; Yin, C.; Chu, H.C. The era of flipped learning: Promoting active learning and higher order thinking with innovative flipped learning strategies and supporting systems. *Interact. Learn. Environ.* 2019, 27, 991–994. [CrossRef]
- Flipped Learning Network. Definition of Flipped Learning. 2014. Available online: https://flippedlearning.org/definition-offlipped-learning/ (accessed on 31 March 2022).
- Foldnes, N. The flipped classroom and cooperative learning: Evidence from a randomised experiment. Act. Learn. High. Educ. 2016, 17, 39–49. [CrossRef]
- Cevikbas, M.; Kaiser, G. Flipped classroom as a reform-oriented approach to teaching mathematics. ZDM Math. Educ. 2020, 52, 1291–1305. [CrossRef]
- Karabulut-Ilgu, A.; Cherrez, N.J.; Jahren, C.T. A systematic review of research on the flipped learning method in engineering education. Br. J. Educ. Technol. 2018, 49, 398–411. [CrossRef]
- 15. Hashim, H. Application of technology in the digital era education. Int. J. Res. Couns. Educ. 2018, 2, 1–5. [CrossRef]
- Doolittle, P.E. Vygotsky's zone of proximal development as a theoretical foundation for cooperative learning. J. Excell. Coll. Teach. 1997, 8, 83–103.
- 17. Leung, A. Boundary crossing pedagogy in STEM education. Int. J. STEM Educ. 2020, 7, 1-11. [CrossRef]
- Joubert, J.; Callaghan, R.; Engelbrecht, J. Lesson study in a blended approach to support isolated teachers in teaching with technology. ZDM Math. Educ. 2020, 52, 907–925. [CrossRef]
- Kettunen, J.; Kairisto-Mertanen, L.; Penttilä, T. Innovation pedagogy and desired learning outcomes in higher education. *Horizon* 2013, 21, 333–342. [CrossRef]
- Eryilmaz, M.; Cigdemoglu, C. Individual flipped learning and cooperative flipped learning: Their effects on students' performance, social, and computer anxiety. *Interact. Learn. Environ.* 2019, 27, 432–442. [CrossRef]
- 21. Lin, Y.T. Cooperative experiential learning in a flipped translation classroom. Int. J. Engl. Linguist. 2019, 9, 842–859. [CrossRef]
- Zhang, L. English flipped classroom teaching model based on cooperative learning. Educ. Sci. Theory Pract. 2018, 18, 3652–3661. [CrossRef]
- Lee, G.G.; Jeon, Y.E.; Hong, H.G. The effects of cooperative flipped learning on science achievement and motivation in high school students. Int. J. Sci. Educ. 2021, 43, 1381–1407. [CrossRef]
- Lai, H.M. Understanding what determines university students' behavioral engagement in a group-based flipped learning context. Comput. Educ. 2021, 173, 104290. [CrossRef]
- Ng, W.; Nicholas, H. A progressive pedagogy for online learning with high-ability secondary school students: A case study. *Gift. Child Q.* 2010, 54, 239–251. [CrossRef]
- AbuSeileek, A.F. Cooperative vs. individual learning of oral skills in a CALL environment. *Comput. Assist. Lang. Learn.* 2007, 20, 493–514. [CrossRef]
- Jafarkhani, F.; Jamebozorg, Z. Comparing cooperative flipped learning with individual flipped learning in a biochemistry course. J. Med. Life 2020, 13, 399–403. [CrossRef]
- 28. Branson, R.K. The interservice procedures for instructional systems development. Educ. Technol. 1978, 18, 11–14.
- 29. Bennet, D.; Bennet, A. The depth of knowledge: Surface, shallow or deep? VINE 2008, 38, 405-420. [CrossRef]
- Kinnari-Korpela, H. Using short video lectures to enhance mathematics learning experiences on differential and integral calculus course for engineering students. *Inform. Educ.-Int. J.* 2015, 14, 67–81. [CrossRef]
- Mokhtar, M.Z.; Tarmizi, R.A.; Ayub, A.F.M.; Nawawi, M.D.H. Motivation and performance in learning calculus through problem-based learning. *Int. J. Asian Soc. Sci.* 2013, *3*, 1999–2005.
- Lazareva, A.; Cruz-Martinez, G. Digital storytelling project as a way to engage students in twenty-first century skills learning. Int. Stud. Perspect. 2020, 22, 1–24. [CrossRef]





Article Engaging ICT Engineering Undergraduates in a Management Subject through First Day of Class Activities: An Empirical Study

Josep Petchamé^{1,*}, Ignasi Iriondo¹, Xavi Canaleta¹, David Riu² and Silvia Necchi³

- ¹ Department of Engineering, Universitat Ramon Llull (URL), La Salle, 08022 Barcelona, Spain; ignasi.iriondo@salle.url.edu (I.I.); xavier.canaleta@salle.url.edu (X.C.)
- ² Department of Management and Technology, Universitat Ramon Llull (URL), La Salle, 08022 Barcelona, Spain; david.riu@salle.url.edu
- ³ Department of Architecture, Universitat Ramon Llull (URL), La Salle, 08022 Barcelona, Spain; silvia.necchi@salle.url.edu
- * Correspondence: josep.petchame@salle.url.edu

Abstract: The expectations, attitudes, engagement, and motivation of students are key elements when designing learning activities. Several studies have been implemented and different strategies and activities have been analyzed to improve the aforesaid aspects of learning content. In the context of the New Learning Context (NLC), this paper presents the findings of two first day of class activities aimed at engaging engineering students in a business and management subject from the very first moment: an empirical study conducted by means of a survey answered by engineering students in Information and Communication Technologies (ICT), followed by an interactive activity between students and instructors carried out through a reciprocal interview activity. The survey was performed with the objective of identifying what they 'liked' and 'disliked' on their first day of class of a business subject. The findings are presented and compared with previous studies and have proven to be mostly consistent with previous academic work. Finally, a reciprocal interview activity was chosen to potentially enhance the students' engagement and motivation. According to the feedback received, this activity was positively valued by the students.

Keywords: engagement; engineering students; first day of class; ICT; management; motivation; reciprocal interview activity

1. Introduction

All first encounters that human beings have with someone or with something generate initial impressions or perceptions that tend to remain in their minds for some time. Just by looking at someone's face, even for less than a second, people make judgments about that person [1–3], although accuracy is not granted [4,5]. In fact, short behavioral observations, from half a minute to five minutes, seem to be enough [6,7]. First impressions about people may be shaped by clothing [8], body language [9], and so on. Shaping first impressions also applies even when thinking about products, such as website aesthetics [10]. Furthermore, first impressions can condition beliefs and behaviors [11]. In terms of education, the first day of class may have a multisided impact on students [12–14].

Several researchers have studied whether students' first impressions have a longlasting impact on their perceptions of subjects and instructors. Buchert et al. [15] concluded that students formed lasting impressions about academic staff within the period of the first two weeks of class. Laws et al. [16] found that the impressions students had formed during the first week persisted until the end of the semester.

Most ICT engineering syllabi include, in addition to specific engineering knowledge, other topics that provide an all-round formation. In fact, both specific technical knowledge and nontechnical competencies are required [17,18], including teamwork, communication,

Citation: Petchamé, J.; Iriondo, I.; Canaleta, X.; Riu, D.; Necchi, S. Engaging ICT Engineering Undergraduates in a Management Subject through First Day of Class Activities: An Empirical Study. *Sustainability* **2021**, *13*, 7440. https://doi.org/10.3390/su13137440

Academic Editor: Larry J. Grabau

Received: 27 May 2021 Accepted: 29 June 2021 Published: 2 July 2021

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2021 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). problem solving, or leadership skills [19–23]. Along these lines, basic knowledge in business and management was introduced in engineering programs, since these topics are necessary to complete the training of students and improve their employability [24]. However, some engineering students have shown a degree of reluctance when they were first presented with subjects that they perceived to be quite far from their area of personal interest [25,26], for example, business or management subjects taught in ICT engineering programs. Therefore, all the efforts made by instructors in the first session of a subject to enhance engagement and motivation should be carefully planned because of the positive (or negative) impact that they could have on the students [12,13,27,28].

The general aim of this research was to explore two activities carried out on the first day of class of a management subject, in order to increase the students' expectations, engagement, and motivation. Specifically, the first purpose of this this paper is to describe an empirical study on what undergraduate ICT engineering students liked and disliked on the first session of class, contextualized in a management subject. The second objective of this study is to show the students' assessment of a reciprocal interview activity carried out during the second part of the session. Both activities were designed to enhance the appeal of the management subject and allow students to meet their peers and instructors.

The article contributes to the literature by analyzing the outcomes of two different first day of class activities hypothesized to enhance the appeal of a management subject. Given that all the participants in this research were second-year ICT undergraduates, all the findings about the efforts oriented to increase expectations, attitudes, and engagement can shed light in designing good practices to be included in subjects that are not included as core topics according to the perceptions of some students.

2. State of the Art

All the efforts and activities that instructors may implement to increase expectations, attitudes, engagement, and motivation on the first day of class of a subject can be crucial to their success in teaching the entire subject. The relevance of this topic is reinforced by the fact that most books dealing with teaching have a chapter dedicated to the first day of class, e.g., [29–33]. Therefore, the findings of previous research works are synthesized as follows.

2.1. Enhancing Engagement and Motivation the First Day of Class

Motivation to learn is a construct that has been defined by different authors, i.e., [34,35], and can be defined as identifying chosen individual behaviors to reach a specific goal [36]. The motivation to learn has been formalized by means of different theories as shown in [37], the contemporary ones being summarized in [38] as follows: expectancy-value, attribution, social cognitive, goal orientation, and self-determination.

The engagement construct has been conceptualized through different definitions that may include different components [39,40]. Engagement is an observable action, as it can be defined as 'energy and effort in action' [40], and some tips to enhance engagement have been identified in research works [41].

On the one hand, according to several research works, students' motivation is correlated with academic success [42,43] or with an impact on their engagement [44]. In fact, motivation may improve different academic outcomes [45–47]. On the other hand, engagement is related, among other issues, to improved achievement [48,49], decreased dropout rates [50], or to the creation of a positive class climate [51].

The very first day of class can be seen as an exceptional opportunity to implement activities that may help develop students' expectations [12,52,53], attitudes [13,54], engagement [12,55–58], or motivation [13,14], while also determining the learning environment and class atmosphere for the remaining sessions of the subject [59,60]. Although such actions are obviously designed to improve class dynamics, some may have negative effects on students' perceptions.

2.2. Studies about 'What Likes' and 'What Dislikes' to Students the First Day of Class

Several studies have been performed with the aim of identifying both the 'likes' and 'dislikes' that students preferred on their first day of class. The main works and findings include the following.

In an empirical study, Perlman & McCann [53] identified what they labelled as 'works well' and 'peeves' on the first day of class by means of two open-ended questions. They identified and taxonomized seventeen different categories and added the number of occurrences for each one of the normalized options that they identified from a survey of 570 undergraduate students. It should be noted that in some cases an item that 'worked well' for a student could be a 'peeve' for another student. The general trends of students' preferences were the following: general information about the subject (syllabus, overview of the subject, etc.); grading system and information about the instructor (background, teaching style). Among the peeves, homework assignments and beginning the subject content the first day of class were ranked at the top of the list.

Henslee, Burgess & Buskist [61] asked 146 undergraduate students by means of a twenty-nine item survey (twenty-two items to be ranked and seven open-ended questions) about the first day activities, with the aim of identifying student preferences. Results detected that student's favorites were 'information about the class structure' and 'coursework'.

Basset [62] surveyed 249 university students, identifying the following as valued preferences: information about the subject difficulty, professional information about the instructor, structure and content of the classes, procedures followed in class, and also personal information about instructors and peers.

In an empirical study, Eskine & Hamer [63] asked 230 undergraduate students to replicate the aforementioned Perlman & McCann empirical study [53]. The authors asked the identical open-ended questions and classified the answers according the same seventeen categories that were formerly identified in the study performed by Perlman & McCann. In terms of 'likes', the top findings were the same, whereas when talking about 'dislikes', the two top topics were 'poor use of class time' and 'beginning subject content'.

2.3. Activities Carried out the First Day of Class

During the first day of class, a variety of different actions and activities can be performed to achieve different goals [27,55,64]. Iannarelli, Bardsley & Foote [64] highlighted four basic actions that can be performed on the first day of class: explaining subject expectations, where the content of the syllabus plays a key role; learning about students; introducing the instructor; and establishing the right tone.

Some of the actions that have been experienced during the first day of class are listed as follows:

- The most basic activity could be to introduce the academic staff and present the syllabus. Sometimes the instructors start giving contents after the presentation or choose to end the first session of class. Along this line, several activities are described in [65] as examples of 'what not to do' during the first session of class session.
- Creating positive and/or negative 'experiences' on purpose during the first session of class. For instance, Wilson & Wilson [13] showed two different videos explaining the syllabus to different groups of students. In one of the videos, the instructor gave the presentation in a friendly way, whereas in the other one the instructor presented the syllabus while avoiding emotional tone and followed the syllabus presentation by another video that generated a homework assignment to be performed. Another experience, related to a psychology subject, is described in LoSchiavo, Buckingham & Yurak [66], where an instructor showed up at the classroom and after asking the students to fill out some information, he told them to stand up and face the back of the room; later, after some minutes and once the real instructor appeared, they discussed the topic of obedience.
- Introducing topics to create students' interest in the subject. Within this category, icebreakers could be included. Different activities were performed in different fields

to create interest. Different academic experiences can be mentioned as an example, as follows: regarding economics, Helmy [67] played a lottery to assign a country to students in order to discuss their development problems; as for statistics, Bartsch [54] asked their students to generate anonymous questions to be answered during class on the first day, and Bennet [68] also analyzed probability by means of matching students' dates of birth; in the context of physics, Gaffney & Whitaker [69] asked students to answer Fermi's questions, in other words, to quantify questions to which it was quite difficult to obtain the exact solution in terms of their quantification, an experiential learning activity to introduce topics about 'operations management' [70]; using a Readers' Theatre technique [71]; or just whipping [72], a teaching activity to promote students' participation.

• Reciprocal interview activity. As described in different papers [12,73], a reciprocal interview activity consists of following these steps: (1) create groups of students; (2) offer a potential list of questions, as examples, to ask the instructor during the interview phase; (3) each group of students discusses the set of predefined questions that will then be asked to the instructor during the interview activity once the speaker of the group has been selected; (4) carry out the reciprocal interview activity in class, or the instructor asks the different groups what is the same, and finally; (5) students ask the instructor.

According to the academic literature, several objectives may be achieved by means of a reciprocal interview activity, as shown in different research works: building an awareness of students' and the instructor's goals and expectations [12,58,73]; gathering information about peers, the instructor, etc. [55,58]; encouraging class discussions and generating more comfortable interactions students-instructor [12,55,58,73]; creating a lasting effect on students' attitudes [52]; influencing students' motivation by increasing their perception of the interest and usefulness of the subject and also by transmitting attention to them [14] and; establishing a positive climate at class [28].

2.4. NLC at La Salle URL or Leveraging the First Class Session

Due to the new reality and possibilities of the educational sphere (digital natives, new technologies, etc.) the way of teaching may change and evolve [74,75]. In December 2018, La Salle Educational Mission Assembly (AMEL 2018) agreed to design and implement a new educational model in all its educational centers for all the different educational stages, from children's education to universities. This new model was named the 'New Learning Context' (NLC) and is currently being deployed in Spain after two years of design and implementation [76]. This deployment involved 104 centers, including two university colleges, such as La Salle Campus Barcelona (Universitat Ramon Llull) and La Salle Campus Madrid (Universidad Complutense de Madrid).

The NLC model [76,77] is based on five pedagogical principles that constitute a nonvisible substrate of the educational model. These principles are implemented through five learning environments, as shown in Figure 1. In short, the five pedagogical principles are as follows:

- Interiority: The educational model transcends the academic field, considering personal growth as an inseparable part of education.
- Mind (body and movement): This principle tries to convey the idea that learning takes
 place beyond the classroom. The NLC considers the use of space and its organization
 as the third educational agent, with students and instructors being the other agents.
- Thought Construction: The NLC should generate cognitive skills and abilities, structures, procedures, and strategies that develop different thinking processes and their use.
- Self-Regulated Behavior: The NLC creates spaces and experiences in which autonomous learning habits are encouraged, where each student learns to self-regulate his/her own pace, intensity, effort, commitment, and time required to reach the learning goals, which can be achieved through different paths.

• Social Dimension of Learning, which is structured on three fundamental levels: (1) the educational spaces, as a pedagogical element which favors social learning; (2) the organizational proposal, which specifies the pedagogical framework of coexistence; (3) the community, as a learning structure.





The concept learning environment was widely studied and analyzed from different academic viewpoints, e.g., [78,79]. The learning environment constitutes an essential part of the NLC pedagogical model. In terms of methodology and teaching, the environment can be considered as a separate space with its own educational purpose. The NLC learning environments are as follows:

- Seminar (i.e., focusing on 'knowing'): seminars are organized as teaching areas to allow the student's acquisition of the concepts. In other words, seminars are knowledge areas in which different learning methodologies (such as, lectures, flipped classrooms, peer-to-peer learning, etc.) can be implemented to achieve the learning outcomes.
- Workshop (i.e., focusing on 'knowing how to be'): workshops are pedagogical environments in which the students use their own strengths to construct their own learning process. In the global context of the NLC Methodological Framework, workshops represent the integration of knowledge, allowing students to fully connect with multifaceted elements of their life. Workshops are orientated to build and develop the students' competences which in turn help them develop their own personality.
- Project (i.e., focusing on 'knowing what to do'): this interdisciplinary learning area enables students to learn competences through complex tasks. Those tasks are characterized by their transversal integration of knowledge, being developed in an interdisciplinary way by means of several different sources (scientific, social, historical, artistic, etc.). Projects are usually focused on a specific source, which is then complemented by the other ones, thus creating a learning environment in which students can truly grow, develop, and construct knowledge.
- Welcoming: this is an area of experience that can help students develop healthy study habits by means of different tools. It ranges from internal elements of the human being (such as reflexional, interiority, consciences, motivation) to organizational needs (planning, to-do lists, and objectives, etc.). Not all the welcome activities are mandatorily programmed at the beginning of the session. In fact, some activities may be scheduled just at the beginning of a specific activity or project.
- Closure: this implies the completion of the task. At this point, students assess the
 work done, make insights for the future, celebrate their achievements, and finish
 their session. This activity enables students to truly appreciate what they have learnt:
 conclusions about what they can make, or simply how they can take advantage of
 these conclusions, as well as being aware of the mistakes they have made and how
 to learn from them. In the same way as the welcoming sessions, closure sessions are

not necessarily programmed at the end of the day, nor do they always last the same length of time.

During the design period of the NLC, the focus was on substantiating and validating the pedagogical model, while different lines of research were created to assess the impact on the deployment of the model. Along this line, academic research has been performed on several NLC topics (e.g., about redesigning a subject [36], assessment issues [26,80], etc.). Three of the previously aforementioned areas (seminars, workshops, and projects) are well-established elements, which have been widely used and studied in a great amount of academic research. Nevertheless, similar efforts in terms of research have not been made for the other two elements (welcoming and closure, in NLC terminology). Therefore, all the efforts in researching items related to the first day of class will positively affect the effective implementation of the NLC.

3. Methods

The research was focused on obtaining data that helped to adjust both the activities and the content of the first day of class in a management subject taught to ICT engineering undergraduates. In this case, the research objectives related to the first day of class were: (1) to obtain information from the students about which activities they preferred to do, (2) to assess students' reactions to a reciprocal class-interview activity carried out in the first session.

Figure 2 shows a methodology to continue finetuning a subject taught in the context of an official undergraduate program once feedback from the students is collected. The initial design of a subject is clearly marked by the requirements established in the official program of the studies, according to the 'VSMA Framework' [81] (in Spanish, VSMA is the acronym that stands for Ex-Ante Assessment, Monitoring, Modification, and Accreditation). The main inputs of this design of each subject are the definition of content and methodologies along with the needs of academic staff and infrastructures. At this level should also be included the Smart Classroom (SC) [82,83], a technology that offers new teaching and learning possibilities and that was deployed in most of classrooms and laboratories in September 2020 at La Salle URL. Once classes begin, a review and update mechanism must be established. Usually, the main elements to analyze the operation of the subject are the surveys completed by students as well as the opinions of the academic staff. The text written in blue refers to the new elements that the present research work incorporates. On the one hand, the NLC establishes a general framework for the use of new teaching methodologies that the subjects should incorporate. On the other hand, two first-class instruments are proposed to complement the usual mechanisms to refine the subject.

The research related to both activities that were carried out on the first day of class was conducted in the context of the second-year management subject 'Value Chain and Financial Economics'. This is a core subject of all the ICT engineering programs taught at La Salle URL, where seven undergraduate ICT engineering programs are taught: Audiovisual Engineering, Computer Engineering, Electronic Engineering, Engineering in ICT Management, Multimedia Engineering, Telecommunications Systems Engineering, and Telematics Engineering. The activities took place during the first session of class in the first term of the 2020–2021 academic year, and surveys linked to both research activities were handed out to all students that attended the first two-hour class session.



Figure 2. Proposed design and continuous finetuning of a subject in an official undergraduate program once feedback from the students is collected.

3.1. An Empirical Study: First Day of Class, 'What Likes' and 'What Dislikes'

The empirical study to determine the 'likes' and 'dislikes' of ICT engineering students the first day of class was performed during the first session of a second-year management subject. The questionnaire was handed out to the students in class just before a brief presentation of the instructors, after saying 'Hello, good morning' and giving the instructors their first name. At the top of the first page of the survey, data about participants were collected to allow segmentation of the data: 'university degree that you are studying', 'entrance' or access to engineering studies, 'age', and 'gender'. However, no identification in terms of names or any other data that could identify the student was required, resulting in an anonymous form. Each side of the paper contained one of the two open-ended questions: 'Which things would you like an instructor to do on the first day of class of a subject?' on the front side of the paper and 'Which things would you not like an instructor to do on the first day of class of a subject?' on the back side of the paper. Students were invited to answer the questions with a short sentence per idea (ideally, from one to five words, despite not being a specific restriction) to force students first to think and then write a synthesized idea.

Once all the questionnaires had been collected, the answers given by the students were reclassified in homogeneous categories independent of the literal wording of the answers. The methodology that was followed was the same performed by Perlman & McCann [53] and Eskine & Hammer [63]. However, homogenization was not initially restrained to the former taxonomy resulting from the Perlman & McCann's empirical study [53] in order to allow new items to be identified, resulting in the items listed in Table 1. Once the classification was completed, eventual matches in terms of different wording were identified to compare findings, despite keeping some topics disaggregated that were linked to motivation and utility of the subject once the data was analyzed.
7440
13,
2021,
Sustainability

Items ¹	,Ti	kes,	ĮQ,	islikes'
	u	%	ц	%
General overview, syllabus, content, & evnertations	107	78.10	16	11.68
Describing assessment & grading	75	54.74	12	8.76
Utility & objectives of the subject	44	32.12		
Instructor: introducing background	42	30.66	7	5.11
lcebreaker: doing activities	38	27.74	J	3.65
Getting to know classmates	31	22.63		
Positive attitude of instructor towards students	25	18.25		
Doing a 'nonconventional' class	22	16.06		
Motivating students	22	16.06		
Beginning subject content	15	10.95	95	69.34
Instructor's advice to pass the	15	10.95		
Class takes up full session (2 h)	8	5.84	6	4.38
Explaining instrumental elements (software_etc.)	4	2.92		
Reviewing content (that should be	б	2.19		
known))			
Doing a test to check initial knowledge	1	0.73	24	17.52
Poor use of class time			20	14.60
Homework assignments			6	6.57
Instructor: poor teaching			8	5.84
Instructor: uncaring, intimidating			7	5.11
Instructor: not being empathetic			6	4.38
Beginning subject content without			4	2.92
prior introduction			c	
Instructor: bad attitude			7	1.40
instructor: not pering enurusiastic about the subject			2	1.46
	¹ Items have be	an inferred from findings of two open-en	nded questions.	

Table 1. First day of class (all ICT engineering students): 'likes' & 'dislikes'.

3.2. Instructors' and Students' Interactions: A Reciprocal Interview Activity

The reciprocal interview activity was designed to allow the interaction of the whole class with the instructors and vice versa. The mechanics of the reciprocal interview were described briefly in Section 2.3 of this paper. Groups of four students were created and the activity was carried out. One of the instructors sat at the instructor's table writing all the answers given by the different groups of students to the questions raised by the instructors (fifteen minutes were left to prepare the interview, while the interview lasted around fifteen minutes). The final stage of the activity was the instructors answering the different questions asked by the students (eight minutes were left to prepare the interview, while the interview, while the interview, while the interview, activity lasted twenty minutes).

Once the reciprocal interview was completed, a survey to assess the activity was handed out to all the students to be answered individually to assess students' perceptions of the activity. Again, data about participants were collected to segment the data, in fact the same that was collected in the previous survey: 'university degree that you are studying', 'entrance', 'age', and 'gender'. Yet again, no identification in terms of names was required, resulting in an anonymous form. The survey was structured in four blocs: (1) comfort with approaching the instructor (four items); (2) student comfort with class participation (three items); (3) evaluation of the activity, a reciprocal interview (two items); and (4) ' ... the activity helped me:' (four items). A space under the tittle 'Any comments?' was left to include students' commentaries. The content handed out to the students was an evolution of the questions surveyed in the Hermann & Foster questionnaire [12].

4. Findings and Results

This section presents the findings obtained once students carried out two activities specifically designed to improve their expectations, engagement, and motivation.

4.1. First Day of Class, 'Likes' and 'Dislikes'

The main findings of the first survey were obtained from the answers to two openended questions. Students were asked what they liked and what they disliked, in terms of the class content of the first day and the activities to be carried out during the first session of the subject 'Value Chain and Financial Economics'.

The number of students enrolled in the subject was 164, of which 7 did not participate in any class activity throughout the course. The survey was completed by 137 of them (M = 19.42 years old, SD = 1.32), who answered the open-ended questions. In terms of gender, there were 33 females (24.09%; M = 19.09 years old, SD = 1.08) and 104 males (75.91%, M = 19.52 years old, SD = 1.38). Once the data were collected, responses were normalized without trying to match all of them with the previous taxonomy presented in [53], in order to avoid being conditioned by previous findings. Once the collected data had been classified and reworded, resulting in the items listed in Table 1, the contents were compared with the aforementioned research and then further reworded in a second stage to be able to compare findings. Table 1 shows what students 'liked' and 'disliked' in terms of actions and percentages on the first day of class for the surveyed students. The students' answers shown in Table 1 were standardized to allow adding students' assessments under the identical concepts.

Table 2 shows the actions taken and the percentages of 'likes' and 'dislikes' expressed by the ICT Management engineering students on the first day of class. In all seven engineering programs, the same core subjects are taught in the first academic year. In second year, students take different subjects according to their specific engineering degree program. The distinctiveness of the ICT Management engineering program is that the weight that management subjects have in terms of ECTS (acronym that stands for European Credit Transfer and Accumulation Systems) is much bigger in comparison with the other six ICT engineering programs. The total number of ICT Management engineering students that answered the survey was 15 (M = 18.93 years old, SD = 0.45), with 3 females (20.00%, M = 19 years old, SD = 0) and 12 males (80.00%; M = 18.83 years old; SD = 0.57).

Items ¹	'Likes'		'Dislikes'	
	n	%	n	%
General overview,				
syllabus, content, &	13	86.67		
expectations				
& grading	10	66.67		
Instructor: introducing				
background &	8	53.33	1	6.67
experience				
Getting to know	6	40.00		
classmates Icebreaker: doing				
activities	5	33.33		
Motivating students	4	26.67		
Utility & objectives of	3	20.00		
the subject	0	20100		
Instructor's advice to	2	13.33		
Good instructor's				
attitude towards	1	6.67		
students				
Doing a		=		
'nonconventional' class	1	6.67		
Beginning subject				
content	1	6.67	9	60.00
Reviewing previously	1	6.67		
acquired content	1	0.07		
Doing a test to check			4	26.67
Initial knowledge			1	6.67
Instructor: uncaring			1	0.07
intimidating			1	6.67
Instructor: bad attitude			1	6.67

Table 2. First day of class (ICT Management engineering students): 'likes' & 'dislikes'.

¹ Items have been inferred from findings of two open-ended questions.

Table 3 shows the actions and percentages of all ICT engineering students in terms of 'likes' and 'dislikes' on the first day of class, excluding ICT Management engineering students. The total number of students matching this criterion was 122 (M = 19.48 years old, SD = 1.39), with 30 females (24.59%; M = 19.10 years old, SD = 1.14) and 92 males (75.41%; M = 19.60 years old, SD = 1.44).

A Chi-squared analysis compared listed and unlisted frequencies of the ICT Management engineering students to all the other ICT engineering students. No significant differences emerged in the list of 'likes' and 'dislikes' except for the one of the 'likes' items. Chi-squared analysis identified that a greater percentage of the ICT Management engineering students (53.3%) listed 'Instructor: introducing background & experience' as a 'like' than all the other ICT engineering students (27.9%), resulting in an $X^2(1, n = 137) = 4.807$, p = 0.04.

Items ¹	4	'Likes'		ikes'
Tiento	n	%	n	%
Comonal commission				
General overview,	04	77.06	16	12 11
synabus, content, &	94	77.00	18	13.11
expectations				
Describing assessment	65	53.28	12	9.84
& grading				
Utility & objectives of	41	33.61		
the subject				
Instructor: introducing		27 0 7		1.00
background &	34	27.87	6	4.92
experience				
Icebreaker: doing	33	27.05	.5	4.10
activities	00	2,100	U	
Getting to know	25	20.49		
classmates	20	20.17		
Positive attitude of				
instructor towards	24	19.67		
students				
Doing a				
'nonconventional' class	21	17.21		
session				
Motivating students	18	14.75		
Beginning subject		11.10	24	T2 40
content	14	11.48	86	70.49
Instructor's advice to				
pass the subject	13	10.66		
Class takes up full				
session (2 h)	8	6.56	6	4.92
Explaining				
instrumental elements	4	3 28		
(software etc.)	1	0.20		
Reviewing content				
(that should be known)	2	1.64		
Doing a tast to shock				
initial knowledge	1	0.82	20	16.39
Poor use of class time			10	15 57
Foor use of class time			19	15.57
Homework			9	7.38
Instructor: poor			8	6.56
teaching				
instructor: uncaring,			6	4.92
intimidating				
Instructor: not being			6	4.92
empathetic				
Beginning subject				
content without			4	3.28
introduction				
Instructor: not being				
enthusiastic about the			2	1.64
subject				
Instructor: bad attitude			1	0.82

Table 3. First day of class (all ICT engineering students, excluding ICT Management engineering students): 'likes' & 'dislikes'.

¹ Items have been inferred from findings of two open-ended questions.

This qualitative research was planned according to the guidelines presented in previous works [84–89]. As presented in [84], the steps followed were: (1) define the objectives of the research; (2) identify potential respondents; and (3) decide on the methods for collecting

the data and the analysis methodology. The tool chosen to collect students' preferences about activities to be done during the first day of class followed the same methodology used in [53,63], while questions about the activity were collected and analyzed in line with the study of [12]. The data from the first day of class were collected from students by means of an open-ended questionnaire, a tool that enables students to answer while minimizing possible biases [90]. It should be noted that the sample of surveyed respondents was homogeneous since all of them were students of a second year ICT engineering undergraduate program who took a management subject. Once data about the first day of class were collected from students by means of open-ended questionnaire, the items were analyzed separately by three different researchers to increase the validity of the obtained findings [84].

4.2. Reciprocal Interview Activity

The reciprocal interview activity took place at the final part of the class session once a synthetic presentation of the syllabus was given. A summary of the main results obtained from the interviews when activity was completed is presented in this subsection.

After completing the reciprocal interview activity, a second survey was performed to evaluate the students' assessment of the interview activity. It was the last activity before finishing the first-class session of the mandatory management subject 'Value Chain and Financial Economics' in all ICT engineering programs. Information was extracted from a survey in which students' statements were rated by means of a 5-point Likert scale (ranging from 1—*not at all*—to 5—*a very great extent*—). The results, in terms of mean, median, and standard deviation, are presented in Table 4.

The total number of students that filled the survey with valid results was 131 (M = 19.33 years, SD = 1.56), and six forms were discarded because data were not completed. In terms of gender, the form was answered by 30 females (22.90%; M = 19.10 years, SD = 1.10) and 101 males (77.10%; M = 19.30 years, SD = 1.56).

To assess reliability, the internal consistency of the different sections of the questionnaire was measured by means of the Cronbach's alpha [91,92]. 'Comfort with instructor interaction' had a Cronbach's alpha of 0.72, and removing sequentially each one of the items of the section, the measures were 0.64, 0.74, 0.57 and 0.68, respectively. Further, removing C2 increased the analyzed value, moving from 0.72 to 0.74. Therefore, there was no need to remove any of the items since the whole former section obtained a value greater than 0.7 [92]. 'Student comfort with class participation' had a Cronbach's alpha of 0.69, and removing sequentially the items results were 0.47, 0.62 and 0.76, respectively. Again, by removing an item, S3, the value increased. As in the other case, the result of the second section was nearly 0.7, so consistency was achieved. All the other items, despite been grouped in different sections, were linked to individual concepts. The questionnaire had content validity according to the view of the researchers since it covered the different aspects to be measured once compared with other studies, e.g., [12].

Answers to the questions asked by instructors were compiled carefully. Two instructors asked questions, while another instructor noted down students' responses in front of the whole class. In the same way, students' questions were answered by instructors. All the opinions and statements given by students were analyzed in depth by the instructors once the session was completed. For illustrative purposes, some of the answers to the questions asked by the instructors were about the grading system, continuous assessment activities, key dates, instructors' experience, instructor motivation to teach the subject, and utility of the subject, among others. In addition, some of the students' answers to the instructors' questions were about their expectations for the subject, what they thought they were to be taught, their previous knowledge in management and business, etc.

Reciprocal Interview			
Questionnaire	M	Mdn	SD
Comfort with instructor			
interaction			
C1. 'Talking to the instructor	3.03	4	0.70
about assignments'	3.93	7	0.70
C2. 'Asking the instructor			
questions during class	3.93	4	0.70
sessions'			
C3. 'Talking to the instructor	3 71	4	0.91
during office hours'	5.71	Ŧ	0.91
C4. 'Emailing the instructor	3.87	4	1 19
with questions'	0.07	-	
Student comfort with class			
participation			
S1. 'Participating in group	4 20	5	1.01
activities during class'	1120	0	1101
S2. 'Sharing ideas and	4.00	4	0.76
opinions during class'		-	
S3. Working group activities	3.67	4	1.23
outside class hours'			
Evaluation of the activity			
E1. 'Would you recommend			
other instructors do this	3.80	4	0.86
activity at the beginning of the			
term?			
E2. Did this activity seem to	1.47	1	0.64
be a waste of time?			
the activity helped me:			
H1. To understand what was	4.13	4	0.74
expected in class			
H2. To work hard to do well	4.00	4	0.88
In the class			
H3. To become more	4.12	4	0.74
connortable participating in	4.13	4	0.74
Class			
ri4. To share concerns with	4.00	4	1.07
the instructor			

Table 4. First day of class (ICT engineering students). Reciprocal interview: Data and statistics.

5. Discussion

This section presents the analysis of the results shown in Section 4 regarding two activities that were carried out on the first day of class of a management subject taught in seven engineering programs.

5.1. What Students 'Like' and What They 'Dislike'

The first part of this research deals with ICT engineering students' opinions on what they prefer to do on the first day of class. The collected data were classified and quantified in terms of percentage of mentioning. All the items sourced from the answers of the whole pack of ICT engineering students are shown in Table 1. Hence, on the top of the 'likes' rank appears, 'General overview, syllabus, content, and expectations', followed by 'Describing assessment & grading', as shown in Table 5. Equally, when thinking about 'dislikes', 'Beginning subject content' is the most common dislike, as shown in Table 6. In addition, both tables include data segmented consistently with Section 4.1.

Poor use of class time

content, & expectations Describing assessment &

General overview, syllabus,

	AICTep ¹	ICTMep ²	ICTeMep ³
General overview, syllabus, content, & expectations	78.10%	86.67%	77.06%
Describing assessment & grading	54.74%	66.67%	53.28%
Utility & objectives of the subject	32.12%	20.00%	33.61%
Instructor: introducing background & experience	30.66%	53.33%	27.87%
Icebreaker: doing activities	27.74%	33.33%	27.05%
Getting to know classmates	22.63%	40.00%	20.49%
Good instructor's attitude towards students	18.25%	6.67%	19.67%
Doing a 'nonconventional' class session	16.06%	6.67%	17.21%
Motivating students	16.06%	26.67%	14.75%

Table 5. Top ranked 'Likes' clustered by engineering programs.

¹ AICTep (All ICT engineering programs); ² ICTMep (ICT Management engineering program); ³ ICTeMep (ICT excluding Management engineering program).

	AICTep ¹	ICTMep ²	ICTeMep
Beginning subject content	69.34%	60.00%	70.49%
Doing a test to check initial knowledge	17.52%	26.67%	16.39%

14.60%

11.68%

8.76%

Table 6. Top ranked 'Dislikes' clustered by engineering programs.

grading	0.70 /0	-	9.0470
¹ AICTep (All ICT engineering]	programs); ² ICTMep (ICT Manage	ment engineering program); 3 IC	TeMep (ICT excluding Management
engineering program).			

As in the previously mentioned research works [53,63], the most common 'dislike' is 'Beginning subject content', as seen in Table 6. 'Doing a test to check initial knowledge' was considered an unpopular activity, according to students' perceptions.

6.67%

_

3

15.57%

13.11%

9.84%

Most of the results found in the research (first column of Tables 5 and 6) are consistent with other previous studies [53,63]. Most of the answers to the ranked items are the same, despite changing the numeric value in terms of percentages associated with citing each one of the items by the students. However, some findings included in Table 5 that differ from the aforementioned previous research studies should be highlighted, as follows: (1) 'Utility and objectives of the subject' are highly ranked (32.12%) on the list compared to previous studies, as they were not found in [63] and only reached 7% in [53]. The fact that this element was in the third position and that it did not appear as a 'Dislike' reveals that it was an action highly appreciated by students, and therefore its inclusion in it on the first day of class could be useful to increase the intrinsic motivation of the students. Intrinsic motivation deals with behaving or doing something in a specific way because the individual (i.e., the student) believes that it is inherently pleasant or interesting [93]. (2) 'Motivating students' was selected by 22 students (16.06%) as a positive action to be done by instructors while this item did not appear in previous referenced research works [53,63].

Analyzing the segmented data subsets, some ideas can be highlighted about the previously mentioned items: (1) 'Utility & objectives of the subject' was cited more often by ICT engineering students (33.61%) than by ICT Management engineering students (20%). This difference could be explained by the fact that undergraduates that are enrolled in the latter program are likely to know more about business topics than students that have chosen a purely ICT engineering program. (2) Quite surprisingly, 'Asking for motivation from the instructor' was cited by more ICT Management students (26.67%) than ICT engineering students (14.75%).

5.2. Analizing the Students' Evaluation of the Reciprocal Interview Activity

Data obtained from the reciprocal interview activity were collected from a second survey articulated to assess pronouncements by means of a Likert scale [94], specifically a 5-point Likert scale. Results of the form were grouped in four categories, two related to students' comfort (with approaching the instructor and with class participation), one evaluating the reciprocal interview activity, and the last one evaluating four statements related to potential benefits of the first day activity that the student actually performed.

All the items related to the label 'comfort with instructor interaction' received a high score value. Within this tag, the statement that received a lower mean value (3.71) was 'talking with the instructor during office hours'. All the other items received values quite close to four. In fact, students asked a lot of questions about the continuous assessment of the subject. When analyzing the elements associated with 'student comfort with class participation', two of the statements were also highly assessed. Again, the item with lower values (and again with the higher variance) was the one related with an action to be done outside class hours, in other words, 'working with their peers outside class hours'. Both low values may suggest that students are initially more oriented to perform their learning activities within class hours. Further research should be done in these specific items to shed light on both statements.

The third section of the form was focused on the assessment of the reciprocal interview activity. Two sentences were provided to check how students assessed the activity, one in positive while the other one is formulated in negative. Both results were consistent, giving good feedback about the activity. Here again, further research to compare the results generated by different first day activities should be carried out. In the framework of the NLC [76,77], a list of available activities to be done in the specific context of engineering and management subjects during the welcoming should be made. This list can be completed after an analysis of the evaluation and effectivity of the potential activities by means of student surveys. It seems that the reciprocal interview activity, according to the results obtained in the survey, worked very well considering the students' opinions, in line with other research works [12].

The last section of the questionnaire was designed to check what specific issues had emerged from the reciprocal interview activity. Understanding expectations, sharing concerns with instructors, and becoming comfortable with their participation in class were very well valuated (all equal to or greater than four).

5.3. Practical Implications of the Findings

Once the main findings of the research are presented, different practical implications emerge. The mere fact that students realize that their opinions are heard and considered increases their engagement [41]. Hence, both activities performed on the first day of class that give voice to the students can enhance their engagement. In addition, the information collected through the answers of both activities is very valuable feedback for the instructors. Moreover, in the case of implementing some students' opinions, engagement can increase since they perceive that their ideas have been valued and applied [41]. Another option that may enhance engagement is to promote the use of new technologies by students [95], and once again, asking students about how they perceive their experience using technology [82] may be crucial. Finally, promoting peer-to-peer interaction also may increase

engagement [96]. In fact, most of these actions took place in the context of this research through both activities that were carried out the first day of class.

5.4. Limitations, Restrictions and Future Research Directions

A first limitation of both studies, the empirical survey and the assessment questionnaire about the reciprocal interview activity, was that surveyed students were restricted to those taking a second-year ICT engineering subject. Consequently, even though students from seven different ICT specialties were asked, which facilitates the comparison of student's perceptions across all the diverse ICT engineering programs, the sample was limited to second-year students. However, and to cope with this limitation, replicating this research work in other subjects has already been already scheduled. A second limitation related to the first day of class activity was that only one option was chosen, a reciprocal interview. Some of the other activities listed in this paper could be implemented in other subjects and in different academic courses to verify and compare results between the different first day activities. A third limitation was that students were invited to answer the survey with short sentences. This option was chosen to force students to first think and then write synthesized ideas. However, this option could lead to limiting the depth of their opinions. In fact, we are going to repeat these activities in the next academic year in different subjects, and we are planning to set a new limit (around twenty words).

Future research works should aim at properly developing the 'New Learning Concept' that is currently being implemented at La Salle URL [76,77]. Firstly, it would be wise to replicate the study of the 'likes and dislikes' of the first day of class in different subjects. Hence, third and fourth-year ICT engineering students should be surveyed in the context of subjects taken for all ICT engineering students to check if findings are similar in terms of items and its percentages. The study should be carried out in the context of engineering subjects and in the framework of business subjects. Secondly, and along the same line, preferences about 'what likes and what dislikes' could be done in the context of ICT master students to analyze eventual differences in students' preferences that may exist between undergraduate and master programs. Finally, besides the reciprocal interview, some other first day activities could be implemented to check its potential success in a technological context.

6. Conclusions

The main contribution of the empirical studies presented in this paper was to shed light about the preferred actions to be done on the first day of class in the specific framework of ICT engineering programs through collecting students' opinions. In fact, asking students preferences and carrying out an activity that promotes their participation is a powerful way to establish the tone of a subject, which impacts engagement and motivation. Findings were quite consistent with previous research in terms of the list of preferences, despite some differences in ranking and percentages. An item related to requiring 'motivating students' by instructors appears recurrently as one of the students' requests. Further, once a specific first day activity was implemented, by means of a reciprocal interview between students and instructors, students' perception about the activity were surveyed. According to the results of the survey based on those interviews, the activity was very positively assessed by students. These findings, results and experiences are very valuable in the framework of the NLC, specifically in the Welcoming stage, because the outcomes obtained from this research will help to develop in a tangible way the new learning strategy that is being implemented at this moment.

Author Contributions: Conceptualization, J.P., I.I. and X.C.; methodology, J.P., I.I. and X.C.; validation, J.P., I.I., X.C. and S.N.; formal analysis, J.P., I.I. and X.C.; investigation, J.P., I.I. and X.C.; resources, J.P., I.I. and X.C.; data curation, J.P. and I.I.; writing—original draft preparation, J.P., I.I. and X.C.; writing—review and editing, J.P., I.I., X.C., D.R. and S.N.; visualization, J.P., I.I., X.C. and S.N.; supervision, I.I. and D.R.; project administration, J.P., I.I. and X.C.; funding acquisition, X.C. and S.N. All authors have read and agreed to the published version of the manuscript.

Funding: The research that has given rise to these results has been carried out through funds from the Secretariat of Universities and Research of the Department of Business and Knowledge of the Generalitat de Catalunya and Ramon Llull Univ. Project: 2020-URL-Proj-058.

Institutional Review Board Statement: The research presented, as well as the design, collection, and management of its data, has been POSITIVE evaluated and APPROVED by the Ethics Committee of the Ramon Llull University with the file number: CER URL_2020_2021_009.

Informed Consent Statement: Informed consent was obtained from all subjects involved in the study.

Data Availability Statement: Data available on request from the authors.

Conflicts of Interest: The authors declare no conflict of interest.

References

- Todorov, A.; Pakrashi, M.; Oosterhof, N.N. Evaluating faces on trustworthiness after minimal time exposure. Soc. Cogn. 2009, 27, 813–833. [CrossRef]
- Willis, J.; Todorov, A. First impressions: Making up your mind after a 100-ms exposure to a face. *Psychol. Sci.* 2006, 17, 592–598. [CrossRef]
- Zebrowitz, L.A.; Montepare, J.M. Faces and first impressions. In APA Handbook of Personality and Social Psychology, Volume 1: Attitudes and Social Cognition; Mikulincer, M., Shaver, P.R., Borgida, E., Bargh, J.A., Eds.; American Psychological Association: Washington, DC, USA, 2014; pp. 252–276.
- Olivola, C.Y.; Todorov, A. Fooled by first impressions? Reexamining the diagnostic value of appearance-based inferences. J. Exp. Soc. Psychol. 2010, 46. [CrossRef]
- 5. Zebrowitz, L.A. First Impressions from Faces. Curr. Dir. Psychol. Sci. 2017, 26, 237–242. [CrossRef] [PubMed]
- 6. Ambady, N.; Rosenthal, R. Thin slices of expressive behavior as predictors of interpersonal consequences: A meta-analysis. *Psychol. Bull.* **1992**, *64*, 431. [CrossRef]
- Ambady, N.; Rosenthal, R. Half a Minute: Predicting Teacher Evaluations from Thin Slices of Nonverbal Behavior and Physical Attractiveness. J. Personal. Soc. Psychol. 1993, 64, 431. [CrossRef]
- Howlett, N.; Pine, K.; Orakçıoğlu, I.; Fletcher, B. The influence of clothing on first impressions. J. Fash. Mark. Manag. Int. J. 2013, 17, 38–48. [CrossRef]
- Greenlees, I.; Bradley, A.; Holder, T.; Thelwell, R. The impact of opponents' non-verbal behaviour on the first impressions and outcome expectations of table-tennis players. *Psychol. Sport Exerc.* 2005, *6*, 103–115. [CrossRef]
- Reinecke, K.; Yeh, T.; Miratrix, L.; Mardiko, R.; Zhao, Y.; Liu, J.; Gajos, K.Z. Predicting Users' First Impressions of Website Aesthetics with a Quantification of Perceived Visual Complexity and Colorfulness. In Proceedings of the Conference on Human Factors in Computing Systems (CHI'13), Paris, France, 27 April–2 May 2013; pp. 2049–2058.
- 11. Yu, M.; Saleem, M.; Gonzalez, C. Developing trust: First impressions and experience. J. Econ. Psychol. 2014, 43, 16–29. [CrossRef]
- 12. Hermann, A.D.; Foster, D.A. Fostering approachability and classroom participation during the first day of class: Evidence for a reciprocal interview activity. *Act. Learn. High. Educ.* 2008, *9*, 139–151. [CrossRef]
- Wilson, J.H.; Wilson, S.B. The First Day of Class Affects Student Motivation: An Experimental Study. *Teach. Psychol.* 2007, 34, 226–230. [CrossRef]
- 14. McGinley, J.J.; Jones, B.D. A Brief Instructional Intervention to Increase Students' Motivation on the First Day of Class. *Teach. Psychol.* **2014**, *41*, 158–162. [CrossRef]
- Buchert, S.; Laws, E.L.; Apperson, J.M.; Bregman, N.J. First impressions and professor reputation: Influence on student evaluations of instruction. Soc. Psychol. Educ. 2008, 11, 397–408. [CrossRef]
- Laws, E.L.; Apperson, J.M.; Buchen, S.; Bregman, N.J. Student evaluations of instruction: When are enduring first impressions formed? N. Am. J. Psychol. 2010, 12, 81–91.
- Martin, R.; Maytham, B.; Case, J.; Fraser, D. Engineering graduates' perceptions of how well they were prepared for work in industry. *Eur. J. Eng. Educ.* 2005, 30, 167–180. [CrossRef]
- Robinson, M.; Sparrow, P.; Clegg, C.; Studies, K.B.-D. Design engineering competencies: Future requirements and predicted changes in the forthcoming decade. *Des. Stud.* 2005, 26, 123–153. [CrossRef]
- Chan, C.K.Y.; Zhao, Y.; Luk, L.Y.Y. A Validated and Reliable Instrument Investigating Engineering Students' Perceptions of Competency in Generic Skills. J. Eng. Educ. 2017, 106, 299–325. [CrossRef]
- Passow, H.J. Which ABET competencies do engineering graduates find most important in their work? J. Eng. Educ. 2012, 101, 95–118. [CrossRef]
- Passow, H.J.; Passow, C.H. What Competencies Should Undergraduate Engineering Programs Emphasize? A Systematic Review. J. Eng. Educ. 2017, 106, 475–526. [CrossRef]
- ABET. Criteria for Accrediting Engineering Programs, 2020–2021 | ABET. Available online: https://www.abet.org/accreditation/ accreditation-criteria/criteria-for-accrediting-engineering-programs-2020-2021/ (accessed on 20 April 2021).

- International Engineering Alliance. Celebrating International Engineering Education Standards and Recognition. Available 23. online: https://www.ieagreements.org/assets/Uploads/Documents/History/25YearsWashingtonAccord-A5booklet-FINAL. pdf (accessed on 20 April 2021).
- Male, S.A.; Bush, M.B.; Chapman, E.S. Perceptions of Competency Deficiencies in Engineering Graduates. Australas. J. Eng. Educ. 24. 2010, 16, 55-68. [CrossRef]
- 25. Pons, D. Relative importance of professional practice and engineering management competencies. Eur. J. Eng. Educ. 2016, 41, 530–547. [CrossRef]
- 26. Petchamé, I.; Iriondo, I.; Riu, D.; Masi, T.; Almazano, A.; Fonseca, D. Self & Peer to Peer Assessment: Evaluating Oral Presentations in a Final Year Engineering Subjecte. In Proceedings of the TEEM'20: 8th International Conference on Technological Ecosystems for Enhancing Multiculturality, Salamanca, Spain, 21-23 October 2020; pp. 784-790.
- 27. Deluse, S. First Impressions: Using a Flexible First Day Activity to Enhance Student Learning and Classroom Management. Int. J. Teach. Learn. High. Educ. 2018, 30, 308-321.
- 28. Foster, D.A.; Hermann, A.D. Linking the First Week of Class to End-of-Term Satisfaction: Using a Reciprocal Interview Activity to Create an Active and Comfortable Classroom. Coll. Teach. 2011, 59, 111–116. [CrossRef]
- 29. Nilson, L.B. Teaching at Its Best: A Research-Based Resource for College Instructors; Wiley: Hoboken, NJ, USA, 2010.
- Rotenberg, R.L. The Art & Craft of College Teaching: A Guide for New Professors & Graduate Students; Routledge: London, UK, 2010. 30. McKeachie, W.; Svinicki, M. McKeachie's Teaching Tips; Cengage Learning: Belmont, CA, USA, 2013. 31
- 32.
- Erickson, B.L.; Peters, C.B.; Strommer, D.W. Teaching First-Year College Students; John Wiley & Sons: San Francisco, SA, USA, 2006. 33 Wolcowitz, J. The First Day of Class. In The Art and Craft of Teaching; Harvard University Press: Cambridge, MA, USA, 1984.
- 34. Keller, J.M. Motivation, Learning, and Technology: Applying the ARCS-V Motivation Model. Particip. Educ. Res. (PER) 2016, 3, 1–13. [CrossRef]
- Pintrich, P.R. A Motivational Science Perspective on the Role of Student Motivation in Learning and Teaching Contexts. J. Educ. 35. Psychol. 2003, 95, 667-686. [CrossRef]
- Petchamé, J.; Iriondo, I.; Riu, D.; Masi, T.; Almazano, A.; Fonseca, D. Project Based Learning or the Rethinking of an Engineering 36 Subject: Measuring Motivation. In Proceedings of the TEEM'20: 8th International Conference on Technological Ecosystems for Enhancing Multiculturality, Salamanca, Spain, 21-23 October 2020; pp. 267-272.
- Kovach, M. A Review of Classical Motivation Theories: Understanding the Value of Locus of Control in Higher Education. 37. Peer-Rev. Artic. J. Interdiscip. Stud. Educ. 2018, 7, 34-53. [CrossRef]
- 38. Cook, D.A.; Artino, A.R. Motivation to learn: An overview of contemporary theories. Med. Educ. 2016, 50, 997–1014. [CrossRef]
- Christenson, S.; Reschly, A.; Wylie, C. Handbook of Research on Student Engagement; Springer Science & Business Media: New York, 39. NY, USA, 2012.
- 40. Bond, M.; Buntins, K.; Bedenlier, S.; Zawacki-Richter, O.; Kerres, M. Mapping research in student engagement and educational technology in higher education: A systematic evidence map. Int. J. Educ. Technol. High. Educ. 2020, 17, 2. [CrossRef]
- 41. Peters, H.; Zdravkovic, M.; João Costa, M.; Celenza, A.; Ghias, K. Twelve tips for enhancing student engagement. Med. Teach. 2018, 41, 632-637. [CrossRef]
- 42. Busato, V.; Prins, F.; Elshout, J.; Individual, C.H.-P. Intellectual ability, learning style, personality, achievement motivation and academic success of psychology students in higher education. Personal. Individ. Differ. 2000, 29, 1057–1068. [CrossRef]
- 43. Linnenbrink, E.A.; Pintrich, P.R. Motivation as an enabler for academic success. Sch. Psychol. Rev. 2002, 31, 313–327. [CrossRef]
- 44. Skinner, E.; Furrer, C.; Marchand, G.; Kindermann, T. Engagement and disaffection in the classroom: Part of a larger motivational dynamic? J. Educ. Psychol. 2008, 100, 765-781. [CrossRef]
- Wigfield, A.; Wentzel, K.R. Introduction to motivation at school: Interventions that work. Educ. Psychol. 2007, 42, 191–196. 45. [CrossRef]
- Valle, A.; Regueiro, B.; Rodríguez, S.; Piñeiro, I.; Freire, C.; Ferradás, M.; Suárez, N. Perfiles motivacionales como combinación de 46. expectativas de autoeficacia y metas académicas en estudiantes universitarios. Eur. J. Educ. Psychol. 2015, 8, 1–8. [CrossRef]
- 47. Sánchez-Prieto, J.C.; Olmos-Migueláñez, S.; Garcia-Penalvo, F.J. Motivation and innovation: Mobile technology acceptance among student teachers. Rev. Iberoam. De Educ. A Distancia 2017, 20, 273–292. [CrossRef]
- Finn, J.D.; Zimmer, K.S. Student engagement: What is it? Why does it matter? In Handbook of Research on Student Engagement; 48. Springer: Boston, MA, USA, 2012; pp. 97-131.
- 49. Lee, J.S. The relationship between student engagement and academic performance: Is it a myth or reality? J. Educ. Res. 2014, 107, 177–185. [CrossRef]
- Fredricks, J.A.; Blumenfeld, P.C.; Paris, A.H. School engagement: Potential of the concept, state of the evidence. Rev. Educ. Res. 50. 2004, 74, 59-109. [CrossRef]
- Reyes, C.R.; Brackett, M.A.; Rivers, S.E. Classroom Emotional Climate, Student Engagement, and Academic Achievement. J. 51 Educ. Psychol. 2012, 104, 700. [CrossRef]
- Hermann, A.D.; Foster, D.A.; Hardin, E.E. Does the First Week of Class Matter? A Quasi-Experimental Investigation of Student 52. Satisfaction. Teach. Psychol. 2010, 37, 79-84. [CrossRef]
- 53. Perlman, B.; McCann, L.I. Student perspectives on the first day of class. Teach. Psychol. 1999, 277–279. [CrossRef]
- Bartsch, R.A. Improving attitudes toward statistics in the first class. Teach. Psychol. 2006, 33, 197–198. 54.
- Anderson, D.M.; Mcguire, F.A.; Cory, L. The first day: It happens only once. Teach. High. Educ. 2011, 16, 293–303. [CrossRef] 55.

- 56. Kreizinger, J. Critical Connections for the First Day of Class. Teach. Profr. 2006, 20, 1.
- Robinson, D. Engaging students on the first day of class: Student-generated questions promote positive course expectations. Scholarsh. Teach. Learn. Psychol. 2019, 5, 183. [CrossRef]
- Case, K.A. The class interview: Student engagement in courses covering sensitive topics. *Psychol. Learn. Teach.* 2011, 10, 52–56. [CrossRef]
- Patrick, H.; Anderman, L.H.; Ryan, A.M.; Edelin, K.C.; Midgley, C. Teachers' communication of goal orientations in four fifth-grade classrooms. *Elem. Sch. J.* 2001, 102, 35–58. [CrossRef]
- Patrick, H.; Turner, J.C.; Meyer, D.K.; Midgley, C. How Teachers Establish Psychological Environments During the First Days of School: Associations with Avoidance in Mathematics. *Teach. Coll. Rec.* 2003, 105, 1521–1558. [CrossRef]
- Henslee, A.M.; Burgess, D.R.; Buskist, W. Student preferences for first day of class activities. *Teach. Psychol.* 2006, 33, 189–191. [CrossRef]
- Bassett, J. Students' First Day of Class Preferences: Factor Structure and Individual Differences. N. Am. J. Psychol. 2011, 13, 373–381.
- Eskine, K.E.; Hammer, E. Students' Perspectives on the First Day of Class: A Replication. Int. J. Scholarsh. Teach. Learn. 2017, 11, Article 5. [CrossRef]
- Iannarelli, B.A.; Bardsley, M.E.; Foote, C.J. Here's Your Syllabus, See You Next Week: A Review of the First Day Practices of Outstanding Professors. J. Eff. Teach. 2010, 10, 29–41.
- 65. Dorn, D.S. The first day of class: Problems and strategies. Teach. Sociol. 1987, 15, 61–72. [CrossRef]
- Loschiavo, F.M.; Buckingham, J.T.; Yurak, T.J. First-Day Demonstration for Social Psychology Courses. *Teach. Psychol.* 2002, 29, 216–219. [CrossRef]
- Helmy, H.E. A lottery on the first day of classes! An innovative structured steps assignment on a partially randomly selected topic. Int. Rev. Econ. Educ. 2016, 21, 41–47. [CrossRef]
- Bennett, K.L. How to start teaching a tough course: Dry organization versus excitement on the first day of class. Coll. Teach. 2004, 52, 106. [CrossRef]
- 69. Gaffney, J.D.H.; Whitaker, J.T. Making the Most of Your First Day of Class. Phys. Teach. 2015, 53, 137–139. [CrossRef]
- Snider, B.; Southin, N. Operations Course Icebreaker: Campus Club Cupcakes Exercise. Decis. Sci. J. Innov. Educ. 2016, 14, 262–272. [CrossRef]
- Lom, B. Classroom activities: Simple strategies to incorporate student-centered activities within undergraduate science lectures. J. Undergrad. Neurosci. Educ. 2012, 11, 64–71.
- Tanner, K.D. Feature Approaches to Biology Teaching and Learning Structure Matters: Twenty-One Teaching Strategies to Promote Student Engagement and Cultivate Classroom Equity. *Life Sci. Educ.* 2013, *12*, 322–331. [CrossRef] [PubMed]
- Case, K.; Bartsch, R.; Mcenery, L.; Hall, S.; Hermann, A.; Foster, D. Establishing a Comfortable Classroom from Day One: Student Perceptions of the Reciprocal Interview. Coll. Teach. 2008, 56, 210–214. [CrossRef]
- Sein-Echaluce, M.L.; Fidalgo-Blanco, Á.; Alves, G. Technology behaviors in education innovation. Comput. Hum. Behav. 2017, 72, 596–598. [CrossRef]
- Fonseca, D.; Conde, M.Á.; García-Peñalvo, F.J. Improving the information society skills: Is knowledge accessible for all? Univers. Access Inf. Soc. 2018, 17, 229–245. [CrossRef]
- 76. La Salle Distrito ARLEP. NCA, Otra Manera de Hacer Escuela; La Salle ARLEP: Madrid, Spain, 2018; ISBN 8472219342.
- 77. La Salle Distrito ARLEP. NCA, Nuevo Contexto de Aprendizaje; La Salle ARLEP: Madrid, Spain, 2020; ISBN 8472219359.
- Moore, J.L.; Dickson-Deane, C.; Galyen, K. E-Learning, online learning, and distance learning environments: Are they the same? Internet High. Educ. 2011, 14, 129–135. [CrossRef]
- 79. Land, S.; Jonassen, D. Theoretical Foundations of Learning Environments; Routledge: New York, NY, USA, 2012.
- Herrero-Martín, J.; Canaleta, X.; Fonseca, D.; Rodríguez-Merino, C.; Kinnear, L.; Amo, D. Designing a multi-scale and multidimensional assessment for a new national educational context. In Proceedings of the TEEM'20: 8th International Conference on Technological Ecosystems for Enhancing Multiculturality, Salamanca, Spain, 21–23 October 2020; pp. 791–796.
- AQU: VSMA Framework. Available online: https://www.aqu.cat/en/universities/titulacions/vsma-framework (accessed on 22 April 2021).
- Petchamé, J.; Iriondo, I.; Villegas, E.; Riu, D.; Fonseca, D. Comparing Face-to-Face, Emergency Remote Teaching and Smart Classroom: A Qualitative Exploratory Research Based on Students' Experience during the COVID-19 Pandemic. *Sustainability* 2021, 13, 6625. [CrossRef]
- ZOOM. Ways of Learning Are Changing: La Salle Campus Barcelona Gives Students Flexibility to Learn from Anywhere Using Zoom. Available online: https://blog.zoom.us/wp-content/uploads/2021/03/Case-Study_La-Salle-University.pdf (accessed on 10 May 2021).
- 84. Bengtsson, M. How to plan and perform a qualitative study using content analysis. Nurs. Open 2016, 2, 8–14. [CrossRef]
- Cypress, B.S. Rigor or reliability and validity in qualitative research: Perspectives, strategies, reconceptualization, and recommendations. *Dimens. Crit. Care Nurs.* 2017, 36, 253–263. [CrossRef]
- Noreña, A.L.; Alcaraz-Moreno, N.; Rojas, J.G.; Rebolledo-Malpica, D. Aplicabilidad de los criterios de rigor y éticos en la investigación cualitativa. Aquichán 2012, 12, 263–274. [CrossRef]

- Morse, J.M. Critical Analysis of Strategies for Determining Rigor in Qualitative Inquiry. Qual. Health Res. 2015, 25, 1212–1222. [CrossRef]
- Morse, J.M.; Barrett, M.; Mayan, M.; Olson, K.; Spiers, J. Verification Strategies for Establishing Reliability and Validity in Qualitative Research. Int. J. Qual. Methods 2002, 1, 13–22. [CrossRef]
- 89. Leung, L. Validity, reliability, and generalizability in qualitative research. J. Fam. Med. Prim. Care 2015, 4, 324. [CrossRef]
- Reja, U.; Manfreda, K.L.; Hlebec, V.; Vehovar, V. Open-ended vs. Close-ended Questions in Web Questionnaires. *Dev. Appl. Stat.* 2003, 19, 159–177.
- 91. Cronbach, L.J. Coefficient Alpha and the Internal Structure of Tests. Psychometrika 1951, 16, 297–334. [CrossRef]
- 92. Taber, K.S. The Use of Cronbach's Alpha When Developing and Reporting Research Instruments in Science Education. *Res. Sci. Educ.* 2018, 48, 1273–1296. [CrossRef]
- Ryan, R.M.; Deci, E.L. Intrinsic and Extrinsic Motivations: Classic Definitions and New Directions. Contemp. Educ. Psychol. 2000, 25, 54–67. [CrossRef] [PubMed]
- 94. Likert, R. A Technique for the Measurement of Attitudes; Columbia University Press: New York, NY, USA, 1932.
- Peñarrubia-Lozano, C.; Segura-Berges, M.; Lizalde-Gil, M.; Bustamante, J.C. A Qualitative Analysis of Implementing E-Learning during the COVID-19 Lockdown. Sustainability 2021, 13, 3317. [CrossRef]
- Christopoulos, A.; Conrad, M.; Shukla, M. Increasing student engagement through virtual interactions: How? Virtual Real. 2018, 22, 353–369. [CrossRef]





Article A STEM Model to Engage Students in Sustainable Science Education through Sports: A Case Study in Qatar

Ruba Ali, Jolly Bhadra, Nitha Siby, Zubair Ahmad and Noora Jabor Al-Thani *

Young Scientists Center (YSC), Qatar University, Doha 2713, Qatar; ruba.ali@qu.edu.qa (R.A.); jollybhadra@qu.edu.qa (J.B.); nitha.siby@qu.edu.qa (N.S.); zubairtarar@qu.edu.qa (Z.A.) * Correspondence: n.al-thani@qu.edu.qa; Tel.: +974-4403-3979

Abstract: Sports has the potential to integrate with different scientific subjects, including materials science and engineering, making it an ideal approach to enhance the students' affinity toward sustainable education in science, technology, engineering, and mathematics (STEM). Amid gradual educational reformations in the state of Qatar, a distinctive STEM program titled, "Science in Sports" (SIS) was launched to investigate STEM integrated learning in secondary school students. The participant students, 248 students (112 females and 136 males) from 15 different government-operated (public) secondary schools, from rural and urban areas, were given STEM workshops on one of the sports materials, during this pilot study, resultantly challenging them to engineer a sports product. The study employed a mixed-method study in which quantitative approaches were applied to analyze the program effectiveness, with a t-test statistical analysis performed over data collected from a period of five continuous years from 2012 to 2017 in five different cycles. A more dominant data collection included pre and post surveys, substantiating observations of the program facilitator and their schoolteachers were included in this research and development (R&D) study to review the student learning behavior for a qualitative approach. Moreover, the results of the strength, weakness, opportunities, and threats (SWOT) analysis provided an overview of the program's effectiveness in implicating the engagement of the students in exhibiting their prototypical skills in engineering sports products along with STEM literacy. Apart from understanding the scientific concepts/principles applied in simple sports applications, student attitudes toward STEM fields augmented, as witnessed by the student productivity.

Keywords: sustainable science education; science implementation; youth empowerment; high school students; STEM; sports science

1. Introduction

Sports can promote social integration, economic development, develop teamwork capabilities, and improve physical and mental health. The state of Qatar has been recently brought to the limelight of the sports industry because it has grabbed the opportunity to host the most sought-after sports event—the FIFA world cup 2022. The coverage and discussions on the event and the wide-spread enthusiasm for the sports is a well-timed opportunity for the educational community to emphasize the positive impact of sports on students of all ages. In this context, trends show a growing inclination in youth toward sports and its related disciplines [1]. A significant percentage of high school students actively participate in sports and enjoy the competitive spirit, recreation, or personal development. Educators believe sports as paramount while providing a wholesome educational experience [2]. They are inspired by successful programs that use sports to empower the youth and educate them. There are studies on successful programs launched in Latin America and the Caribbean that achieved profound results. Additionally, several case studies have been performed on working with youth who were hooked to formal education through sports, changing their lives for the better and paving the way to better

Citation: Ali, R.; Bhadra, J.; Siby, N.; Ahmad, Z.; Al-Thani, N.J. A STEM Model to Engage Students in Sustainable Science Education through Sports: A Case Study in Qatar. Sustainability 2021, 13, 3483. https://doi.org/10.3390/su13063483

Academic Editor: Larry J. Grabau

Received: 4 March 2021 Accepted: 18 March 2021 Published: 21 March 2021

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2021 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/).

41

employment opportunities [2]. Their study describes the success of programs such as "A Ganar," which implemented team-based sports to help at-risk youth learn self-employment skills and enhance their physical, cognitive, and intellectual growth. In light of these studies, a rich background literature was collated by the research team on similar programs to understand the outcomes of implementing sports or sports-based techniques in learning. They were trying to overcome the shortcomings in students' learning goals in Qatar.

Our research program aims to couple science, technology, engineering, and mathematics (STEM) learning with sports in an advanced and different way than the previous reports, taking into account various influential factors in generating considerable interest in science education. We developed different concepts to create an innovative and thoughtful sports-based program capable of enhancing students' attitudes to STEM curriculum and STEM-driven careers. The program requirements needed to consider the factors of an out-of-school environment and supported the construction of a workshop for high school students, providing STEM activities. Apart from improving STEM literacy, this research was also capable of observing multiple competencies in students that could alleviate their cognitive behavior.

1.1. Relevant Literature Background

An uphill challenge faced by educators in Qatar in the recent past was the lack of sufficient development of STEM skills in students. Despite massive effort and investment made toward enriching the STEM program by authorities in the past two decades, the field is still unable to completely meet the great demand of a highly educated workforce that can support a knowledge-based economy. Countries that focused on STEM early on have embarked on a journey of innovations that has started steering the wheel of the global economy to their benefit [3,4]. Hence, the significance of STEM and increasing reliance on the STEM workforce cannot be downplayed for innovation-driven countries trying to maintain leadership in the world's economy [5-7]. Unfavorably for Qatar, various reports cite concerns regarding the lack of qualified Qatari STEM professionals and an obvious deficit in the level of learned skills itself [8–11]. This means strengthening the STEM discipline—its courses, content, or faculties—in itself is not enough; instead, educators should delve deeper into understanding the attitudes or underlying approach of Qatari youth toward STEM education. Identifying what quickly generates interest in sports and applying the same principles to education might be a pioneering but straightforward step in creating a positive impact in young minds to instill interest in STEM learning.

Many educational researchers provided direct insights into improving students' STEM interests with the help of activities instead of traditional instruction [12]. Citing numerous examples of integrating STEM, Wendell et al. [13] incorporated engineering design into the science curriculum with LEGO Mindstorm Kits. Subsequent results were profound, and it proved that students learn scientific concepts faster and better than an instruction-only curriculum. Having students participate in STEM activities is considered an assuring way to increase interest in STEM subjects and foster STEM literacy [6].

Role of Self-Motivation and Motivational Factors

Personal interest is another crucial factor in a student choosing to learn any discipline aiming to establish a career path [14]. Students across all socio-economic backgrounds find school education and learning disconnected from the real world, which makes schoolwork uninspiring or boring, gradually leading to underperformance in school. One of the key traits here that sets some students apart is self-motivation or personal interest. Evidence from previous studies suggests that students set their dispositions toward disciplines such as mathematics and science early in life much before higher education [12,15], indicating the crucial role of various motivation enhancing tools or techniques in modeling student interest at the right stage of growth. It was hypothesized by the Texas Academy of Mathematics and Science (TAMS) that students of different age groups might possess the competencies and skills required to become STEM professionals at varying degrees. It could be an area of importance and interest for educational researchers to identify the key influential factors such as individual interests, teachers, curriculum, and out-ofschool activities that can cultivate motivation in children of all ages. They believe that certain instigating factors can lead many students to favor STEM disciplines, target STEM careers, and promote broader participation in the future STEM workforce. Kilpatrick [16] recommended sports participation as an effective intrinsic motivator, which is also closely related to being a recommended youth development driver in Qatar and can inspire involvement without external incentives. Sports-based activities also act as drivers as in one of the dimensions of motivation, "Motivation to accomplish" [17]. It is also well supported by Choi and Song [18] study in which they discuss the effectiveness of sports in developing life skills and its potential to contribute to a wide range of subjects to make them palatable for students. Hence, we focused on an approach that could apply the students' interest in sports in encouraging them to participate in STEM activities that "speak" science through sports.

1.2. Conceptual Framework

Our method of projecting science through sports-driven activities was based on many analytical findings reported in different previous studies [19–21]. A detailed systematic study was performed on diverse research evidence determining factors for a student's STEM-driven "interest" in secondary education, focusing on both personal and social aspects (both parental and cultural). While introducing the concept of motivational factors, Hanrahan [20] implicated in her study that implementing more activities to reinforce positive attitudes for self-direction is preferential in active learning. Sports-driven activities creating situations similar to that in real-life problems were proven to enhance positive attitudes with their contribution to cognitive skills such as peer collaboration, problem solving, curiosity. These activities implemented to promote student-centered learning are a realistic approach [19,21,22] because they stimulate curiosity through problem-based situations [21]. These activities also enhance logical reasoning, helping the children attain better emotional, social, and physical well-being [23].

Taking into account sports-driven activities, it was crucial for us to construct a wellcoursed framework of activities that can cultivate curiosity, STEM interest and consequently motivate participants to consider careers in similar fields, etc. The Material World Module (MWM) program, which aimed at instilling collaborative scientific research and engineering skills in high school students [24], provided adequate research backings to follow an inquiry and design methodology in our study. This Methodology introduces students to familiarize themselves through technological design, urging them to engage in hands-on activities to demonstrate how science is applied in our daily life routine. We developed the workshops, keeping the study evidence related to the importance of inquiry [25], problem solving, technical experience, and engineering projects to influence students' attitudes toward learning and enhance their science process skills. The first set of workshop activities was developed initially to aim to design sports products such as golf balls and baseballs. These two products were chosen randomly, inspired by the study conducted by [24]. Later, another set of workshop activities were constructed to consider designing sports products such as "concrete bowling ball" and "concrete boat" inspired from the competitions held by the American Concrete Institute (ACI) [26] and American Society of Civil Engineers (ASCE) National Concrete Canoe Competition (NCCC) [27]. However, both workshop activities (details in Figure 1 and Methodology Section, page 13) helped the students understand the structure and physical behavior of diverse composites such as concrete and other multi-layered sports materials and realize the science and correlating their application in daily life. In both sets of sports products (from workshops 1 and 2), we intended to develop the correlation between the underlying scientific concepts in each workshop offering the scientific principles in sports product development. The structure of implementing activities was similar in both the workshops, and the activities complement each other and lead the students to their respective design challenges.



Figure 1. Schematic diagram of "Science in Sports" program methodology. The activity layout is provided for each workshop, demonstrating the daily activities for each day, Workshop 1 and Workshop 2 are directed to learn about concrete materials and sports materials, respectively. Regardless of the layout, the design project does not culminate on day 5 and takes nearly three months to complete after repetitive testing.

1.3. Research Objectives

The study was based on a program, "Science in Sports" (SIS), intended to steer high school students' passion in sports to explore the embedded science and gradually evolve into aspirants in STEM fields. The study emphasizes the program's successful integration of science in sports by solving engineering design challenges that arise while developing a sports product. The research objective was solely focused on acquainting them with STEM and its relevance in daily life by engineering simple sports products. The research questions that need to be addressed during our study are

- 1. Did the student understand the importance of STEM by experiencing science through sports product engineering?
- 2. Will the student consider a STEM-driven career resultantly in the future?

3. Were the students able to understand and correlate scientific principles to the applications in daily life?

Apart from the above questions, the research was capable of deriving the exceptional secondary students' dispositions and self-efficacy in carrying out scientific experiments. The learning environment bolstered by sports products-driven challenges is an influencing factor that will be utilized in attracting Qatar's secondary youth to attain the program goals. Integrating sports as a carrier for acquainting scientific knowledge, the study will examine the diverse possibilities in promoting science process skills, reinforcing their attitudes toward science and engineering fields by intriguing them for creative learning.

2. Materials and Methods

The SIS program was designed and evaluated by employing a mixed-method study that implemented both quantitative and a more dominant qualitative analysis. While the former method used statistical tools to assess student attitudes toward STEM importance, relevance in daily life and future careers, the latter involved the observations of facilitators and school teachers who mentored the students during the program. Other qualitative data were retrieved from field notes, media (photos and videos), and student artifacts to derive an outlook on developing reasoning competencies and dispositions of the students as they progressed through the research. The annual program functioned within a three-month tentative period. The students experienced science and engineering through a science workshop directing them to construct a sports product for the engineering design challenge (Figure 1).

2.1. Participants and the Educational Context

The current study was conducted on 248 high school students including 112 girls and 136 boys from grades 10 and 11, who attended the SIS program from 2012 to 2017 in five cycles, over the five continuous years. The three-month time taken by each batch of students in one year for participating in the program was addressed as one cycle, and therefore, the study witnessed five batches of students from 15 schools over five cycles, as in Table 1. The participant students were provided with STEM workshops on different types of composite materials. While some schools were provided with "Workshop 1," for familiarizing themselves with concrete composites, other participating schools engaged in "Workshop 2" on some other sports material used for rebounding balls. The schools were assigned either of the topics on a random choice. For the first two cycles, the schools were from urban areas, whereas the schools involved in the experimental study during the third cycle were from rural areas, hence a smaller number of participants. However, in consequent years, we experimented on students from both urban (fourth cycle) and rural schools (fifth cycle), respectively.

However, there we	ere student withdrawals, and hence t	he study was conducted on a	fewer number of participants	s (248) according to
the questionnaire a	analysis, for those who have complet	ted both pre and post question	nnaires.	
Cycle	Total Number of Students	Number of Withdrawals (Nw)	Qatari Students	Schools

Table 1. Demographic details of participants and Qatari student percentage. The total participant count for all five years is 275.

Cycle	Iota	Numbe	r of Stu	aents	Withdrav	vals (Nw)		Qatari S	tuaents		Sch	0015
	Μ	%	F	%	Nw	%	Μ	%	F	%	Μ	F
Cycle 1: 2012	24	47.1	27	52.9	0	0	19	79.1	26	96.2	1	1
Cycle 2: 2013	36	60	24	40	4	6.6	36	100	24	100	2	1
Cycle 3: 2014	24	63.5	16	36.5	5	12.5	18	75	14	91.3	2	1
Cycle 4: 2015	36	60	24	40	8	13.3	31	86.1	21	87.5	2	1
Cycle 5: 2016	29	45.3	35	54.6	10	15.6	27	93.1	35	100	1	2
Total	149	54.1	126	45.8	27	9	131	87.9	120	95.2	8	6

The number of schools participating per cycle was then limited to three due to the restricted working time that could be engaged by the facilitator for the workshop hours and the product development phase that took place during their after-school period. The engineering of sports products consumed most of the program schedule because it was carried out within two months. Even though the program tenure was three months, a certain amount of time (as mentioned in the next section) was taken off for their regular school activities, including examinations scheduled according to the academic calendar.

The schools with maximum national student enrollment were chosen for the program during the initial screening process because the study primarily targeted Qatari students and their development. The schools were also requested to strictly maintain at least 75% of national students for the program (due to the aim of the national capacity building). The final filtrate of students was selected according to the first-come, first-served basis, regardless of their intellectual capacity, and was grouped into students of three or four, maintaining a demographic ratio of non-national students and national students as 1:2 or 1:3, respectively, as in the case of participation of non-national students. (refer to Table 1).

The program was organized and delivered by the program facilitator, who is also the first author of this article. She holds a degree in engineering and has seven years of experience in STEM education. She is skilled in identifying the key areas of interest in a student and helps in applying them to overcome their learning deficits. She has experience designing and delivering around 90 engineering workshops for school students ranging from primary and secondary stages. The findings stated in this article corresponds to her observations noted in the daily report for each session and interaction with the students. For this study, with the assistance of other STEM professionals (PhD/Masters/bachelor's degree holders) of the Al-Bairaq team at the Qatar University Young Scientists Center (YSC), field notes, observations, and various resources were employed for the detailed analysis. Here, it is essential to mention that the YSC team has an extensive 10 years of experience in constructing workshops and educational activities. Table 1 details the number of participants and the percentage of Qatari student participation during each cycle.

2.2. Methods

The facilitator offered extensive two-hour daily workshops for a week with diverse hands-on activities (refer to Figure 1) to engage the students in experiential learning. The main goal of providing workshops to the students was to acquaint them with enough scientific information to design and construct the sports products—concrete boat, bowling ball, golf ball, and baseball. Each school was either provided with STEM workshops-"Workshop 1" and "Workshop 2" (refer to Figure 1)—leading to the respective sports products. The workshops were followed by informal sessions for designing and constructing the product. During these sessions, the students had free choice to carry out the procedure; however, they would have to adhere to the facilitator's evaluation criteria, as in Table 2. These sessions were also allocated to repeated testing and designing of the product prototypes. In case of failure in creating the product according to the desired criteria, they would have to start over again or fix the products for positive results. The facilitators recorded student responses and learning behavior during the activities from attending workshops until they accomplished the final products. The facilitator observations were later analyzed to understand the student response or learning behavior exhibited during their project course.

The method exercised on the students through each of the following workshops is briefed below.

2.2.1. Workshop 1

Students' groups provided with "Workshop 1" were monitored by their respective schools for successful completion of a concrete boat or a concrete bowling ball as a final product. Workshop 1 taught students about concrete materials, the making of concrete, the different properties of concrete composites, and their diverse applications through a set

of diverse activities developed by integrating all STEM subjects. The activities that were carried out to satisfy the workshop objectives are detailed hereafter (Figure 1):

- Activity 1: Comparing different kinds of cement—The workshop started with an ice-breaking experiment, wherein the students studied the different types of cement, one of the key ingredients in the concrete composite. They explored their hardening rates by measuring in different weights and analyzing the consequent results. They plotted graphs and interpreted them, thereby understanding the necessary science behind the process;
- Activity 2: Hunting for concrete objects—They researched different concrete applications, thereby increasing students' awareness of the most common infrastructure material. This activity was crucial in enhancing students' inquiry and research skills as they observe, question, learn, and deduce reasons behind using concrete materials;
- Activity 3: Comparing different concrete formulations—They familiarized themselves
 with the different ratios of ingredients/aggregates, i.e., sand, gravel, and cement, by
 experimenting with them to understand the properties of concrete, mainly density,
 hardness, and strength, according to the difference in ratios. They compared cement
 with concrete blocks and studied the packing of different aggregates to understand
 how each aggregate ratio affects concrete density;
- Activity 4: Reinforcing and testing concrete—The students grasped the effect of reinforcing materials on concrete properties, understanding their effect on strength and hardness. Moreover, they also tested the reinforced concrete blocks and compared them with non-reinforced concrete using strength-test apparatus, confirming the superior quality over the other.

Design project—All the above activities led the students to acquire the necessary knowledge in creating one of the sports products, i.e., a concrete bowling ball or a concrete floating boat, as a part of the design project. The workshop put forth design challenges for the students to choose the sports product for the design project and construct according to the preset criteria (refer to Table 2). The students interpreted the correct mixing-proportions for adequate sports product results through trial-and-error methods. The students also made careful decisions to meet the criteria, thereby choosing or preparing the correct casting mold, which will hold the concrete mixture.

The design criteria to create the respective sports products are elaborated hereafter:

1. Bowling ball—The group of students who chose to design a bowling ball considered certain conditions (refer to Table 2) to meet parameters such as surface smoothness, the ball's weight, and the distance traveled by it when rolled on a solid surface. The evaluators observed balls' surface smoothness by touching the balls and comparing them between that of each group to score the maximum. The entire construction of the bowling ball was wholly administered and carried out by the participants based on their judgment. They chose spherical mold and prepared concrete mixtures as per their choice of shape and ingredient proportions, although the facilitator pronounced the optimum ratio. They also performed casting, i.e., filled the molds with the fresh concrete mixtures and left them to dry for 1-2 weeks. Based on their judgment, they removed molds to carry out the curing process. The curing process is by which the concrete samples are hydrated to improve their strength. The students were solely responsible for deciding the curing time for concrete structures, which is an essential step in the construction because it affects the strength of the product. The students were also provided permission to smoothen the surface using any external tools such as a metal file rather than depending on the mold's texture, which naturally gives a smooth surface. This whole process was repeated if the end product broke or failed in meeting the diameter due to shrinkage. The ball was also dropped from a height multiple times to test its strength or success of the product. Once the ball met the required smoothness and weight, the test bowling ball was assessed by rolling through a V-shaped ramp to strike the bowling pins set across the lane. The wooden ramp was 1220 mm long and raised 400 mm on one end to achieve a reasonable ball

speed (refer to Figure 2a). The ball had to "strike" between two edges spaced 600 mm apart at the end of the 3000 mm long lane. The test product, positioned at the top of a V-shaped ramp with no dust, was propelled down by gravity's force at room temperature onto a flat-surfaced lane. The ramp was made from two rectangular slabs of plywood, 440 mm wide, placed at an angle of 40 degrees from the flat surface (refer to Figure 2a). The product ball was rolled in two orientations as arbitrarily marked "mostly perpendicular" on the ball. For each orientation, the ball was tried twice to pass through the strike zone successfully;

- 2. Concrete boat—The design product alternative to the student's choice of "concrete bowling ball" was the concrete boat. The concrete boat was expected to meet certain assessment conditions (refer to Table 2) in parameters such as buoyancy, load-carrying capacity, and float time. Similar to the bowling ball design, the participants either made a mold out of raw materials such as foam, cardboard, wood, etc. or chose a mold with an appropriate size that could be immersed in a rectangular box of $40 \times 70 \times 40 \text{ cm}^3$ filled three-quarter with water as a part of testing (refer to Figure 2f). The students went under similar conditions in choosing the correct amount of concrete aggregates, casting, and curing identical to that of the bowling ball. The testing of the boat, however, was conducted in the water-filled tank as mentioned earlier and was carried out in three phases as follows:
 - Assessment phase 1—The empty boat was placed in the water to examine its behavior under external factors; for example, the boat was tested to analyze its stability in the water and float freely for a minimum of 5 s. The students also ensured that the boat was self-balanced while floating on the water (without touching the sides of the water-filled tank/box) (refer to Figure 2c);
 - Assessment phase 2—It was assessed for another 60 s to monitor the internal parameters such as the concrete boat's quality in terms of leakage, water displacement, water absorption by the concrete walls, etc. While water leakage was observed visually, looking for traces of wetness in the interior of the boat, water displacement was monitored by examining the volume of the boat and the height by which the water rises in the tank. The interior of the boat was observed for wetness or moisture to test the absorption of concrete;
 - Assessment condition 3—The boat would be lastly tested for its load-carrying capacity. The initial cargo added to the boat was 5 g. Students incremented the weights only if the boat was in a stable condition without rocking to and fro in the tank. Cargo of 5 g was usually added in increments until the point when the vessel began to take on water. However, in some cases, students also used 10 g weights, depending upon the availability of weights. There was no minimum requirement for the cargo weight, and hence, the boat carrying maximum cargo without taking in water was considered for scoring the most in the score sheet. The leakages and similar defects observed in a few cases were fixed before the final trial for marking scores.

2.2.2. Workshop 2

The groups of students provided with Workshop 2 were monitored by the respective school to design a baseball or golf ball successfully. During the workshop, the students were familiarized with different sports materials and their properties, thereby cultivating their observational skills and analyzing capabilities. As the workshop progressed, the students performed hands-on experiments that provided a visual representation of scientific concepts such as friction, rebound, and energy transformations that are typically conveyed as regular school lectures. They were also acquainted with the different parameters that affected the functioning of the products. The various STEM activities (refer to Figure 1) carried out during the workshop is detailed as follows:

 Activity 1: Exploring rebound effect from different balls—As the workshop started with an ice-breaking activity, the students studied the rebound ability of different sports balls that were identical in shape and size, despite the materials, using a drop test (freely dropping from a specific height and measuring the rebound length);

- Activity 2: Hunting for different ball designs and materials—This activity, bolstering inquiry skills in high school participants, encouraged research on different sports materials such as cowhide, rubber, leather, etc. used in balls. Students studied different balls such as basketball, football, baseball, golf ball, etc. and hypothesized the purpose of each material as applied in a specific sports product. They also examined the design for each sports ball and compared its functionality. They interpreted the results mathematically in the form of graphs because STEM integration was encouraged throughout the activities.
- Activity 3: Investigating energy absorption of materials—During this activity, students
 familiarized themselves with energy absorption as they calculated the rebound height
 of a ball when impacted on different materials (floor tile, wooden board, carpet, felt)
 under freefall, thereby learning about energy absorption of the respective materials.
 Students also observed surface deformation in the case of different balls' impact. They
 also determined conclusions based on their acquired knowledge of the relationship
 between surface deformation and energy absorption;
- Activity 4: Comparing rolling friction on different surfaces—During this activity, students examined the behavior of different materials with respect to generating a frictional force on a rolling ball. Students tested different sports balls on each mat material one by one to draw out conclusions. They familiarized themselves with various frictional forces—kinetic, rolling, and static. They also interpreted results drawn from rolling ball surface and mat surface to be distinct from each other, and rolling action entirely depended on the materials.

Design project—This workshop focused on challenging the students to design sports products such as baseball and golf ball with materials that can provide the optimum results to meet several criteria (refer to Table 2). Since they invested a productive period for researching and understanding different materials during the workshop, they also performed binding and arranging different materials to obtain the desired outcomes, as in golf ball and baseball. The students evaluated their products' strengths and drawbacks, keeping records of every detail during their project work.

- Baseball—The students who chose to design a baseball needed to overlook the evaluation criteria such as specificity in its diameter, weight, and the maximum ability to rebound on free fall. Since the regularity in the diameter of a handmade ball could not be guaranteed, the test ball diameter was measured using a Vernier scale (refer to Figure 2g). They transferred the knowledge acquired from the workshops to choose varied materials that could satisfy the requisites while testing the product balls. During the testing, the test baseball was freely dropped from a specific height of 1 m to measure the rebound height;
- Golf ball—The golf ball challenge was staged in two sections, i.e., constructing the golf course pit and designing the golf ball for the students who chose the golf ball design project.

The students initially constructed a golf pit using synthetic materials considering rolling friction, rebound effect, and impact energy (refer to Figure 2f). They employed different materials, such as felt and synthetic grass in the pit area, to restrict a golf ball's rolling within a specific length (refer to Table 2).

While designing the golf ball, the participants considered different materials according to their behavior to accomplish specific scientific concepts that could satisfy the preset conditions (refer to Table 2). They chose the materials that led to building a smaller diameter ball with a specific weight producing lower rebound ability when compared to that of a baseball.

While testing, the ball was rolled down the tube (PVC tube, cut to 30 cm length) and placed fixed at an angle of 30 degrees (as shown in Figure 2e). The students determined the success of the ball, depending upon its ability to bounce once and clear the obstacle,



roll across the green area, and stop within the target area (refer to Figure 2e). The test was repeated enough to obtain a 75% success rate for the respective product, i.e., out of four test attempts, the ball should be successful for three tries.

Figure 2. Schematic diagram of testing the different sports products. (a) An elevation view of the testing pits for the concrete bowling ball. (b) The front view of a schematic representation of the front view of a bowling ball testing kit. (c) The schematic representation of testing a concrete boat in a tank, three-quarter filled with water. (d) The concrete boat should be able to carry weight without sinking or any leakage. (e) The schematic diagram for the testing pit of a golf ball. The students prepare a golf course pit using different materials to test the rolling action of the golf ball. (f) A golf course is set up by the students to test their product golf balls considering the different parameters such as friction, rebound effect, etc. (g) A baseball diameter is tested using a Vernier scale.

Product.	Parameter	Requirement
	Surface smoothness	Smooth surface
Concrete Bowling ball	Weight of the ball	Less than 5.5 Kg
	Distance traveled	3 m and hit the pins as shown in Figure 2b
	Buoyancy without weights (External factors)	Floating < 5 s
Concrete Boat	Buoyancy without weights (Internal factors)	Floating $< 60 \text{ s}$
	Loading capacity	Maximum weights
	Diameter	7.3 cm to 7.6 cm
Baseball	Weight	142 g to 149 g
	Ability to rebound	Maximum
	Diameter	Between 4.5 and 5.5 cm
Golf ball	Ability to rebound	Bounce once and clear the obstacle
	Ability to roll and stop (Friction)	Stop at target area

Table 2. The list of parameters that need to be satisfied for excellent results.

2.3. Data Collection Methods

Different forms of qualitative data were collected during the program, including facilitator observations, field notes, media (photos and videos), and student artifacts. The facilitator recorded daily observations immediately after the session that detailed student regular attendance, their learning outcomes, or cognitive skills acquired during each workshop session, and progress in their daily project work. She also recorded progress

and obstacles during each workshop day, during which field notes, images, and videos were also taken. Questionnaires, as a quantitative data-collecting tool, were provided to the students before and after the workshop, later analyzed to understand the improvement in their attitudes toward STEM fields, the relevance, and STEM-driven careers. The questionnaires were adapted from previous research studies but were modified according to the study objectives [24]. Student artifacts/work that comprised of project presentations, sports products, and documentation video were also collected and presented as a part of a program-adjourning event, which was later utilized for the study. The two primary program evaluation tools that were implemented in data analysis are detailed below.

2.3.1. Pre and Post Questionnaires

The students were obliged to complete a set of study questionnaires (refer to Table S1 in Supplementary Materials) by pen and paper method to assess their attitude on science, technology, engineering, and mathematics (STEM) and its impact in the real world. The questionnaires consisted of multiple-choice questions that stemmed from the student opinion on STEM education, their career interests in scientific fields, and their skills. The students may recommend from four standards of agreement: "agree", "neutral", "disagree," and "do not know". The questionnaires were assessed using standard monitoring and evaluation (M&E) procedures. The mean percentages of agreement of pre and post questionnaire results were compared in the procedure to analyze the outcome of the program. The questionnaire analysis used the Likert Scale (R. Likert, 1932), a one-dimensional psychometric response scale measuring a single trait implemented in questionnaires to achieve the participants' degree of agreement to a statement or set of statements. We used a 3-point scale ranging from "Agree" on one end to "Disagree" on the other with "Neutral" in the middle.

The questionnaires were pivotal in measuring their agreement on the importance of STEM fields, career aspirations, and their views toward independently performing a scientific experiment. Eventually, both the post and pre questionnaires were analyzed together to obtain the results.

2.3.2. Evaluation of Project Presentation

After completion, the products would be examined by external evaluators from sports facilities, academia, and leading economy driving industries. This evaluation was performed during a conference for two days to felicitate the student work. The evaluation was carried out in two stages—day 1 and day 2. On day 1, all the program participants were evaluated on their STEM competencies acquired through the sports product engineering project. They presented their project experience in a PowerPoint presentation, implementing digitalization in the process, and self-mentoring themselves in preparing for the evaluators' queries. The project presentation provided the students with opportunities to showcase their efforts. The examiners also challenged the students with questions that could assess their critical thinking and future outlook on their products.

Although each product had its respective testing criteria, as shown in Table 2, the evaluators also examined for other criteria, which were common to every design project. They had a separate rubric (refer to Table S2 in Supplementary Materials) for each student group presentation to rate the students' organizational ability, self-confidence, research methods employed for the design project, and critical thinking. The students were also rated for the quality of slides that were displayed as a part of their presentation. They were also scored for the students' creative display for choosing product mold, materials, and the quality of work exhibited in accomplishing each task.

The external examiners were from different backgrounds to perform a fair and overall assessment of the student work. For each design project, each sports product was examined on day 1 of the conference by a set of three judges. While one judge who held PhD qualifications and served as an academic in a university assessed the depth of scientific details and principles applied in the product, the other two, from the industry with

backgrounds in engineering, focused on the presentation quality and the scope of the product in the future.

On the other hand, day 2 of the conference witnessed the participation of selected students who scored the maximum on day 1 evaluation. Only one group of participants from each category of sports product presented before the judges, who, this time, were from a wide range of fields that even included sports science specialty. There were five judges, one from academia, one from a sports training facility, and the other three from leading oil and gas industries. The first two judges were focused on science and sports, exploring the scientific knowledge and application of students. The latter three who hailed from the training and development department of heavy industries evaluated students' competency display because they were more focused on the STEM skill development and presentation of the national participants.

3. Results

In this section, we lay out findings that include summaries of the journey of the students with each sports product based on the observation made by the facilitator during each session with the student. The findings drawn below are also based on their PowerPoint presentation from participant artifacts during their course-adjourning event. The students elaborated on their project journey and the number of trials and failures, providing the examiners the opportunity to picture the complete journey in the PowerPoint presentations. We also provided results drawn from statistical analysis of pre and post questionnaires that demonstrate the student attitudes toward STEM-driven careers and the importance of STEM in daily lives. According to conclusions in PowerPoint presentations, the participant students recorded their complete attempts, their reason for failure, and their successful attempts. These records were crucial in analyzing their problem-solving skills that provided adequate backings for facilitator observations.

3.1. Analysis of Sports Products

3.1.1. Design Product 1

Design product 1 chosen by a few participants of the concrete workshop was a concrete bowling ball that could roll on a solid dust-free surface and strike the pins set at the end of the course. The participants successfully constructed and designed bowling balls that met different criteria, such as surface smoothness, specific weight, and distance rolled, for the final evaluation (refer to Figure 3a). Since surface smoothness was one of the three requisites for successful completion as in Table 2, they chose suitable molds for casting concrete mixture that could provide a smooth texture, appropriate size, weight, and shape for the product. The design of the mold also influenced the resulting functioning of the product. The application of the right mold for casting a concrete mixture requires fine techniques and an understanding of texture, physical strength, and measurement. The students were ensured to choose a mold that can hold heavy concrete mixtures and can be easily broken once the concrete is ready. While some students used volleyball as a mold, most of the students used spherical light bulbs made from plastic or glass.



Figure 3. Examples of students' sports products that were displayed during the program- adjourning event. (a) A sample concrete boat is made as an engineering design challenge by the students applying knowledge acquired. (b) The students applying knowledge acquired during the concrete workshop session make a sample concrete bowling ball as an engineering design challenge. (c) A golf ball made by the participant groups is illustrated through the images. The students transfer knowledge acquired during the sports material workshop session by making a sample golf ball as an engineering design challenge. (d) Different samples of baseballs as developed by participant students. The latter images also showcase the binding of the materials in different patterns.

Another criterion of specific weight was also met by the students' different measures to accomplish the task. For appropriate weights, students considered different reinforcing materials that could contribute to balancing the weight and strength, and quality of the product (refer to Figure S1 in Supplementary Materials). Moreover, restricting the weight of the ball according to the evaluation criteria was a major challenge as they repeatedly tested various combinations of concrete mixture ingredients to obtain the desired weight within the specific diameter. It was completely students' choice to fix the composition of concrete ingredients, and thus, they had to ensure that they had enough concrete to fill the mold.

Since the students were faced with multiple conditions, i.e., restricting weight as in Table 2 and maintaining a spherical shape simultaneously, the assignment nurtured their thinking capacity and solving capabilities, in turn enhancing their critical thinking and problem-solving skills. The students also familiarized themselves with the impact and different techniques for proper mixing, as in the boat challenge, to obtain the desired texture and consistency without any lumps. The finished product was tested by dropping freely down the ramp, set 3 m away from the bowling pins (refer to Figure 2b). While rolling the ball on the ground, the students identified the possible flaws in the product that might have resulted from improper texture, weight, or shape depending on the rolling action of the balls. The testing of bowling balls fascinated the students as they indulged in "play and learn," thereby engaging actively in the program.

Table 3 implicates the course of two random groups (1a, 1b) who applied the trialand-error method to create the final successful product, which was obtained from student records in PowerPoint presentations. This table also provides insight into developing their reasoning and problem-solving tactics as they carried different strategies to accomplish the task. They were also challenged while testing their product, concrete ball, by dropping the finished product freely down a ramp set 3 m away from the bowling pins (refer to Figure S1 in Supplementary Materials).

Sample Groups	Sample Products	Trial Count	Reason of Failure (Refer to Table 2 for Criteria)	Reason for Success
Group 1a		two attempts	The mold used was big.	Changed the mold to make a smaller ball
Group 1b	Concrete bowling ball	four attempts	 The size of the ball does not meet the competition's requirements. The product was heavier than the weight limit, which is 5.5 kg and had many holes that were difficult to remove. Adding the paper balls made it easier to break as they had a lot of cracks and took a lot of space, which made the bowling ball weaker; after several tests, It got broken. 	Added a plastic ball at the core that balanced the weight required keeping the diameter under the condition.

Table 3. The information on two random groups attempting design product 1.

The students assigned with the bowling ball challenge hardly had opportunities to vary the design of their product regarding the shape, owing to the fixed spherical shape and diameter, in contrast to the concrete boat challenge. Hence, most of their creativity was exhibited in designing the final product. They gave careful consideration in painting their products (Figure S1 in Supplementary Materials) and presenting it to the judges as creativity was one of the criteria in the rubric (refer to Table S2 in Supplementary Materials) set for them.

3.1.2. Design Product 2

Design product 2, chosen by some of the concrete workshop participants, was a concrete boat capable of floating and carrying cargo. The students exhibited a wide caliber of proficiency while designing boats that satisfied the buoyancy and load carrying capacity (refer to Figure 3b). They were examined for their product quality in terms of scientific application and various skillset as they paced from the basic steps of the engineering challenge to successfully presenting a fully functional product. The resulting boats also supported the expected quality of concrete that displayed efficiency in terms of parameters such as workability, durability, strength, cost, and finished appearance. The quality assurance of the product was solely attributed to the student competencies, which ensured not to overlook factors such as the ratio of ingredients, mixing of the ingredients, handling of the fresh concrete, and finally, placing for curing until the ultimate strength was achieved (refer to Figure S2 in Supplementary Materials). Once the concrete boats exhibited fine strength and texture, it was tested by floating in water-filled tanks. In the case of failures such as leakage or inability to float or/and carry weight, the product was subjected to reconstruction or improvisations and retesting.

The students also exhibited remarkable problem-solving skills as they followed a trial-and-error method to construct the sports product and optimize the experiment in a scientific method. From Table 4, it was identifiable that they were able to recognize the reasons for failure and success distinctly, for example, cracks, leakages, inability to float, or even concrete collapsing on mold dismantling. Some groups of students attempted several trials, ranging from a minimum of two to an utmost nine trials, in accomplishing the nearest best solution. Table 4 provides the course of two random groups, 2a and 2b, with their number of attempts and improvisations as represented in PowerPoint presentations. Eventually, this concrete boat challenge was also suitable for drawing their creativity out. It encouraged the students to display artistic talent in addition to STEM learning while sketching their product structure or painting the finished product (refer to Figure S2 in Supplementary Materials).

Sample Groups	Sample Products	Trial Count	Reason of Failure (Refer Table 2 for Criteria)	Reason for Success	
Group 2a	Concrete Boat	three attempts	 Amount of Cement is less than the fly ash and silica. The boat design was not uniform 	 The ratio of concrete materials was appropriate. The design was modified, and the boat floated and held weight. 	
Group 2b		three attempts	 The ratio of concrete aggregates was not appropriate. The reinforcing material for the concrete could not hold as the group tried a green concrete mixture. 	Succeeded in the right ratio of materials.	

Table 4. The information on two random groups attempting design product 2.

3.1.3. Design Product 3

Design product 3, chosen by some of the sports materials workshop students, was a golf ball conditioned to pass an obstacle, rolling down the course to stop at a certain point. The golf ball challenge was carried out in two distinct phases: Phase 1—constructing the golf course pit and Phase 2—designing a golf ball.

- Phase 1—The students designed and constructed a miniature golf course pit (refer to Figure S3 in Supplementary Materials) so that they could test their product golf balls. They set up a course pit, as shown in Figure 2e, according to the required parameters and implemented synthetic materials that contributed to the effective testing of the sports products. The students also applied the knowledge acquired during the sports materials workshop in successfully designing and building the pit. They constructed the course pit using felt and synthetic grass sheet made from polypropylene, polyethylene, and nylon. They were successful in restricting their product balls within the specified length and width of the designed course. While constructing the course pit, they used a standard golf ball for testing the course pit in meeting the adequate condition. The students tested different materials to lay on the pit and rolled the ball down the PVC tube to pass the obstacle, as shown in Figure 2e. The expected functionality of the laid materials was to restrict the rolling of the ball to a specific length of 1.25 m, as shown in Figure 2b.
- Phase 2—The students successfully designed and tested golf balls (refer to Figure 3c) that were able to satisfy the preset conditions (refer to Table 2). They chose optimum materials that contributed to accomplishing their design criteria, specifically balancing the required rebound ability to cross the obstacle and the desired diameter. Evaluating their techniques, the students also considered the efficiency of the ball's binding process as crucial in obtaining the best results. If the ball was not tightly bound, the rebound results may vary due to the irregularity in its shape and uneven energy transfer. The students also experimented in binding and varying techniques resulting in varying outcomes, as shown in Figure 3c. They studied and applied different scientific principles such as the energy of absorption of different materials when subjected to impact on solid ground.

Resultantly, students applied trial-and-error methods to solve their task, as similar to the other three design product challenges (refer to Table 5), which were retrieved from PowerPoint presentations. They could clearly identify the shortcomings, assess their product, and apply the acquired knowledge to create a successful outcome. In some cases, the students also reattempted to improve the results of the product, even though the first attempt did clear the task. This also detailed their attitude of creating room for continuous self-improvement. In due process, they also bolstered inquiry skills as they continually asked questions such as, "what are the factors" or "what are the materials" for deriving a successful output. Table 5 implicates the course of two random groups who applied the trial-and-error method to create the final successful product.

Sample Groups	Sample Products	Trial Count	Reason of Failure (Refer to Table 2 for Criteria)	Reason for Success
Group 3a	Golf ball	two attempts	The rebound was not enough to jump the obstacle (refer to Figure 2e)	A rubber core was made using glue to create the bouncy effect without increasing weight
Group 3b		two attempts	The golf ball made during the first attempt was considered a failure because the students made a better product that gave better results	Used different ratios of the materials.

Table 5. The information on two random groups attempting design product 3.

The students' creativity was also tested in their choice of materials (as in Figure 3c) used in designing the balls to obtain the desired weight, diameter, and rebound properties. They experimented in applying different materials, such as glue, rubber bands, bouncy balls, etc., to construct either the outer layers or the inner core material, thereby testing their limits to creative imagination.

3.1.4. Design Product 4

Design product 4, chosen by some group of Workshop 2 participants, was a baseball exhibiting a considerable rebound height. The students were focused on choosing materials that contributed to accomplishing the design criteria, specifically the rebound ability. They sandwiched different materials, such as rubber, yarn, fabric, etc., depending upon their weight and rebound ability to construct a ball structure. The order of placing different materials was also varied, looking for the best rebound effect. In some cases, they varied the usage of rubber material, for example, they used a rubber ball as the inner core as well wrapped rubber bands outside to form the outer core. In some cases, they also put forth innovative ideas such as a plastic mini ball as a core, contributing to less weight and best rebound results under restricted diameter. The students also applied techniques while binding the materials that contributed to the product's different properties (refer to Figure 3d) such as spherical shape and rebound effect, etc. A loosely bound ball was observed to rebound less when compared to a tightly bound ball as the ball deformation was varied on-ground impact and energy was distributed unevenly. These observations inspired students to understand basic concepts of science in a better way. The students tested their completed baseball products (refer to Figure 3d) by freely dropping them from a height of 1 m, thereby addressing the product's possible flaws. The students also considered the scientific facts, e.g., how the efficiency of binding contributes to the fineness of the ball surface. The participants also tried different binding patterns and materials, as shown in Figure 3d, which influenced the outcome of each product.

The students also solved their challenges in obtaining a balanced baseball with the desired weight and diameter by implementing the trial-and-error method (refer to Table 6, according to the conclusions from PowerPoint presentations). Moreover, as they introduced unique solutions to solve the problem, portraying the creative side of their cognizance, they referred to materials such as cowhide, yarn, etc. that could help solve the task drawing out the creative collaboration within the group. Table 6 implicates the details of two random groups who applied the trial-and-error method to create the final successful product.

Table 6. The information on two random groups attempting design product 4.

Sample Groups	Sample Products	Trial Count	Reason of Failure (Refer to Table 2 for Criteria)	Reason for Success	
Group 4a	Baseball	three attempts	The bouncy ball (the core) was small, so the diameter was big with the right weight. The final surface was not regular and was not smooth enough.	Changed materials by using a bigge bouncy ball, cowhide, yarns, and glu	
Group 4b		two attempts	The rebound was less.	The binding of the materials was made stronger.	

3.2. Demonstration and Presentation of Design Projects

A conference was organized at the end of the program as an adjourning event for two days, displaying and examining the students' final products in the presence of an evaluation panel. The panel included leaders from industries (refer to Figure 4) in the STEM domains. Day 1 witnessed all groups' participation, portraying their program journey in PowerPoint presentations, videos, and photographs. They presented their products in the presence of the judges to demonstrate their accomplishments. On day 2, three winning groups (one from each sports product category) displayed their products to a different set of panels that comprised leaders from the best sports facility and community in Qatar.



(a)

(b)

Figure 4. Students displaying their products for evaluation. Students in the image (**a**) are testing their concrete boat by incrementing the weights in the boat thereby testing its load-carrying capacity. (**b**) Bowling pins are laid 3 m away from the ramp for the evaluation of the bowling ball.

Before the evaluation was held on a public platform in the presence of a larger audience, the students improved their oration, body language, public speaking confidence, and vocabulary. As they presented their products to the panel, the students displayed their presentation skills along with the communication dispositions. The participants were assessed critically for their organizational ability, research methods implementation, and creative collaboration as they presented their project experience. They integrally replied to the judges' queries with confidence, proving their skill-based improvement.

3.3. Descriptive Statistical Analysis

Analysis of the study was quantitatively carried out by conducting pre- and postworkshop surveys. The study clearly manifested the increment in students' interest in an interactive STEM curriculum. The analysis also reported considerable growth in their self-efficacy and innovative skills along with their positive inclination toward STEM, both generally and career wise. The pre–post questionnaire analysis of four objective study indicators directed toward students' STEM attitudes was demonstrated, as in Figure 5 and Table 7. The students' pre- and post-questionnaire results were statistically calculated and analyzed using the *t*-test statistical calculator to determine the program effectiveness indicators (*p*-value, t value, mean differences). The *p*-value for the *t*-test conducted for all five cycles was less than 0.05, as in ideal condition (except for the fourth cycle p = 0.229). Hence, the results indicate that there was a significant difference between the two sets of data means (pre and post means), concluding the considerable improvement of their attitudes after the program (refer to Table 7 and Figure 5, as well as detailed Tables S3 to S12 in Supplementary Materials). In addition, considering the t value for the confidence interval ($t \le 2.262$ to reject the null hypothesis), there was strong evidence (except for the fourth cycle t = 1.22) to validate that the experiment falls under the range of 95% confidence interval (95% CI). The 95% confidence interval (95% CI) holds a key factor in determining the program's constructive role in the student outlooks toward STEM. For the given set of data, the outcome improved scores, on average, by an approximate value of 0.8. The following STEM interest indicators were used to perform the *t*-test analysis:

- We live in a better world because of science, technology, engineering, and math (STEM);
- Learning science, technology, engineering, and math (STEM) is essential for my future success;
- I would like to have a career as a scientist or researcher;



• I have the skills to implement a scientific experiment.

Figure 5. Normalized pre-test and post-test mean difference comparison of the student population for five cycles (n = 248). The blue line and red line display the pre-test mean and post-test mean percentages of the questionnaire analysis data collected from the five-year experiment, respectively. It is evident that despite the results drawn from the fourth cycle, the program proves to be effective in improving students' attitudes by a significant margin.

Table 7. "Science in Sports" (SIS) cycles by *t*-test analysis (n = 248). The "n" for each cycle is less than the initial number as stated in Table 1, as the analysis is performed only for participants who have performed both pre-test and post-test questionnaires.

Cycle	Statistics			Sample Tests		
-)	Participants N	Pretest Mean (SD)	Posttest Mean (SD)	Mean Diff (SD)	t Value	<i>p</i> -Value
1st	51	3.07 (0.932)	3.79 (0.865)	0.730 (1.185)	4.4	0.000
2nd	56	3.38 (1.205)	3.98 (1.032)	0.603 (1.680)	2.68	0.01
3rd	35	2.98 (0.570)	3.81 (0.900)	0.836 (0.889)	5.56	0.000
4th	52	3.46 (0.865)	3.66 (0.913)	0.202 (1.195)	1.22	0.229
5th	54	2.31 (0.746)	3.03 (0.693)	0.722 (0.757)	7.02	0.000

While analyzing the data of student attitude differences between pre-program and post-program, we observed an increasing trend line from cycle 1 to cycle 5. However, we noticed that the dips in the graph areas for cycle 4 and cycle 3 were because those data correspond to students from urban schools. We deduced that students in urban schools are already familiarized with STEM relevance and this had an impact on having less difference between pre- and post-program attitudes. They initially developed high individual opinions toward STEM fields. However, as we started experimenting the study on students from rural areas, we observed that there exists a clear distinction of improvement in their attitudes post program.

4. Discussion

The sports-based program has conferred productive insight into students' cognitive skills development and their attitudes toward STEM domains and aspirations. The SIS program has positively influenced participant students' cognitive development because they successfully designed their sports products. This section explored program outcomes, the improvement in students' learning, sense of efficacy, and the ability to solve problems. The three forms of data for evidence in assessing STEM workshop outcomes, design products, and strength, weakness, opportunities, and threats (SWOT) analysis are the following: (a) workshop outcomes (based on facilitator observation); (b) facilitator's assessments of students' work (design products); and (c) demonstrations of student dispositions (e.g., presentations, facilitator observations, school teachers). We hereafter discuss the outcomes in detail.

4.1. STEM Workshop Outcomes

The STEM workshop conducted during the one week witnessed students' active learning involvement in various sports-driven activities cultivating inquiry, research, logical reasoning, and creativity. The research-directed activity was successfully administered to implement digital tools in their inquiry-based learning process, improving their scientific knowledge in the subject context and productively implementing technology for learning. Experiencing the various hands-on scientific activities, the students enhanced their technical knowledge in terms of precision, accuracy, and measurement, later applied in the design project challenge. The diversity in the scientific experiments and vitality of the sports offered the students positive settings, henceforth cultivating curiosity, creative engagement, and appreciation toward science. The students were seen to be entirely fascinated by exploring the interconnection between sports and engineering, hence engaging keenly in understanding the underlying scientific concepts in sports and thereby addressing STEM learning. The activities also helped them apply different strategies and effectively transfer the acquired knowledge from the workshops to meet the engineering design challenge criteria. The four activities from Workshop 1 were crucial in developing scientific interest because they explored the scientific subject content aiming toward a design goal in informal learning set up [28,29]. As the students learned the necessary knowledge to construct a fail-proof concrete object, they were experiencing learning through hands-on activities similar to the studies conducted prior [22]. For the design project, they integrated scientific content such as floating conditions, Archimedes principle, rolling friction, and frictional interaction between bowling lane surface and ball surface from their regular school lessons into the workshop outcomes to design the ultimate sports product [30].

On the other hand, Workshop 2 acquainted them with visual information and handson experience on the rebound behavior and coefficient of restitution, included in their school physics curriculum, however, through the introduction of sports products such as baseball, basketball, etc., which was a relatively new experience. Workshop 2 also provided a novel approach to teaching the energy absorption of different materials such as wood, cow-hide leather, rubber, yarn, etc. through a hands-on approach, which was easier for the students to apply in their design project. The cognitive development of the students was the main focus of designing and constructing the sports product. On cognitive measures, students enhance their fundamental thinking skills such as inquiry, hypothetic deductive reasoning, comparing and contrasting, and question generating similar to that of workshop activities as observed by previous researchers [25,31]. These skills were embedded in the sports design products, wherein the students nurtured them as they needed to make choices that led to accomplishing their tasks, such as choosing different weight compositions of concrete ingredients or choosing viable materials. Even while choosing the materials, there were different levels of decision making they had to partake, considering weight or surface absorption or any other parameters. Many of the researchers employ these methods in science teaching under the experiential learning process [22,32,33]. The evidence demonstrating the above conclusive results was portrayed in the Results Section analyzing different sports products.

While examining the workshop outcomes and addressing RQ 1 and 2, the *t*-test analysis was conducted on the 248 participants, providing mathematical representations of the difference in the students' attitudes displayed pre and post workshop. *T*-test was conducted on five batches of students, who participated in the SIS program in consecutive years (2012–2017). As shown in Figure 5, a positively increasing trend line displaying the significant difference between pre-test mean and post-test mean shows the effect of similar STEM mentoring programs, capable of improving the STEM preferences in school students. In addition, the trend shows that the study results fall under a 95 % confidence interval, validating the success of the experiment. The outcomes from the success of the product development and application of different STEM methods along with the statistical representations from Table 7 clearly directed toward the RQs, 1 and 2, "Did the student understand the importance of STEM by experiencing science through sports product engineering?" and "Will the student consider STEM driven career resultantly in the future?", respectively. Their future aspirations were dominantly influenced by the program, as demonstrated in Table 7.

4.2. A Facilitator's Outlook on Student Journey with the Design Products

The facilitator keenly observed the growth of students and their dispositions from the first day of the workshop. The field notes and daily reports on the different student groups provided detailed analysis results on their display of diverse STEM dispositions such as problem solving and logical reasoning. Moreover, the facilitator also noticed the progress in their self-confidence as they creatively collaborated with peers and mentors (facilitator and schoolteacher). Their journey with reference to the design product is detailed hereafter.

The facilitator recognized the improvement in students' learning behavior while analyzing the different reasons for failures during the construction of the bowling ball. The students were able to assess the variant flaws accurately either during mold selection, product finishing (removing mold from the dried concrete product), or product testing. In case of unsuccessful attempts, they observed that the students could attribute the cause of failure to valid scientific reasons. They observed that students frequently burst with quick queries, "why" and "what if" much more often than the regular workshop days because they were impatient to solve their challenge.

As the students decided the mold (shape), proportion, and weight of concrete ingredients (appropriate weight), and finish the final product with adequate testing, the facilitator observed pride and satisfaction in achieving a target successfully. It was observed that some of the groups failed during the first attempt due to poor mold choices. For example, while choosing the mold, they attempted to introduce novel ideas such as paper mold, which did not have the strength to hold the concrete mixture, thereby collapsing the end product (as shown in Figure S1 in Supplementary Materials). Apart from choosing mold and making decisions on proper aggregate composition, the groups considered adequate atmospheric conditions to ensure the finished concrete models' essential curing. They performed curing to prolong the hydration (chemical reaction with water molecules) of cement in the concrete structure to develop calcium hydrate silicate gel, a strong binding material to reduce porosity and increase concrete density. This process was crucial in preventing surface crumble or sample breakage. Students also acknowledged the reason for failure in some cases due to breakage or cracks as the lack of providing adequate curing time. The main goal achieved in leading the students through the activities was to ensure that they acquire scientific knowledge in practice rather than a one-sided instructional delivery [34]. These experiences were acting as intrinsic motivators for the students' active participation [20,35]. In due process, the students also invested in learning the scientific principles from sports product engineering, which was an additional practice for the students. They investigated different factors that affected the desired level of "skid-hook-roll" action during the testing process, which was crucial in hitting the pins (Freeman and Hatfield, 2018). It was also observed that the testing of the products offered a consolidated "play-and-learn" opportunity to the participants, enhancing their interest in STEM integration in their regular school driven science concepts. As in the case of bowling ball, they acquainted themselves on the scientific principles such as rolling friction and frictional interaction between lane surface and ball surface, improved by polishing the ball surface to a "glossy" finish, which guaranteed a longer skid and a sharper hook in the back end for the best results (Freeman and Hatfield, 2018). They also correlated the concept of friction, inelastic deformation, and energy conservation (the kinetic energy of the rolling ball was converted in contact with the pins), and as in the case of concrete boats, buoyancy, density, and material compactness were visually experienced through this learning method.

The students were provided with experience studying and testing multiple materials, such as concrete materials, wood, carpet padding, felt, Astroturf, foam rubber, acoustic tiles, linoleum, shag carpet, etc., in laying out the course pit material or choosing materials, as in the baseball challenge. Since this method of bringing the students into the experimental field through an integrated STEM approach and providing opportunities for intermingling with peers successfully enhances STEM literacy, the facilitator found an enhanced level of motivation in the students to address the challenges [30]. Participants also had a wide experience while experimenting with yarn, rubber bands and shreds, felt, fabric scraps, aluminum foil, glue, etc. while constructing a ball that could deliver the desired rebound height and rolling action.

Interestingly, this study witnessed the effective implementation of mathematical concepts such as applied geometry, measurement, ratio, and proportion for studying the physical behavior of concrete under varying conditions, with STEM learning being integrated into the procedure, as effective as in previous studies [5,29,30]. The students applied different mathematical expressions that include density (D), where D = mass/volume, which was relevant to sketch the shape of the boat with increased length, breadth, height ratio, thereby increasing the volume to be greater than the mass of the boat. They learned and applied the coefficient of rolling friction (CR) and coefficient of restitution (COR) formulas in the case of products that involved ball action (baseball, golfball and bowling ball); COR = $\sqrt{\text{height of rebound}/\sqrt{\text{drop height and CR}}}$, $\mu = \text{rebound height/distance}$ traveled by the ball.

The facilitator put forward the challenge of building the test golf course pit to the entire batch of participants in the respective school to exploit their problem-solving skills, knowledge transfer capability, teamwork, leadership, and creative collaboration. However, because this was a collaborative endeavor, it was easy for the facilitator to understand the students' development in leadership and peer collaboration. The golf ball challenge provides an excellent "play-and-learn" opportunity for the students because they learned basic science concepts such as rolling friction, rebound, and energy absorption of materials through integrated STEM-driven plays. They explored the interplay between sports and science, attracting the students and influencing their desire for science and STEM subjects. The design challenge has played an imperative role in amplifying science's perception as a positive, "fun" experience. These observations made by the facilitator based on the validated responses from the students addressed the RQ3 research question, "Were the students able to understand and correlate scientific principles to the applications in daily

life?". The workshops provided ample grounds for the students to understand the scientific principles while the students employed diverse tactics to solve the problems from a STEM perspective during the trial-and-error method.

The facilitator recalled that students fostered their creativity from effective peer collaboration, apart from critical thinking and reasoning skills as they designed each product's shape and dimensions, reinforcing previous studies [34]. They also noticed that the students, in the process, gained learning outcomes in regard to the application of scientific concepts such as in sports-based real-life scenarios. Creative designing was observed as the students had to limit boat size to be immersed in the testing tank. These creative expressions were also applied during the construction of the mold because some participant groups constructed the mold for the boat from raw materials such as foam sheet, cardboard, or wood (refer to Figure S2 in Supplementary Materials), while the other groups opted for readily available toy boat molds (refer to Figure S2 in Supplementary Materials).

4.3. Discussion on Schoolteachers' Observations

The qualitative responses from schoolteachers and students substantiate their enthusiasm for the program. Teachers especially liked the variety of inquiry-based activities and resources provided in the workshop. They observed positive responses in their students with increased analytical thinking. Teachers also regarded the workshops and design challenges as an inspiring experience that nurtured students' interest in STEM. School teachers also validated that the students were enthusiastic during hands-on activities because they were involved in constructing sports products [36,37]. Due to their cultural orientation toward sports, participants also clearly exhibited their enthusiasm for sports-based engineering assignments. Male participants typically showed less interest in writing activities, as recalled by the teacher, in contrast to hands-on activities that included measurement, preparing concrete mixtures, and testing the products. This type of behavioral differences by different validating researches [12,38]. They generally were motivated during teamwork to work zealously on the products and even collaborated cordially with the facilitator.

4.4. Strength, Weakness, Opportunities, and Threats (SWOT) Analysis

The facilitator well examined the experimental methods and results to develop a SWOT analysis matrix. The SWOT analysis laid out the key observations that pointed out strengths, weaknesses, and future possibilities in the study.

4.4.1. Strengths and Weaknesses

The layout and execution of activities that led the students to design a sports product effectively were pointed out to be a strength. The layout of activities clearly manifested the necessary aid for the students in accomplishing their target objectives. The inquiry-based methodology in the activities opening into a trial-and-error experimental method in satisfying a set of conditions led the students to examine their project from their perspectives, thereby elaborating their thinking capability. In addition, the study methods' creative collaboration was also considered a plus point for the research study.

While assessing the weaknesses, it was observed that the experiment chose a random selection of sports products that were easy to construct in terms of high school students' competency. Although the study was conducted through a well-laid out activity plan leading to design challenges, the relationship between the four sports products was not clearly established. However, this shortcoming was not addressed before the students because they were introduced to a STEM workshop, which by the end, led to the design of two sports products that fall under the same category. The weakness that was highlighted by the facilitator was its limitation in assessing the student competencies using an internationally standardized testing rubric, which will be further considered for future research.

4.4.2. Opportunities and Threats

Taking into account the positive results observed, the study has opened windows for improvement in terms of adding new activities to Workshop 2 that was capable of acquainting students with knowledge on aerodynamics. This new addition could open better sports products that could promise superior quality and educate the students on concepts such as terminal velocity, which is remarkably addressed in aviation and space technology fields. Moreover, the facilitator was very much looking forward to opening pathways for the student to design a new sports equipment/game applying the scientific concepts that were introduced in the activities of Workshop 2. Although the workshop provided ample opportunities for identifying and learning different materials, their physical properties, and their behavior under the limited school laboratory testing conditions, it promises a future scope for experimenting state of art testing conditions such as scanning electron microscopy (SEM) for better results. Resultantly, tests such as SEM analysis will be able to test the water absorption of concrete walls and concrete composition to microscopic detail. Drazan et al. [39-42] experimented in basketball events, exploring different perspectives of implementing science and engineering curriculum. The studies provide us with a solid background, thereby demonstrating opportunities for students in designing performance monitoring devices by analyzing the player performance. The students can relate to the application of analytics in real-life scenarios through similar additional projects.

Threats were minimally observed while assessing the SIS program; however, the facilitator noticed that the students were exhibiting behavioral setbacks while acknowledging failures. Since the design project was carried out through trial-and-error methods, a few students showed impatience and reluctance in moving forward during lapses in the experiment. Moreover, few students did withdraw from the program, which seemed to affect the morale of their peers, causing further setbacks. Meanwhile, the facilitator ensured that the remaining group members were always motivated with positive remarks and applause to heighten their interest.

5. Conclusions

Sports, being one of the critical motivators that contribute toward upbringing socially intimidated youth to the forefront, is also favored by educators around the globe for enhancing their academic and creative development. The implementation of the sportsdriven program, "Science in Sport," in an informal school setting exercising diverse STEM workshops that lead to engineering design challenges, could enhance high school students' inclination toward STEM fields and careers. Participants, self-motivated by sports, bolstered competencies by engaging through the engineering design process while developing a sports product during the challenge. The program exploited the students' enthusiasm in sports to participate in an inquiry-driven learning approach to experience design and engineering. The program outcome also offers opportunities for student enthusiasm and pride through product achievements. They expressed this by learning and applying science concepts such as friction, density, and material compactness in engineering design. The positive observations during the experimental phase complemented the pre-post questionnaires' statistical analysis results that interpreted the improvement in their attitudes toward STEM fields and careers. A normalized analysis of pre-post workshop attitude improvement did highlight the effect of the study on rural- and urban-based participants. The SWOT matrix laid out the strengths of the well-structured program, in addition to the limitations in providing an analysis of participant performance in international standard assessment tests as the aftermath.

The program keenly considers room for expansion with new activities and engineering challenges that could pave for an exciting STEM career pathway. This study can be replicated with university undergraduate students (UG) from sports science disciplines as a capacity-building means and enhancing their knowledge base in sports. We expect that the experiment on UG students may influence their career choices and align them toward the scientific background of the sports industry. The study offers scope in constructing
activities based on aerodynamics, thereby leading students to acquire innovative scientific perspectives and build new sports products. Currently, based on the results of this study, we are experimenting on students' ability to innovate sports games, if successful could be patented in the future. The ultimate goal in future development will align toward heightening the students' competencies and knowledge, preparing them for International research and science exposure.

Supplementary Materials: The following are available online at https://www.mdpi.com/2071-105 0/13/6/3483/s1, Table S1: Questionnaire sample, Table S2: Example of evaluation rubric handed to judges, Tables S3 to S12: Paired-samples statistics (pre/post tests), Figure S1: Different stages of making a concrete bowling ball from making a mould to testing the end products, Figure S2: Students during a concrete boat product challenge, Figure S3: A golf course is set up and tested by the students in the class.

Author Contributions: Conceptualization, R.A. and N.J.A.-T.; methodology, R.A. and N.J.A.-T.; formal analysis, N.S., J.B. and Z.A.; investigation, N.S. and J.B.; resources, R.A. and N.S.; writing—original draft preparation, N.S.; writing—review and editing, J.B. and Z.A.; supervision, N.J.A.-T.; project administration, N.J.A.-T. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Institutional Review Board Statement: Not Applicable.

Informed Consent Statement: Informed consent was obtained from all subjects involved in the study.

Data Availability Statement: The data presented in this study are available on request from the corresponding author. The data are not publicly available due to privacy.

Acknowledgments: The authors acknowledge the support from Qatar University Al-Bairaq program to conduct the study.

Conflicts of Interest: The authors declare no conflict of interest.

References

- Brannagan, P.M.; Giulianotti, R. Qatar, global sport and the 2022 FIFA world cup. In Leveraging Legacies from Sports Mega-Events: Concepts and Cases; Springer: Berlin/Heidelberg, Germany, 2014; pp. 154–165.
- Jaitman, L.; Scartascini, C. Deporte Para el Desarrollo. Banco Interamericano de Desarrollo. Departamento de Investigación y Economista Jefe. III. Título. IV. Serie. 2017. Available online: https://publications.iadb.org/publications/spanish/document/Deporte-para-eldesarrollo.pdf (accessed on 25 November 2020).
- Craig, E.; Thomas, R.; Hou, C.; Mathur, S. No Shortage of Talent: How the Global Market is Producing the STEM Skills Needed for Growth. Accenture Institute for High Performance (October, 2011). 2011. Available online: www.accenture.com/institute (accessed on 5 December 2020).
- Van Aalderen-Smeets, S.I.; Walma van der Molen, J.H.; Xenidou-Dervou, I. Implicit STEM ability beliefs predict secondary school students' STEM self-efficacy beliefs and their intention to opt for a STEM field career. J. Res. Sci. Teach. 2019, 56, 465–485.
 [CrossRef]
- Banning, J.; Folkestad, J.E. STEM education related dissertation abstracts: A bounded qualitative meta-study. J. Sci. Educ. Technol. 2012, 21, 730–741. [CrossRef]
- 6. Sahin, A. STEM clubs and science fair competitions: Effects on post-secondary matriculation. J. Stem Educ. 2013, 14, 5–11.
- Sahin, A.; Top, N. Teachers' Reflections on STEM Students on the Stage (SOS) Model. In A Practice-Based Model of STEM Teaching; Brill Sense: Rotterdam, The Netherlands, 2015; pp. 205–224.
- Abdulwahed, M.; Ghani, S.; Hasna, M.O.; Hamouda, A. Life Is Engineering Program: Impact of an Engineering Outreach Project in K-12. In Proceedings of the 2013 IEEE Global Engineering Education Conference (EDUCON), Berlin, Germany, 13–15 March 2013; pp. 827–833.
- Osman, I.H.; Anouze, A.L.; Irani, Z.; Lee, H.; Medeni, T.D.; Weerakkody, V. A cognitive analytics management framework for the transformation of electronic government services from users' perspective to create sustainable shared values. *Eur. J. Oper. Res.* 2019, 278, 514–532. [CrossRef]
- Shediac, R.; Samman, H. Meeting the Employment Challenge in the GCC *The Need for a Holistic Strategy*. Available online: https://www.iqpc.com/media/8251/10415.pdf (accessed on 1 December 2020).
- Weber, A.S. Linking Education to Creating a Knowledge Society: Qatar's Investment in the Education Sector. In Handbook of Research on Higher Education in the MENA Region: Policy and Practice; IGI Global: Hershey, PA, USA, 2014; pp. 52–73.

- 12. Sadler, P.M.; Sonnert, G.; Hazari, Z.; Tai, R. Stability and volatility of STEM career interest in high school: A gender study. *Sci. Educ.* 2012, 96, 411–427. [CrossRef]
- Wendell, K.; Connolly, K.; Wright, C.; Jarvin, L.; Rogers, C.; Barnett, M.; Marulcu, I. Incorporating Engineering Design into Elementary School Science Curricula. In Proceedings of the American Society for Engineering Education Annual Conference & Exposition, Louisville, Kentucky, 20–23 June 2010.
- 14. Aschbacher, P.R.; Ing, M.; Tsai, S.M. Boosting student interest in science. Phi Delta Kappan 2013, 95, 47–51. [CrossRef]
- George, P.S. The Middle School—And Beyond; ERIC. 1992. Available online: https://eric.ed.gov/?id=ED343254 (accessed on 21 March 2021).
- Kilpatrick, M.; Hebert, E.; Bartholomew, J. College students' motivation for physical activity: Differentiating men's and women's motives for sport participation and exercise. J. Am. Coll. Health 2005, 54, 87–94. [CrossRef]
- Vallerand, R.J.; Losier, G.F. An integrative analysis of intrinsic and extrinsic motivation in sport. J. Appl. Sport Psychol. 1999, 11, 142–169. [CrossRef]
- 18. Choi, J.-S.; Song, J. Students' preferences for different contexts for learning science. Res. Sci. Educ. 1996, 26, 341–352. [CrossRef]
- Assessment, P.f.I.S.; Co-Operation, O.f.E. Problem Solving For Tomorrow's World: First Measures Of Cross-curricular Competencies From Pisa 2003; OECD Publishing: Paris, France, 2004.
- Hanrahan, M. The effect of learning environment factors on students' motivation and learning. Int. J. Sci. Educ. 1998, 20, 737–753. [CrossRef]
- 21. OECD. Understanding the Brain: Towards a New Learning Science; OECD Paris: Paris, France, 2002.
- 22. Dewey, J. Experience and Education. Educ. Forum 1997, 50, 241-252. [CrossRef]
- Strong, W.B.; Malina, R.M.; Blimkie, C.J.; Daniels, S.R.; Dishman, R.K.; Gutin, B.; Hergenroeder, A.C.; Must, A.; Nixon, P.A.; Pivarnik, J.M. Evidence based physical activity for school-age youth. J. Pediatrics 2005, 146, 732–737. [CrossRef] [PubMed]
- Pellegrini, B.J. Materials World Modules—2002: A Nationally Representative Evaluation of Classroom Gains. J. Mater. Educ. 2010, 32, 185.
- Minner, D.D.; Levy, A.J.; Century, J. Inquiry-based science instruction—what is it and does it matter? Results from a research synthesis years 1984 to 2002. J. Res. Sci. Teach. Off. J. Natl. Assoc. Res. Sci. Teach. 2010, 47, 474–496.
- Contributors, W. FRC Bowling Ball Competition. Available online: https://www.concrete.org/students/studentcompetitions/ frcbowlingballcompetition.aspx (accessed on 13 December 2020).
- Sirianni, V.L.; Lee, K.L.; LeFevre, M.D.; Lindholm, J.W.; Aghayere, A.; Valentine, M. Assessing the Impact of the Concrete Canoe and Steel Bridge Competitions on Civil Engineering Technology Students. In Proceedings of the 2003 American Society for Engineering Education Annual Conference & Exposition, Salt Lake City, UT, USA, 20–23 June 2003.
- Kitchen, J.A.; Sonnert, G.; Sadler, P.M. The impact of college-and university-run high school summer programs on students' end of high school STEM career aspirations. Sci. Educ. 2018, 102, 529–547. [CrossRef]
- Mohr-Schroeder, M.J.; Jackson, C.; Miller, M.; Walcott, B.; Little, D.L.; Speler, L.; Schooler, W.; Schroeder, D.C. Developing Middle School Students' Interests in STEM via Summer Learning Experiences: See Blue STEM Camp. Sch. Sci. Math. 2014, 114, 291–301. [CrossRef]
- Anjarsari, P.; Prasetyo, Z.; Susanti, K. Developing Technology and Engineering Literacy for Junior High School Students through STEM-Based Science Learning. J. Phys. Conf. Ser. 2020, 1440, 012107. [CrossRef]
- Halim, L.; Soh, T.; Arsad, N. The effectiveness of STEM mentoring program in promoting interest towards STEM. Proc. J. Phys. Conf. Ser. 2018; 1088, p. 012001.
- Schwichow, M.; Zimmerman, C.; Croker, S.; Härtig, H. What students learn from hands-on activities. J. Res. Sci. Teach. 2016, 53, 980–1002. [CrossRef]
- 33. Turner, K. Materials world modules: A view from the science classroom. J. Mater. Educ. 2010, 32, 241.
- 34. Conrad, R.-M.; Donaldson, J.A. *Engaging the Online Learner: Activities and Resources for Creative Instruction;* John Wiley & Sons: Hoboken, NJ, USA, 2011.
- Teppo, M.; Soobard, R.; Rannikmäe, M. A Study Comparing Intrinsic Motivation and Opinions on Learning Science (Grades 6) and Taking the International PISA Test (Grade 9). Educ. Sci. 2021, 11, 14. [CrossRef]
- 36. Martin, S.H. Environment-Behaviour Studies in the Classroom. J. Des. Technol. Educ. 2009, 9, 77–89.
- Yager, R.E.; Penick, J.E. Perceptions of four age groups toward science classes, teachers, and the value of science. Sci. Educ. 1986, 70, 355–363. [CrossRef]
- Jones, S.; Myhill, D. 'Troublesome boys' and 'compliant girls': Gender identity and perceptions of achievement and underachievement. Br. J. Sociol. Educ. 2004, 25, 547–561. [CrossRef]
- Drazan, J.F.; Scott, J.M.; Hoke, J.I.; Ledet, E.H. Using Biomedical Engineering and "Hidden Capital" to Provide Educational Outreach to Disadvantaged Populations. In Proceedings of the 2014 36th Annual International Conference of the IEEE Engineering in Medicine and Biology Society, Chicago, IL, USA, 26–30 August 2014; pp. 5160–5163.
- Drazan, J.F.; Danielsen, H.; Vercelletto, M.; Loya, A.; Davis, J.; Eglash, R. A Case Study for Integrated STEM Outreach in an Urban Setting Using a Do-It-Yourself Vertical Jump Measurement Platform. In Proceedings of the 2016 38th Annual International Conference of the IEEE Engineering in Medicine and Biology Society (EMBC), Orlando, FL, USA, 16–20 August 2016; pp. 3027–3030.

- Drazan, J.F.; Loya, A.K.; Horne, B.D.; Eglash, R. From Sports to Science: Using Basketball Analytics to Broaden the Appeal of Math and Science among Youth. In Proceedings of the MIT-Sloan Sports Analytics Conference, Boston, MA, USA, 3–4 March 2017; pp. 1–16.
- 42. Drazan, J.F. Biomechanists can revolutionize the STEM pipeline by engaging youth athletes in sports-science based STEM outreach. J. Biomech. 2020, 99, 109511. [CrossRef] [PubMed]





Article Finland, A Package Deal: Disciplinary Climate in Science Classes, Science Dispositions and Science Literacy

Larry J. Grabau^{1,*}, Jari Lavonen² and Kalle Juuti²

- ¹ Department of Plant and Soil Sciences, University of Kentucky, Lexington, KY 40546, USA
- ² Faculty of Educational Sciences, University of Helsinki, 00014 Helsinki, Finland; jari.lavonen@helsinki.fi (J.L.); kalle.juuti@helsinki.fi (K.J.)

* Correspondence: larry.grabau@uky.edu

Abstract: Finland's educational prowess, though tempered by recent international assessments, has remained intact. This report focused on lessons that could be learned regarding secondary-level science education from the Program for International Student Assessment (PISA) 2015, science-focused assessment. That PISA iteration included not only science literacy but also students' science dispositions (epistemology, enjoyment, interest, and self-efficacy) and the schools' science climate measures (disciplinary climate and teaching support). Due to the hierarchical nature of the PISA data, multilevel models were employed in this Finnish study, involving 5582 students from 167 schools. Science dispositions (as outcome measures) were differently associated with teaching support and disciplinary climate (epistemology with neither; enjoyment and interest, with both). Science literacy (as an outcome measure) was associated with all four science dispositions, whether modeled with each science disposition separately or all four simultaneously. Science literacy was also associated with the disciplinary climate in science classes for all tested models. We concluded that, in the Finnish context, science dispositions and the disciplinary climate were predictive of science literacy. Furthermore, we presented evidence from the literature indicating that these conclusions may well extend to other international contexts.

Keywords: school science climate; disciplinary climate; science dispositions; epistemology; enjoyment; interest; self-efficacy; science literacy

1. Introduction

Finland's educational journey, from a decidedly ordinary past to its current highfunctioning status, has proved to be of enduring interest internationally [1]. While reading literacy has been a particularly strong area for Finland, triennial evaluations by the Program for International Student Assessment (PISA) have revealed high student performance in mathematics and science as well [2–4]. Even in the face of modest declines in science literacy outcomes [3,4], this nation has continued to attract interest from international educators. Sahlberg [1] identified multiple underlying causes for Finland's educational strength, including the national prominence of teaching as a profession (drawing many of their very best students), ongoing professional development programs (finely tuned to teachers' interests), solid financial support for schools (minimizing school-to-school variability in literacy outcomes), and also the firm rejection of "answers" to educational concerns that involve frequent high-stakes testing, extensive homework and tutoring, and national/regional control of content and practice. We note that the science curriculum in Finland emphasizes science literacy as an essential outcome for all students [5], underscoring the value ascribed to science literacy by the Finnish educational community. Furthermore, the literature review below was crafted to provide readers with a brief introduction to the context of this research by reviewing published papers that were both current and relevant.

Citation: Grabau, L.J.; Lavonen, J.; Juuti, K. Finland, A Package Deal: Disciplinary Climate in Science Classes, Science Dispositions and Science Literacy. *Sustainability* **2021**, 13, 13857. https://doi.org/10.3390/ su132413857

Academic Editor: Michail Kalogiannakis

Received: 12 November 2021 Accepted: 10 December 2021 Published: 15 December 2021

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2021 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/).

67

In a global context of increasing concern about the misunderstanding (and even outright rejection) of science as a contributor to personal life choices and to public decisionmaking [6,7], enhanced science literacy as an outcome for secondary education is absolutely critical [8]. Some authors [9] have argued that "open-mindedness", particularly as evinced by the adaptive intellectual traits of humility, courage and diligence—as opposed to the maladaptive intellectual traits of arrogance, cowardice and laziness [10]-needs to be developed among the world's populace. In contrast with the highly stratified secondary schools of some European nations (e.g., Benelux nations, Germany), Finland employs a "paraskoulu" (that is, "best school") system of primary and early secondary education. The goal of this "best school" system is to ensure that all schools are of consistently high quality; the mechanisms utilized in the pursuit of that goal include masters-level teacher training programs that produce teaching professionals [11,12], along with local curricular planning, teacher-conducted assessments of effectiveness, and numerous other innovations [13,14]. This approach has successfully "raised all boats"; that is, helped students across schools to attain higher science literacy [15,16]. In contrast, several European nations stratify students by track and grade; for example, Dutch-speaking Belgium's system (invoking significant grade retention and aggressive tracking) has trifurcated science literacy, with substantial differences in science literacy among the three main tracks in early secondary schools [17].

Science achievement has been assessed using a wide variety of metrics (grades in science classes [18,19], performance on standard science content evaluations [20], or large-scale international assessments [21,22]). Science literacy has been the focus of PISA's science assessment since its first science-focused interaction (in 2006); a key part of PISA's science literacy construct [23] is that students are asked to use what they know about science to develop solutions to complex science problems (versus simply supplying science information or solving routine science problems with a variety of included data). Some nations (for example, Greece) may be disadvantaged by the PISA science assessment, since their instructional models for science focus more on information retrieval and solving routine problems [24]. The PISA science-focused cycles (2006, 2015) have included both science dispositions (e.g., enjoyment, interest, self-efficacy) as well as selected science-teaching methods (hands-on activities and student investigations in 2006 [25], and disciplinary climate and teaching support—both in science classes in 2015 [26]).

Science dispositions have been shown to be associated with science literacy in a variety of international contexts [17,25,27,28]. Science enjoyment generally relates to a positive attitude toward science, particularly science learning processes [27,29]. Science interest indicates that a student has inclinations toward a particular science discipline (or multiple science disciplines) [29,30]. Note that this concept represents a broader construct of interest in science than does the "situational interest/situational engagement" widely investigated by Finnish researchers and their colleagues [31–34]. Students with high science self-efficacy believe that they "can do" particular science-related tasks successfully [27]. Science epistemology generally refers to students' beliefs about the nature of science [20,35–37]; in the PISA 2015 context, an adaptative suite of student beliefs about science epistemology involves both an emphasis on experimentation and on the changeability of science inferences [38]. Hereinafter, we simply refer to the aforementioned science dispositions as enjoyment, interest, self-efficacy and epistemology.

Importantly, research in Finland [32,39] and elsewhere [22,27,29,30,40] has considered how certain teaching methods in science classes (which we will generally term the "school science climate") may promote such science dispositions. Indeed, if science dispositions may be associated with enhanced science literacy, the school science climate (primarily experienced by students in their science classrooms) may very well be an important agent by which science teachers (and, thus, schools) are able to foster adaptive science dispositions. The school science climate has been primarily evaluated based on teachers' methods in science classrooms [22,27,29,30,40]; however, an innovation of the 2015 iteration of PISA was to ask individual students to evaluate their experiences in science classrooms [41]. Two key measures—disciplinary climate (in science classrooms) and teaching support (also in science classrooms)—were collected at the student level and designed to be aggregated to the school level. Prompts regarding the disciplinary climate focused on factors that allowed the whole class to "get down to business" in terms of learning about science (e.g., moving past distractions at the outset of class sessions quickly, on to the day's science concepts). Items included in order to construct teaching support in science classes similarly asked about students' experiences of teachers helping them to understand the concepts (versus leaving them to struggle). From a study of 21 nations/regions from PISA 2015, these two key school science climate measures were often associated with a pair of science dispositions—epistemology and enjoyment—as well as with science literacy per se [26]. A comparative study of classrooms, teachers and schools found that school-level features were more important predictors of science achievement than the other two measures [42].

Given the equity-promoting nature of Finnish schools [15,16], along with Finland's strong focus on teacher training [11,12] and teacher empowerment [13,14], we believed that an investigation of the associations among science dispositions, the school science climate, and science literacy could well provide lessons of interest to international science educators. In particular, identifying the extent to which science teaching interventions may enhance science dispositions could allow educators to more finely attune such outcomes. In addition, determining the strength of the associations between specific science dispositions most likely to have the potential to enhance science literacy. To achieve those ends, our twin research questions for this project were: (i) might the school science climate, over and above student background and school context, be associated with students' science dispositions (once again, over and above student background and school context) be associated with science literacy?

2. Materials and Methods

Data for this research were drawn from the most recent PISA iteration (2015) that was science-focused [3]. The corresponding dataset [43] provided a range of science dispositions (at the student level) along with an intriguing pair of school science-climate measures (aggregated, for our purposes, to the school level). We took advantage of this combination of observations to explore their relationships with science literacy, the specific measure of science achievement assessed under the PISA construct. Owing to the hierarchical nature of the data, with some measures relating to students within schools and others being expressed at the school level, we chose to implement multilevel modeling methods as our analytical tool. The ensuing sections describe data collection approaches, characterize outcome measures, and expand upon our analytical approaches.

2.1. Data Collection

National-level research specialists identified schools for participation in each round of the triennial PISA cycle; these schools were chosen to represent, as well as possible, the full range of school contexts (e.g., student enrollment, community size) expressed in that nation. For Finland in 2015, this sample included 167 schools. Within each school, approximately 35 students from the 15-year-old cohort were randomly chosen for this assessment (if a school had fewer than 35 students, all students were included). For Finland in 2015, this sample included 5882 students. Sampling weights at both the student and school levels were provided in the datasets [43] and were invoked as part of the analysis below. Missing values at both student- and school levels were managed differently. Missing values for students were treated at the analytical step of multilevel modeling; meanwhile, missing values for schools were treated by using the multiple imputation function within SPSS [44].

2.2. Outcome Measures

In the case of our first research question, four science dispositions (epistemology, enjoyment, interest, and self-efficacy) were successively treated as outcome measures. For our second research question, science literacy was the required outcome of a series of multilevel models. While not treated as outcome measures, two school science climate variables (the disciplinary climate in science classes and teaching support in science classes) were considered in both sets of multilevel models.

Tables A1 and A2 provide detailed information on the constructs that gave rise to the above suite of science-related variables (students' science dispositions and the school science climate). Here, we provide a brief characterization of each of these science measures, within this PISA context [38]. Among science dispositions, epistemology was tied to students' beliefs in experimentation and the possibility of change in scientific inferences. The variable of enjoyment is related to the positive emotions experienced by a student while learning science. The variable of interest took a composite look at students' interest in a series of scientific disciplines. Self-efficacy was derived from the level of students' agreement with statements about their ability to complete specific science-related tasks.

Science literacy, in the PISA context, refers to a student's capacity to implement science in actual life settings; specifically, it is "the ability to understand the characteristics of science and the significance of science in our modern world, to apply scientific knowledge, identify issues, describe scientific phenomena, draw conclusions based on evidence, and the willingness to reflect on and engage with scientific ideas and subjects" [23], p. 22. Note that science literacy was defined in the same way for the 2015 iteration of PISA; one key change was the move from five to ten plausible values for the later study. Other, more nuanced, constructions of science literacy exist [6,7,45]; however, the ready availability of the PISA-derived data based on the above framework for science literacy made it useful for this project.

The disciplinary climate and teaching support for science were both measured in the context of science classes; in addition, both were assessed at the student level but then aggregated to the school level. In this way, we were able to develop student-driven, school-level assessments of both of these school science climate measures. Such student ratings (to assess other teaching approaches) have been validated in a substantive way for 69 PISA 2015 nations/regions [46]. The disciplinary climate essentially addressed the degree to which the science classroom was orderly—getting onto the lesson topic promptly, students being able to hear the teacher, and other variables. Teaching support was related to the sense that individual students' learning mattered to the teacher, that it mattered enough for them to make sure that students were grasping the science ideas being considered.

2.3. Analytical Approaches

Our premise was that additional student- and school-level variables, apart from the science-related measures presented above, could also be associated with the outcome measures of science dispositions (research question #1) and science literacy (research question #2). Thus, we identified a series of likely candidate variables from among both student background and school context measures, then included those potentially important variables at the appropriate level of our multilevel models [47].

2.4. Exogenous Variables

Seven student-background measures were included in these analyses: gender (dummy), age, immigration background (double dummy), home language (double dummy), and an index of economic, social and cultural status (ESCS index). For gender, males were coded as 1, females as 0; no alternative gender identities were included as options on the student questionnaire. Student age was reported in years (within a 12-month bracket centered around 15.75 years). Immigration background considered second-generation immigration status as the reference group (coded as 0 for both dummy variables). Native and first-generation immigrant statuses were each coded as 1 for two separate dummy variables. Similarly, a Swedish home language was the reference group (coded as 0) for both dummy home language variables. Finnish home language and all other home languages were each coded as 1 for the two separate dummy variables regarding home

language. Note that home-language coding (for both dummy variables) was based on a match between students' home language and the language in which they took the PISA assessment. Setting aside those students for whom home languages were not recorded (61 and 5 of whom took Finnish and Swedish assessments, respectively), the vast majority of the assessments were taken in Finnish, by Finnish home-speakers (5213/5473; 95.2%). Meanwhile, a less substantial majority of Swedish assessments were taken by Swedish home-speakers (254/343; 74.1%). Of the other home languages group, 246 took the Finnish assessment, while only 6 took the Swedish assessment. In addition, Finnish home-speakers who took their assessment in Swedish (n = 83; 1.6% of the total Finnish home-speakers) and Swedish home-speakers who took their assessment in Finnish (n = 14; 5.2%) were assigned to the "other home language of their assessment. Finally, the ESCS index, while showing substantial variability, averaged somewhat higher than the standardized OECD mean for this index (0.259 versus 0.000).

In addition, four school context variables were included: school location, school size, index of the proportion of science teachers who were fully certified, and ESCS index—aggregated (from the student level). School location (as a dummy variable) compared schools located in communities of 100,000 or more residents (coded as 1) with those schools located in smaller communities (coded as 0). School size, reported by PISA as actual enrollment counts, was indexed to a basis of 1000 students (in order to make our primary tables more readable). The remaining pair of school context variables (proportion of teachers certified (ESCS index—aggregated)) were also included to capture a variability in outcome measures that is potentially explained by school context features.

2.5. Multilevel Modeling

In order to properly address the data hierarchy inherent in this PISA dataset, we made use of multilevel modeling strategies [47,48]. Our chosen software package was HLM8 [49]. For our first research question, four successive models were run (one for each of the four science dispositions under consideration as outcome measures). For our second research question (for which science literacy was the outcome), six successive models were invoked—a "basic" model (that included all student background variables, as well as all school context and school climate variables), four models that each included one of the four science dispositions, as independent variables (in addition to all variables from the basic model), and one "combined" model that included any of the science dispositions that achieved both significance (p < 0.05) and at least a small effect size ($d \ge 0.20$), once again, in addition to all variables from the basic model.

For all models, the full maximum likelihood approach was utilized; in addition, PISAsupplied weighting factors were included at both the student and school levels. Null models were run for all outcome variables (science dispositions as well as science literacy); such null models allowed us to not only identify the proportion of student- and school-level variance accounted for by subsequent models but also allowed us to calculate intraclass correlation coefficients (ICC), which can help determine the propriety of the inclusion of school-level analyses. In pursuit of parsimonious models, we included all student-level variables (level one included student background for both research questions, and science dispositions, as applicable, for research question #2 only) in the initial model. Student-level variables with the least significant results were iteratively dropped from the level-one model until all remaining variables met the p < 0.05 criterion. At that point, all school-level variables were loaded into the level two model; once again, the least significant variable was removed from successive model iterations until all level-two variables were also significant at p < 0.05. In order to evaluate model performance, we calculated the proportion of variance explained at both the student and school levels for each parsimonious model [48]. Finally, in order to assess the effect size of each significant variable, we calculated Cohen's d (by dividing the variable's modeled coefficient by its standard deviation). We used

Cohen's [50] relativization of effect sizes (0.20 to 0.49 was taken as small; 0.50 to 0.79, as medium; and \geq 0.80 as large).

3. Results

Additional materials for this paper include Tables A1 and A2 (providing the underlying prompts used to construct science dispositions and school climate measures, respectively [23]), as well as descriptive statistics (Table A3) and bivariate correlations among science dispositions (Table A4) and school-level variables (whether context or climate; Table A5). Selected information from those Additional materials is briefly noted here. For example, the Finnish sample implemented for this research included 96% native students, 89.6% of whom spoke Finnish at home, and a mean ESCS index of 0.259 (indicating that the Finnish mean income exceeded the OECD mean). Science dispositions closely approximated OECD means, as did school climate measures. While bivariate correlations were significant among both science dispositions and school-level variables, only one such correlation (between enjoyment and interest) reached the moderate level among science dispositions (none reached this level among school-level variables).

Table 1 presents the results of multilevel models with each of the four science dispositions treated as outcome variables. Among student background variables, ESCS was the only measure associated with all four science dispositions; this association was positive in each case. Perhaps more intriguingly, home language showed a complex association with epistemology. Students who spoke Swedish at home (and took the PISA assessment in Swedish) showed higher levels of epistemology than either Finnish speakers (who took this assessment in Finnish) or speakers of other languages. Among school climate measures, teaching support in science classes was associated with three of the four science dispositions (epistemology was the exception). One could, thus, argue that school-level teaching support may have promoted enjoyment, interest, and self-efficacy. The disciplinary climate in science classes was significantly (and positively) associated with two of the studied science dispositions: enjoyment and interest. We note that these models accounted for modest proportions of the student-level variance (0.02 to 0.08); in addition, ICC data for these models showed a modest level of school-to-school variability (0.02 to 0.05).

	Epistem	ology	Enjoyment		Interest		Self-Efficacy	
Variable ¹	Coeff.	d ²	Coeff.	d	Coeff.	d	Coeff.	d
Student background:								
Gender ³	-	-	-	-	0.29	0.12	0.26	0.07
Age	-	-	-	-	0.13	0.03	0.16	0.04
Other home lang.	-0.34	0.04	-	-	-	_	-	-
Finn. home lang.	-0.21	0.03	_	-	-	_	-	-
ESCS index	0.23	0.16	0.25	0.16	0.23	0.13	0.31	0.17
School context:								
ESCS indag.	-	_	_	-	0.23	0.17	-	-
School climate:								
Disc. clim. sci. cl.	-	_	0.28	0.35	0.11	0.16	-	-
Tch. sup. sci. cl.	-	-	0.19	0.39	0.17	0.27	0.12	0.20
Prop., student var.	0.0	8	0.0)2	0.0)6	0.	04
Intraclass corr.	0.0	3	0.0)3	0.0)5	0.	02

Table 1. Multilevel models for science dispositions (as outcomes) for Finland.

¹ Significant variables in this column remained in a given model by meeting the p < 0.05 criterion; all student background and school variables were tested in each model. Those student- and school-level variables that were not included dropped out of all multilevel models shown. ² Bolded "d" values met a minimum 0.20 criterion. ³ For more complete variable descriptions, please see the Methods section and Appendix A. Dashes (–) in the table indicate that a given variable dropped out of a given model.

Table 2 catalogs the results of our "basic" (no science dispositions included) multilevel model, taking science literacy as an outcome. Student background variables of female gender, increasing age (within the 15-year-old cohort), native status, Finnish home lan-

guage and ESCS level were each associated with science literacy. Note that the coefficients in this table directly represent the differences among science literacy scores for the contrasting levels of a given variable. For example, female students averaged higher (7.72) PISA scale points than did males. Smaller school size (a school context measure) and a stronger disciplinary climate in science classes (a school science climate variable) were both significantly associated with science literacy in Finnish students. Based on our effect size analysis, Finnish home language, ESCS index, and disciplinary climate were best attested as strongly meaningful. We note that both the proportion of student-level variance accounted for and the ICC measure were moderate for this model.

	Basic	Model
Variable ¹	Coeff.	d ²
Student background:		
Gender	-7.72	0.04
Age	22.56	0.08
Native students	45.71	0.19
Finn. home lang.	35.38	0.45
ESCS	30.89	0.53
School context:		
School size (per 1000)	-25.31	0.16
School climate:		
Disc. clim. sci. cl.	41.80	0.32
Prop., student var.	0.5	10
Intraclass corr.	0.0)9

Table 2. A basic multilevel model for science literacy in Finland (no science dispositions included).

¹ Significant variables in this column remained in a given model by meeting the p < 0.05 criterion; all student background and school variables were tested in each model. No science dispositions were included in this basic model. Variables that were not included failed to meet that criterion for all multilevel models shown. For more complete variable descriptions, please see the Methods section and Appendix A. ² Bolded "d" values met a minimum 0.20 criterion.

Multilevel model results with science literacy as the outcome are shown in Table 3. We focus our comments upon the four science dispositions, along with the only significant school science climate measure, this being the disciplinary climate in science classes. Coefficients and d-values for student background and school context measures are included to underscore our modeling methods; that is, each model included all student background and school-level measures. We focused on the "above and beyond" associations of science dispositions with science literacy, after taking into account a broad portfolio of student background and school context variables. While all four science dispositions were significantly associated with science literacy, the strengths of those associations, as evaluated by effect sizes, differed dramatically. A large effect size was the result for enjoyment; a medium effect size for self-efficacy and epistemology; and a small effect size for interest. Full unit increases (two standard deviations from the mean) in epistemology, enjoyment, interest, or self-efficacy indicated a gain of 19 to 31 scale points in science literacy (all with small effect sizes) when separately included in science literacy models. Meanwhile, the disciplinary climate in science classes was associated with science literacy for all four models, including single science dispositions. A full unit increase in disciplinary climate (again, two standard deviations) was associated with a gain of 30-41 PISA scale points. Thus, the disciplinary climate in science classes could be regarded as promoting science literacy in the presence of all four studied science dispositions. For each model, the proportion of student-level variance that was accounted for was substantially greater than the 0.10 level accounted for by our basic model.

	Epistemology Enjoyment		Interest		Self-Efficacy			
Variable ¹	Coeff.	d ²	Coeff.	d	Coeff.	d	Coeff.	d
Student background:								
Gender	-6.49	0.05	-9.70	0.06	-17.20	0.08	-12.64	0.09
Age	19.68	0.07	21.72	0.09	18.56	0.07	20.28	0.08
Native students	36.85	0.16	71.76	0.19	56.06	0.27	46.71	0.22
Other home lang.	-	-	-	-	-23.00	0.25	-	_
Finn. home lang.	37.25	0.48	-	-	-	-	30.41	0.38
ESCS	23.85	0.41	24.04	0.29	22.27	0.29	24.99	0.32
Science disposition	31.12	0.63	24.96	1.40	31.31	0.31	18.81	0.74
School context:								
School size (per 1000)	24.87	0.17	_	_	_	-	-	-
School climate:								
Disc. clim. sci. cl.	39.34	0.29	30.25	0.19	31.32	0.23	40.55	0.29
Prop., student var.	0.1	9	0.1	7	0.2	.0	0.3	15
Delta, student var.	0.0	9	0.0)7	0.1	0	0.0)5

Table 3. Multilevel models for science literacy in Finland, with science dispositions as independent variables.

¹ Significant variables in this column remained in a given model by meeting the p < 0.05 criterion; all student background and school variables were tested in each model. Variables not included in this column dropped out of all multilevel models tested for science literacy as an outcome variable. For more complete variable descriptions, please see the Methods section and Appendix A.² Bolded "d" values met a minimum 0.20 criterion. Dashes (–) in the table indicate that a given variable dropped out of a given model.

For Table 4, we focused on science dispositions and the disciplinary climate in science classes, in the context of our "combined" model. This multilevel model simultaneously considered all four of the studied science dispositions; notably, all four science dispositions remained in our combined model. This indicated that each of the four science dispositions was uniquely associated with science literacy as an outcome in the presence of the other three science dispositions. Effect sizes differed; while epistemology reached a modest effect size, and enjoyment and self-efficacy each achieved a small effect size, interest fell just short of a small effect size. The disciplinary climate in science classes remained in this combined model, testifying to the persistence of its association with science literacy in the presence of the quartet of science dispositions considered.

Table 4. A combined multilevel model for science literacy for Finland.

	Combined Model		
Variable ¹	Coeff.	d ²	
Student background:			
Gender	-14.98	0.12	
Age	16.35	0.07	
Native students	42.18	0.23	
Finn. home lang.	31.56	0.44	
ESCS ind.	16.97	0.21	
Science disposition			
Epistemology	22.07	0.56	
Enjoyment	6.26	0.45	
Interest	19.05	0.17	
Self-efficacy	8.62	0.44	
School context:			
School size (per 1000)	-26.00	0.19	
School climate:			
Disc. clim. sci. cl.	33.70	0.25	
Prop., student var.	0.2	27	
Delta, student var.	0.1	7	

¹ Significant variables in this column remained in a given model by meeting the p < 0.05 criterion; all student background and school variables were tested in each model. Variables not included failed to meet that criterion for all multilevel models shown. For more complete variable descriptions, please see the Methods section and Appendix A.² Bolded "d" values met a minimum 0.20 criterion.

4. Discussion

Our research assessed the potential associations of two school science climate variables with science dispositions and science literacy. The disciplinary climate and teaching support (both in science classes) comprise our first focus here. Furthermore, we have termed this pair of concepts "school science climate" variables, since both specifically refer to the context of science teaching, as evaluated by the students themselves. Note that, for purposes of our analysis, we aggregated student responses to the school level for both disciplinary climate and teaching support variables.

Teaching support and the disciplinary climate were both positively associated with some of the science dispositions evaluated in this research; specifically, these two measures of school science climate were positively associated with both enjoyment and interest (Table 1). Given that both enjoyment and interest remained in our "combined" model, as associated with science literacy (Table 4), one could reasonably infer that both of these aspects of the school science climate could be considered promotive of science literacy. In fact, the disciplinary climate was directly associated with science literacy as an outcome measure for all four models, including a single science disposition (Table 3), and was also associated with science literacy in the combined model (Table 4). Thus, we argue that an emphasis on improving the disciplinary climate of Finnish science classes could potentially be a productive approach to improving science literacy in that national context. Variability between Finnish schools in science literacy for PISA 2015 was guite low (less than 10% between schools, along with Iceland and Norway) in contrast with much more stratified national school systems (nearly 50% variability in Belgium and Germany, and 65% in the Netherlands) [16]. However, variability in the measure of interest here (disciplinary climate in science classes, tested at the school level) was indeed associated with science literacy in all the models we tested (Tables 3 and 4).

4.1. School Science Climate, Selected PISA Measures

Based on PISA 2015 data and invoking multilevel structural equation modeling, potential mechanisms for linkages among students' socioeconomic status (SES), the disciplinary climate in science classes, and science literacy have been investigated in detail [51]. Some support for a "compensation" mechanism was obtained, to wit: "secondary data analysis of the PISA 2015 data from five Nordic countries resulted in consistent and robust evidence supporting the compensation hypothesis, that is, the disciplinary climate's contribution to science achievement above and beyond SES at both the student and school levels" [51], p. 219. Multilevel studies have shown connections among disciplinary climate and science literacy for four Chinese cities (Beijing, Shanghai, Jiangsu, and Guandong) [52]; as well as for Singapore [40]. Notably, the above-mentioned Chinese research was unable to connect teaching support in science classes with science literacy [52]. Recent work has positioned both teaching support and disciplinary climate as associated with a pair of science dispositions (epistemology and enjoyment), as well as with science literacy for many of the 21 diverse nations/regions evaluated in that project [26]. In summary, a diverse suite of studies supports our current conclusions for Finland; particularly, that the disciplinary climate in science classrooms was broadly associated with science literacy.

4.2. Effective Science-Teaching Models, a Student-Centered View

Finnish research, as noted above, has recently emphasized students' engagement in science learning on a nearly instantaneous basis [31,32,34,39]. One profound strength of such work is that it assesses students' engagement in science lessons in response to teaching "moments", rather than focusing on teachers' activities alone [33]. Specifically, these recent studies have shown enhanced situational engagement by bringing a playful, interactive, design and test intervention to physics [34]; developing models and crafting explanations [31], inquiry- and problem-based learning approaches as developed via teacher–researcher collaborations [32], and dialog-based teacher talk [39]. We believe that our current paper aligns well with this research—particularly because it focuses on students' perceptions of their

learning environment (rather than fixating on teachers' strategies and approaches per se). Notably, a broad review of PISA 2015 (69 nations/regions) found that student ratings of four different teaching contexts "captured meaningful differences in teaching quality between schools" [46], p. 275. Estonian research supported teacher-centered science instruction as being connected with positive student outcomes (enjoyment); however, this teacher-centered approach remained most prominent in lower secondary classrooms, in spite of teachers' expressed desires to provide a more student-centered learning environment [29]. Of course, in some cases [22], students' perceptions of teaching quality have not been associated with science achievement. Thus, while the responses of a given set of students to specific science teaching and learning environments have been diverse [21,27,30,53,54], we believe that work considering students' own responses to their learning contexts should be highly valued.

4.3. Science Dispositions

All four science dispositions were positively associated with science literacy, not only when tested in separate models but also when tested together in a single combined model. That approach-utilizing a combined model-allowed us to identify those science dispositions most strongly associated with science literacy. Epistemology rose to the top in that combined model (with a moderate effect size); meanwhile, enjoyment and self-efficacy both registered a small effect size in that same model. Hence, we focused on epistemology, enjoyment, and self-efficacy. Recent Finnish research has associated students' epistemic beliefs with their science performance [55]; Taiwanese studies have identified student clusters that held simultaneously divergent levels of epistemic beliefs, enjoyment, and science literacy [37]. In a US context, teachers' epistemic orientations were related to their adoption of science teaching practices [56], indicating that efforts to enhance students' epistemic beliefs around science may need to first address their teachers' epistemic beliefs. A causal linkage among epistemic beliefs in science, intellectual risk-taking and science literacy has been detected [20]; enjoyment has been linked with science literacy for a suite of high-performing nations/regions (Canada, China, Finland, and Hong Kong) [27] along with Turkey [40] and the US [25]; and self-efficacy has been associated with science literacy in multiple contexts [25,27]. Notably, particular science teaching interventions (models [25,31] and hands-on activities [25,57]) that have evoked situational or sustained interest in science learning have been related to this set of science dispositions. Thus, we suggest that teaching interventions focused on enhancing students' experiences of science may not only be related to enhanced science dispositions but also to elevated science literacy. Furthermore, we suggest that a greater emphasis on the epistemic beliefs of science students may be appropriate.

4.4. Surprising Home-Language Associations

While Swedish-speaking students who took the PISA-2015 assessment in Swedish had significantly higher average epistemology than did either Finnish-speakers who took the assessment in Finnish, or other students whose assessment and home languages did not match, the highest average science literacy scores were detected for the Finnish/Finnish group. An understanding of this surprising feature of this study has been elusive; we suggest that the Swedish/Swedish context may require further scrutiny. Particularly, why might an instructional context (Swedish instruction and assessment for Swedish speakers) give rise to higher levels of epistemology, yet show lower science literacy? We note that science teachers are slightly less qualified in Swedish- (93%) compared to Finnish-speaking (98%) schools [58], providing at least a rationale for higher science literacy in the Finnish-speaking students. Furthermore, since Swedish-speaking concentrations (and therefore, Swedish language schools) differ between the more rural west coast communities and more urban communities on the south coast, the selection of schools for the PISA sample could influence science-related outcomes. Many global contexts include multiple home languages in PISA samples; it is less common that PISA assessments are offered in multiple

languages within a given nation/region. Spain is a remarkable example of providing PISA assessments in several local languages; differences in outcomes (for example, science literacy) have been noted in Spain on a provincial scale (roughly corresponding to some of the local languages) [59]. Flemish students whose home language did not match the assessment language scored substantially lower in science achievement than those students not dealing with a language mismatch [60]. Language congruence (taking the PISA assessment in one's home language) has also given rise to surprising results elsewhere; for example, minority French-speaking students in Canada had higher science literacy than either English-speakers or speakers of other languages, while German speakers in Italy had lower test anxiety than did Italian speakers [61]. We suspect that instruction—and assessment in one's home language may be important not only for students' academic achievement but also for their dispositions toward learning [62]. Perhaps the students' opportunity to demonstrate competence could be improved by enhanced language congruence [63–65]?

4.5. Conclusions and Limitations

We draw general conclusions regarding the school science climate, science dispositions, and home language. The disciplinary climate in science classes was associated with science literacy and science dispositions not only in our Finnish research work but also in a variety of other international contexts. More broadly, an emphasis on school science climate could merit more attention. Research drawing upon students' instantaneous and longer-term science engagement appears to have the potential to add substantive value to conversations about well-crafted teaching and learning environments. Furthermore, it seems that the epistemic beliefs of science students may merit a greater focus among research considering science dispositions. There may be a window for expanded research on language coherence and student learning; that is, students may perform better if they are able to experience instruction and assessment in their home languages. In the context of Hong Kong, both lower- and moderate-performing science students benefitted from science instruction in their home (Chinese) language [65].

The primary limitation of this research is the correlational nature of the data upon which it was based. Experimental and/or longitudinal studies can clearly provide more definitive conclusions since their inferences are based on comparisons across treatments or over time. Thus, correlational studies require the support of more definitive research; however, they can obviously provide some intriguing insights as to the directions of that ensuing work. A second limitation relates to the applicability of some of the constructs included in this research. For example, the underlying prompts for epistemology only related to a pair of conceptual underpinnings of students' epistemological beliefs: experimentation (as an important support for students' inferences) and the changeable nature of science's assertions (depending on the emerging results). Obviously, both support for experimentation and comfort with change are important aspects of an adaptative epistemic belief among students; however, other patterns have also been implicated as part of a positive epistemology of science [35,36].

In conclusion, we harmonize with Sahlberg's voice [1] from the third edition of his well-known text. Perhaps his most significant insights centered on contextualization and commitment; that is, could it be that other nations could experience some of Finland's success, not by trying to precisely match their methods but instead through appropriate contextualization of their underlying principles? Furthermore, can one infer from his celebrated review of Finland's educational system that sustained excellence in science education requires a profound commitment to underlying values?

Author Contributions: Conceptualization, L.J.G., J.L. and K.J.; methodology, L.J.G.; software, L.J.G.; validation, J.L. and K.J.; formal analysis, L.J.G.; investigation, L.J.G.; resources, L.J.G.; data curation, L.J.G.; writing—original draft preparation, L.J.G.; writing—review and editing, J.L. and K.J.; visualization, L.J.G. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Institutional Review Board Statement: Ethical review and approval were waived for this study, since those processes were managed by the research team which collected this publicly available data, rather than the authors of this manuscript.

Informed Consent Statement: Informed consent was obtained from all subjects involved in this study by the national-level managers of the PISA 2015 assessment.

Data Availability Statement: Data utilized in these analyses are available for free download from OECD at the following link: https://www.oecd.org/pisa/data/2015database/ (accessed on 15 October 2021).

Acknowledgments: Both Rianne Janssen (Faculty of Psychology and Educational Sciences, KU Leuven) and Benoît Galand (Faculty of Psychology and Educational Sciences, UC Louvain) supported this project by providing office space at their respective Belgian universities, along with a conducive, collegial research environment during the invited professorships of L.J.G. at their institutions.

Conflicts of Interest: The authors declare no conflict of interest.

Appendix A

Table A1. PISA Items Descriptive of Science Dispositions.

Epistemology (Epistemological Beliefs) (First term, as utilized in manuscript; second term (in parentheses), as utilized by PISA)

How much do you disagree or agree with the statements below?

(a) A good way to know if something is true is to do an experiment. (b) Ideas in
stroad science> sometimes change. (c) Good answers are based on evidence from many different experiments.
(d) It is good to try experiments more than once to make sure of your findings. (e) Sometimes

broad science> scientists change their minds about what is true in science. (f) The ideas in

broad science> science books sometimes change. (1 = Strongly agree, 2 = agree, 3 = disagree, 4 = strongly disagree).

Enjoyment (Enjoyment of Science)

How much do you agree with the statements below?

(a) I generally have fun when I am learning science topics. (b) I like reading about science. (c) I am happy doing science problems. (d) I enjoy acquiring new knowledge in science. (e) I am interested in learning about science. (1 = strongly agree, 2 = agree, 3 = disagree, 4 = strongly disagree).

Interest (Interest in Broad Science Topics)

How much interest do you have in learning about the following science topics?

(a) Topics in physics. (b) Topics in chemistry. (c) The biology of plants. (d) Human biology.
(e) Topics in astronomy. (f) Topics in geology. (g) Ways scientists design experiments. (h) What is required for scientific explanations? (1 = high interest, 2 = medium interest, 3 = low interest, 4 = no interest.

Self-efficacy (Science Self-efficacy)

How easy do you think it would be for you to perform the following tasks on your own?

(a) Recognize the science question that underlies a newspaper report on a health issue.

(b) Explain why earthquakes occur more frequently in some areas than in others. (c) Describe the role of antibiotics in the treatment of disease. (d) Identify the science question associated with the disposal of garbage. (e) Predict how changes to an environment will affect the survival of certain species. (f) Interpret the scientific information provided on the labeling of food items. (g) Discuss how new evidence can lead you to change your understanding about the possibility of life on Mars. (h) Identify the better of two possible explanations about the formation of acid rain. (1 = I could do this easily, 2 = I could do this with a bit of effort, 3 = I would struggle to do this on my own, 4 = I couldn't do this).

Source: [38].

Table A2. PISA Items Descriptive of School Science Climate Variables.

Disciplinary Climate in Science Classes (aggregated from student level)

How often does the following happen?

(a) Students don't listen to what the teacher says. (b) There is noise and disorder. (c) The teacher has to wait a long time for students to quiet down. (d) Students cannot work well. (e) Students don't start working until a long time after the lesson begins. (1 = Every Lesson, 2 = Most Lessons, 3 = Some Lessons, 4 = Never or Hardly Ever).

Teaching Support in Science Classes (aggregated from student level)

How often does the following happen?

(a) The teacher shows an interest in every student's learning. (b) The teacher gives extra help to students with their learning. (c) The teacher continues teaching until the students understand.
(d) The teacher gives students an opportunity to express opinions. (1 = Every Lesson, 2 = Most Lessons, 3 = Some Lessons, 4 = Never or Hardly Ever).

Source: [38].

Table A3. Descriptive statistics for student- and school-level variables included in the multilevel analysis for Finland.

Variable ¹	п	Mean	s.d.	min.	max.
Student background:					
Male gender	5882	0.513	0.500	0	1
Age	5882	15.720	0.284	15.250	16.250
First-gen. immig.	5794	0.022	0.146	0	1
Native students	5794	0.960	0.195	0	1
Other home lang.	5816	0.060	0.238	0	1
Finn. home lang.	5816	0.896	0.305	0	1
ESCS index	5812	0.259	0.748	-4.112	3.567
Science disposition:					
Epistemology	5462	-0.071	0.941	-2.790	2.155
Enjoyment	5599	-0.071	1.006	-2.115	2.164
Interest	5510	-0.093	0.972	-2.498	2.451
Self-efficacy	5470	-0.041	1.183	-3.757	3.278
School context:					
School location	167	0.275	0.448	0	1
School size	167	411.620	217.394	15.000	1044.000
Sci. teach. cert.	167	0.929	0.225	0.000	1.000
ESCS index-ag.	167	0.236	0.336	-0.970	1.370
School climate:					
Disc. clim. sci. cl.	167	-0.078	0.382	-1.114	1.884
Tch. sup. sci. cl.	167	0.190	0.285	-1.702	0.821

¹ Variable explanation (details in Methods): gender, males as 1; age, in years; first-gen. immig., first-generation immigrant students as 1; native students, as 1; other home lang,, other language spoken at home as 1; Finn. home lang,, Finnish spoken at home as 1; ESCS index, index of economic, social, and cultural status; epistemology, epistemological beliefs; enjoyment, enjoyment of science; interest, interest in broad science topics; self-efficacy; science self-efficacy; school location, community of 100,000 or larger as 1; school size, school enrollment; sci. teach. cert., index of science teachers fully certified; ESCS index—agg., index of economic, social, and cultural status, aggregated student variable; disc. clim. sci. cl., disciplinary climate in science classes, aggregated student variable.

DD 1 1 A 4	D' ' '	1		1	C T 1 1
Table A4	Bivariate	correlations and	iong science	dispositions	tor Finland
Incie III.	Divariance	correlation to an	tong belence	anopoontion	, ioi i muana.

Key Variable	Epist. ¹	Enjoyment	Interest	Self-Eff.
Epistemology	1.000	0.302 **	0.295 **	0.229 **
Enjoyment		1.000	0.539 ** ^{,2}	0.363 **
Interest			1.000	0.355 **
Self-efficacy				1.000

¹ Variable explanation (details in Methods): epistemology, epistemological beliefs; enjoyment, enjoyment of science; interest, interest in broad science topics; self-efficacy, science self-efficacy. ² Bolded values indicate a moderate bivariate correlation among indicated pairs of variables (0.400 to 0.799). **, Correlation is significant at the 0.01 level (2-tailed).

Table A5. Bivariate correlations among school-level variables for Finland.

	School	Sci. Teach.	ESCS	Disc. Clim.	Tch. Sup.
Key Variable	size ¹	cert.	indag.	sci. cl.	sci. cl.
School size	1.000	0.200 **	0.232 **	-0.076	0.160 *
Sci. teach. cert.		1.000	0.268 **	-0.337 **	-0.150
ESCS indag.			1.000	0.051	-0.020
Disc. clim. sci. cl.				1.000	0.267 **
Tch. sup. sci. cl.					1.000

¹ Variable explanation (details in the Methods section): school size, school enrollment; sci. teach. cert., index of science teachers fully certified; ESCS index-ag., index of economic, social, and cultural status, aggregated student variable; disc. clim. sci. cl., disciplinary climate in science classes, aggregated student variable; and tch. sup. sci. cl., teaching support in science classes, aggregated student variable; **, *, correlation is significant at the 0.01 and 0.05 levels, respectively (2-tailed).

References

- Sahlberg, P. Finnish Lessons 3.0: What Can the World Learn from Educational Change in Finland, 3rd ed.; Teachers College Press: New York, NY, USA; London, UK, 2021; p. 291.
- OECD. PISA 2006. Science Competencies for Tomorrow's World. Available online: https://www.oecd.org/pisa/pisaproducts/ 39703566.pdf (accessed on 12 October 2021).
- OECD. PISA 2015. PISA 2015 Results in Focus. Available online: https://www.oecd.org/pisa/pisa-2015-results-in-focus.pdf (accessed on 12 October 2021).
- OECD. PISA 2018. PISA 2018 Results—Combined Executive Summaries, Volumes I, II, & III. Available online: https://www. oecd.org/pisa/Combined_Executive_Summaries_PISA_2018.pdf (accessed on 12 October 2021).
- Lavonen, J. How the Finnish compulsory school science curriculum emphasizes scientific literacy. *Eest. Haridusteaduste Ajak.* 2021, 9, 26–46. [CrossRef]
- He, L.; Chen, Y.; Xiong, X.; Zou, X.; Lai, K. Does Science Literacy Guarantee Resistance to Health Rumors? The Moderating Effect of Self-Efficacy of Science Literacy in the Relationship between Science Literacy and Rumor Belief. Int. J. Environ. Res. Public Health 2021, 18, 2243. [CrossRef] [PubMed]
- Howell, E.L.; Brossard, D. (Mis)informed about what? What it means to be a science-literate citizen in a digital world. *Proc. Natl. Acad. Sci. USA* 2021, 118, e1912436117. [CrossRef] [PubMed]
- Autieri, S.M.; Amirshokoohi, A.; Kazempour, M. The science-technology-society framework for achieving scientific literacy: An overview of the existing literature. *Eur. J. Sci. Math. Educ.* 2016, *4*, 75–89. [CrossRef]
- Sharon, A.J.; Baram-Tsabari, A. Can science literacy help individuals identify misinformation in everyday life? *Sci. Educ.* 2020, 104, 873–894. [CrossRef]
- Taylor, R.M. Open-Mindedness: An Intellectual Virtue in the Pursuit of Knowledge and Understanding. *Educ. Theory* 2016, 66, 599–618. [CrossRef]
- 11. Eklund, G. Master's thesis as part of research-based teacher education: A Finnish case. J. Teach. Educ. 2018, 8, 5–20.
- 12. Tirri, K. The last 40 years in Finnish teacher education. J. Educ. Teach. 2014, 40, 600-609. [CrossRef]
- Darling-Hammond, L. Teacher education around the world: What can we learn from international practice? *Eur. J. Teach. Educ.* 2017, 40, 291–309. [CrossRef]
- Niemi, H.; Toom, A.; Kallioniemi, A. Miracle of Education: The Principles and Practices of Teaching and Learning in Finnish Schools; Sense Publishers: Rotterdam, The Netherlands, 2012.
- Beese, J.; Liang, X. Do resources matter? PISA science achievement comparisons between students in the United States, Canada and Finland. *Improv. Sch.* 2010, 13, 266–279. [CrossRef]

- OECD. PISA 2015. PISA 2015 Results (Volume I): Excellence and Equity in Education. Available online: https://www.proquest. com/openview/e8c066902afa5f1a32b207d74370b87f/1?pq-origsite=gscholar&cbl=2026456 (accessed on 13 October 2021).
- 17. Grabau, L.J.; Von Damme, J.; Dockx, J.; All of KU-Leuven, Leuven, Belgium. Personal Communication, 2021.
- Murphy, S. The Impact of School Disadvantage on Senior Secondary Science: A Study of Patterns of Participation and Achievement in Government Secondary Schools in Victoria, Australia. Res. Sci. Educ. 2018, 50, 1603–1618. [CrossRef]
- Megoğullari, M.; Guill, K.; Köller, O. Effectiveness of private tutoring during secondary schooling in Germany: Do the duration of private tutoring and tutor qualification affect school achievement? *Learn. Instr.* 2020, 66, 23. [CrossRef]
- Özbay, H.E.; Köksal, M.S. Middle School Students' Scientific Epistemological Beliefs, Achievements in Science and Intellectual Risk-Taking. Sci. Educ. 2021, 30, 1233–1252. [CrossRef]
- Forbes, C.T.; Neumann, K.; Schiepe-Tiska, A. Patterns of inquiry-based science instruction and student science achievement in PISA 2015. Int. J. Sci. Educ. 2020, 42, 783–806. [CrossRef]
- Zhang, F.; Bae, C.L.; Broda, M. Science Self-Concept, Relatedness, and Teaching Quality: A Multilevel Approach to Examining Factors that Predict Science Achievement. Int. J. Sci. Math. Educ. 2021, 1–27. [CrossRef]
- OECD. PISA 2006. PISA 2006 Technical Report. Available online: http://www.oecd.org/pisa/pisaproducts/42025182.pdf (accessed on 12 October 2021).
- 24. Karakoldis, A.; Educational Research Centre, St Patrick's Campus, Drumcondra, Dublin. Personal Communication, 2020.
- Grabau, L.J.; Ma, X. Science engagement and science achievement in the context of science instruction: A multilevel analysis of U.S. students and schools. Int. J. Sci. Educ. 2017, 39, 1045–1068. [CrossRef]
- Grabau, L.J. International secondary science education: Associations among science epistemology, science enjoyment, school climate and science achievement. In Proceedings of the American Educational Research Association Annual Meeting, Virtual, 8–12 April 2021.
- Lau, K.-C.; Ho, S.-C.E. Attitudes Towards Science, Teaching Practices, and Science Performance in PISA 2015: Multilevel Analysis of the Chinese and Western Top Performers. *Res. Sci. Educ.* 2020, 12. [CrossRef]
- Radišić, J.; Selleri, P.; Carugati, F.; Baucal, A. Are students in Italy really disinterested in science? A person-centered approach using the PISA 2015 data. Sci. Educ. 2021, 105, 438–468. [CrossRef]
- Teppo, M.; Soobard, R.; Rannikmae, M. Grade 6 & 9 student and teacher perceptions of teaching and learning approaches in relation to student perceived interest/enjoyment towards science learning. J. Baltic Sci. Educ. 2021, 20, 119–133. [CrossRef]
- McConney, A.; Woods-McConney, A. Inquiry-based, teacher-directed and adaptive instruction in secondary science: A crossnational analysis of associations with science literacy and interest using PISA2015. In Proceedings of the ICERI2019 Conference, Seville, Spain, 11–13 November 2019.
- Inkinen, J.; Klager, C.; Juuti, K.; Schneider, B.; Salmela-Aro, K.; Krajcik, J.; Lavonen, J. High school students' situational engagement associated with scientific practices in designed science learning situations. Sci. Educ. 2020, 104, 667–692. [CrossRef]
- Juuti, K.; Lavonen, J.; Salonen, V.; Salmela-Aro, K.; Schneider, B.; Krajcik, J. A Teacher–Researcher Partnership for Professional Learning: Co-Designing Project-Based Learning Units to Increase Student Engagement in Science Classes. J. Sci. Teach. Educ. 2021, 32, 625–641. [CrossRef]
- Salmela-Aro, K.; Upadyaya, K.; Cumsille, P.; Lavonen, J.; Avalos, B.; Eccles, J. Momentary task-values and expectations predict engagement in science among Finnish and Chilean secondary school students. Int. J. Psychol. 2020, 56, 415–424. [CrossRef]
- Upadyaya, K.; Cumsille, P.; Avalos, B.; Araneda, S.; Lavonen, J.; Salmela-Aro, K. Patterns of situational engagement and task values in science lessons. J. Educ. Res. 2021, 114, 394–403. [CrossRef]
- Lee, S.W.; Luan, H.; Lee, M.; Chang, H.; Liang, J.; Lee, Y.; Lin, T.; Wu, A.; Chiu, Y.; Tsai, C. Measuring epistemologies in science learning and teaching: A systematic review of the literature. *Sci. Educ.* 2021, 105, 880–907. [CrossRef]
- Schiefer, J.; Bernholt, A.; Kampa, N. A closer look at elementary students' epistemic beliefs—latent profiles capturing concepts of knowledge and knowing in science. *Learn. Ind. Diff.* 2021, 92, 16. [CrossRef]
- She, H.; Lin, H.; Huang, L. Reflections on and implications of the Programme for International Student Assessment 2015 (PISA 2015) performance of students in Taiwan: The role of epistemic beliefs about science in scientific literacy. J. Res. Sci. Teach. 2019, 56, 1309–1340. [CrossRef]
- OECD. PISA 2015. PISA 2015 Technical Report. Available online: http://www.oecd.org/pisa/data/2015-technical-report/ (accessed on 12 October 2021).
- Juuti, K.; Loukomies, A.; Lavonen, J. Interest in Dialogic and Non-Dialogic Teacher Talk Situations in Middle School Science Classroom. Int. J. Sci. Math. Educ. 2019, 18, 1531–1546. [CrossRef]
- Alatlı, B.K. Investigation of Factors Associated with Science Literacy Performance of Students by Hierarchical Linear Modeling: PISA 2015 Comparison of Turkey and Singapore. *Egiptim Ve Bilim Educ. Sci.* 2020, 45, 17–49. [CrossRef]
- OECD. PISA 2015. Student Questionnaire, PISA 2015. Available online: https://www.oecd.org/pisa/data/CY6_QST_MS_STQ_ CBA_Final.pdf (accessed on 12 October 2021).
- Ma, X.; Wu, X.; Yuan, J.; Luo, X. Students, classrooms, teachers, and schools: Competing effects on science achievement. *Teach. Coll. Rec.* 2018, 120, 48. [CrossRef]
- 43. OECD. PISA 2015. PISA 2015 Database. Available online: https://www.oecd.org/pisa/data/2015database/ (accessed on 13 October 2021).

- IBM. SPSS—26 Release Specifications. Available online: https://www.ibm.com/support/pages/release-notes-ibm%C2%AE-spss%C2%AE-statistics-260 (accessed on 13 October 2021).
- 45. Valladares, L. Scientific Literacy and Social Transformation. Sci. Educ. 2021, 30, 557–587. [CrossRef]
- Aditomo, A.; Köhler, C. Do student ratings provide reliable and valid information about teaching quality at the school level? Evaluating measures of science teaching in PISA 2015. *Educ. Assess. Eval. Acc.* 2020, 32, 275–310. [CrossRef]
- Ma, X.; Ma, L.; Bradley, K. Using multilevel modeling to investigate school effects. In *Multilevel Modeling of Educational Data*; O'Connell, A.A., McCoach, D.B., Eds.; Information Age: Charlotte, NC, USA, 2008; pp. 59–110.
- Raudenbush, S.W.; Bryk, A.S. Hierarchical Linear Models: Applications and Data Analysis Methods, 2nd ed.; SAGE: Thousand Oaks, CA, USA, 2002; Volume 1.
- Scientific Software International. HLM 8 for Windows. Available online: http://www.ssicentral.com/index.php/product/hlm (accessed on 13 October 2021).
- 50. Cohen, J. Statistical Power Analysis for the Behavioral Sciences, 2nd ed.; Lawrence Erlbaum: Hillsdale, NJ, USA, 1988.
- Scherer, R. The case for good discipline? Evidence on the interplay between disciplinary climate, socioeconomic status, and science achievement from PISA 2015. In *Equity, Equality and Diversity in the Nordic Model of Education*; Frønes, T.S., Pettersen, A., Radišić, J., Buchholtz, N., Eds.; Springer: Cham, Switzerland, 2020. [CrossRef]
- Chi, S.; Liu, X.; Wang, Z.; Han, S.W. Moderation of the effects of scientific inquiry activities on low SES students' PISA 2015 science achievement by school teacher support and disciplinary climate in science classroom across gender. *Int. J. Sci. Educ.* 2018, 40, 1284–1304. [CrossRef]
- Kang, J. Interrelationship Between Inquiry-Based Learning and Instructional Quality in Predicting Science Literacy. Res. Sci. Educ. 2020, 1–17. [CrossRef]
- Lavonen, J.; Angell, C.; Byman, R.; Henriksen, E.K.; Koponen, I.T. Social Interaction in Upper Secondary Physics Classrooms in Finland and Norway: A survey of students' expectations. *Scand. J. Educ. Res.* 2007, *51*, 81–101. [CrossRef]
- Lonka, K.; Ketonen, E.; Vermunt, J.D. University students' epistemic profiles, conceptions of learning, and academic performance. *High. Educ.* 2020, *81*, 775–793. [CrossRef]
- Park, S.; Kite, V.; Suh, J.K.; Jung, J.; Rachmatullah, A. Investigation of the relationships among science teachers' epistemic orientations, epistemic understanding, and implementation of Next Generation Science Standards science practices. J. Res. Sci. Teach. 2021. [CrossRef]
- 57. Hampden-Thompson, G.; Bennett, J. Science Teaching and Learning Activities and Students' Engagement in Science. Int. J. Sci. Educ. 2013, 35, 1325–1343. [CrossRef]
- Kumpulinen, K. Teachers and Principals in Finland 2016. Available online: https://www.oph.fi/sites/default/files/documents/ opettajat_ja_rehtorit_suomessa_2016_0.pdf (accessed on 15 October 2021).
- Rupérez, F.L.; García, I.G.; Expósito-Casas, E. Educational Effectiveness, Efficiency, and Equity in Spanish Regions: What does PISA 2015 Reveal? Orb. Sch. 2019, 12, 9–36. [CrossRef]
- Van Laere, E.; Aesaert, K.; Van Braak, J. The Role of Students' Home Language in Science Achievement: A multilevel approach. Int. J. Sci. Educ. 2014, 36, 2772–2794. [CrossRef]
- Grabau, L.J. Science Education: Lessons from Abroad, Drawn from PISA-2015; American Society of Agronomy: Salt Lake City, UT, USA, 2021.
- Agirdag, O.; VanLaar, G. Does more exposure to the language of instruction lead to higher academic achievement? A crossnational examination. Int. J. Biling. 2016, 22, 123–137. [CrossRef]
- Goodrich, J.M.; Thayer, L.; Leiva, S. Evaluating Achievement Gaps Between Monolingual and Multilingual Students. *Educ. Res.* 2021, 50, 429–441. [CrossRef]
- Smith, P.; Frazier, P.; Lee, J.; Chang, R. Incongruence Between Native and Test Administration Languages: Towards Equal Opportunity in International Literacy Assessment. Int. J. Test. 2017, 18, 276–296. [CrossRef]
- Fung, D. Teaching Science Through Home and Second Languages as the Medium of Instruction: A Comparative Analysis of Junior Secondary Science Classrooms in Hong Kong. Int. J. Sci. Math. Educ. 2020, 19, 1609–1634. [CrossRef]





Article Sustainable Education with Local-Wisdom Based Natural Reagent for Green Chemical Analysis with a Smart Device: Experiences in Thailand

Kanokwan Kiwfo^{1,2}, Chonnipa Yeerum³, Piyanat Issarangkura Na Ayutthaya³, Kullapon Kesonkan³, Siripat Suteerapataranon¹, Piyatida Panitsupakamol⁴, Dujrudee Chinwong^{2,5}, Pathinan Paengnakorn^{1,6}, Surarong Chinwong^{2,5}, Narong Kotchabhakdi^{1,7}, Chalermpong Saenjum^{1,2,8,*}, Monnapat Vongboot^{1,3,*} and Kate Grudpan^{1,2,9}

- ¹ Center of Excellence for Innovation in Analytical Science and Technology (I-ANALY-S-T), Chiang Mai University, Chiang Mai 50200, Thailand; k.kanokwan11@gmail.com (K.K.); siripat.sutee@gmail.com (S.S.); pathinan.p@cmu.ac.th (P.P.); kotchabhak_n@hotmail.com (N.K.); kgrudpan@gmail.com (K.G.)
- ² Cluster of Excellence on Biodiversity-Based Economic and Society (B.BES-CMU), Chiang Mai University, Chiang Mai 50200, Thailand; dujrudee.c@cmu.ac.th (D.C.); surarong@gmail.com (S.C.)
- ³ Department of Chemistry, Faculty of Science, King Mongkut's University of Technology Thonburi, Bangkok 10140, Thailand; chonnipa.yeerum@gmail.com (C.Y.); piyanat.tp@gmail.com (P.I.N.A.); kullapon.kesonkan@gmail.com (K.K.)
- ⁴ Department of Pharmacy, Maharaj Nakorn Chiang Mai Hospital, Chiang Mai 50200, Thailand; piyatida.panit@cmu.ac.th
- ⁵ Department of Pharmaceutical Care, Faculty of Pharmacy, Chiang Mai University, Chiang Mai 50200, Thailand
- ⁶ Biomedical Engineering Institute, Chiang Mai University, Chiang Mai 50200, Thailand
- ⁷ Faculty of Science, Lampang Rajabhat University, Lampang 52100, Thailand
- ⁸ Department of Pharmaceutical Sciences, Faculty of Pharmacy, Chiang Mai University, Chiang Mai 50200, Thailand
- Department of Chemistry, Faculty of Science, Chiang Mai University, Chiang Mai 50200, Thailand
- * Correspondence: chalermpong.s@cmu.ac.th (C.S.); sumalee.tan@kmutt.ac.th (M.V.)

Abstract: To minimize chemical waste and protect the environment, our team has used green analysis with natural reagents. In this work, we designed a natural-reagent assay kit for iron determination and implemented it in chemistry education in Thailand. The iron assay method was adapted from Thai local wisdom of testing water quality using guava leaves. The guava leaf powder served as a natural reagent in the assay. The kit included equipment, standard and buffer solutions and a manual. A smart device with a built-in camera was used as a detector. Educators in six universities in Thailand implemented the kit in laboratories with modifications depending on their learning outcomes. The kit implementation was evaluated using a survey with questions in four aspects: usability, learning achievement, green chemistry and portability. The high average scores for all questions (> 4.00 of 5.00 points), with the average overall score of 4.53 ± 0.60 , indicated satisfaction regarding in all aspects. Using a locally available bio-resource as a natural reagent for green analysis in chemistry education supported sustainable education in Thailand, in terms of quality education (SDG 4) and reduced inequalities (SDG 10) and environmental sustainability (SDG 6–Clean water and sanitation, 12–Responsible consumption and production and 14–Life be-low water).

Keywords: Thai local wisdom; green chemical analysis; natural reagent; sustainable education; chemistry education; sustainability development

1. Introduction

The green chemistry concept has provided approaches to reduce the risks of chemistry to the environment [1]. Within the twelve principles of green chemistry, proposed

Citation: Kiwfo, K.; Yeerum, C.; Issarangkura Na Ayutthaya, P.; Kesonkan, K.; Suteerapataranon, S.; Panitsupakamol, P.; Chinwong, D.; Paengnakorn, P.; Chinwong, S.; Kotchabhakdi, N.; et al. Sustainable Education with Local-Wisdom Based Natural Reagent for Green Chemical Analysis with a Smart Device: Experiences in Thailand. *Sustainability* **2021**, *13*, 11147. https:// doi.org/10.3390/su132011147

Academic Editor: Larry J. Grabau

Received: 12 September 2021 Accepted: 5 October 2021 Published: 9 October 2021

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2021 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). since 1998 by P. T. Anastas and J. C. Warner [2], analytical chemistry was a branch that was difficult to be environmentally friendly. In fact, some of the green chemistry principles, such as waste prevention, safer solvents and auxiliaries, design for energy efficiency, safer chemistry to minimize the potential of chemical accidents and development of instrumental methods, are directly related to analytical chemistry [3]. However, the need for hazardous chemical reagents for some steps in the analytical process, e.g., sample preservation and preparation and chemical analysis, have been one of the causes. Analytical chemistry have endeavored to develop greener chemical analyses, leading to more sustainable chemistry [4]. The Organisation for Economic Co-operation and Development (OECD) defined the term "Sustainable chemistry" as a scientific concept that seeks to improve the efficiency with which natural resources are used to meet human needs for chemical products and services. Sustainable chemistry encompasses the design, manufacture and use of efficient, effective, safe and more environmentally benign chemical products and processes [5].

Our research group interests, since 1991, are in the development of flow-based analysis (FBA) methods for various aspects of chemical analyses. The FBA is a green analytical method that could reduce reagent consumption and, therefore, waste production, from macro- to nano-volume. Moreover, using FBA helps save energy due to its simplicity and rapidity and prevents accidents due to its closed and, sometimes, remote system. However, the use of hazardous chemicals is unavoidable in some cases. The development of green chemical analysis methods utilizing natural reagents has, hence, become our attempt to meet the sustainable chemistry goals and three of the sustainable development goals (SDGs) identified in 2015 by the United Nations (UN), i.e., goal 6-Clean water and sanitation; goal 12-Responsible consumption and production; and goal 14-Life below water.

In 2005, our group reported the use of a natural reagent (guava leaf extract) as an alternative reagent for the flow injection determination of iron, based on Thai local wisdom [6,7]. This local wisdom has been passed down through generations. The ground guava leaves were put in water samples to test whether the water could be used for washing clothes. If the water turned dark purple, it indicated too much iron and the water should not be used. Recently, it was reported that these guava leaves were a source of phenolic compounds, responsible for binding to or chelating with metal ions such as Fe(III) ions [8]. Our group's article in 2005 found that the extract from guava leaves was a promising natural reagent for the iron determination. From the discovery in this publication, we have further adapted Thai local wisdom in detecting chemicals or elements in real samples to the modern analysis, for examples; using guava leaf extract for quality control of iron content in pharmaceutical formulations [8]; exploiting green tea extracts for the iron determination by flow injection analysis (FIA) [9]; determining iron(III) using *Phyllanthus* emblica Linn. as a natural reagent [10]; using Indian almond (Terminalia Catappa L.) leaf extract to determine aluminum in wastewater [11]; using Morinda citrifolia root to determine aluminum in tea [12] and the attempts succeeded [13–15].

In order to develop sustainability in chemistry education, in this work, the guava leaf extract, among those local wisdom-based innovations, has been investigated further and applied as a green reagent in chemistry experiments for undergraduate education in six Thai universities. The extract was transformed into a ready-to-use reagent powder and packed up with a set of simple experimental equipment and a user manual, as a natural reagent iron assay kit, for the experiment. Each university adapted the kit to achieve its aim of the experiment. Implementing the natural reagent iron assay kit in chemistry courses was studied. Satisfaction with the utilization of the developed assay kit was evaluated using a questionnaire consisting of nine questions, rating satisfaction level in four aspects: the usability of the developed assay kit, the learning achievement, the green chemistry and the portability of the developed assay kit.

2. Materials and Methods

In order to standardize the natural reagent from guava leaf, it was produced in a laboratory at Chiang Mai University, as the supplier and was sent to cooperating universities. Preparation of the natural reagent from guava leaf, assembling the natural-reagent iron assay kit and the implementation of the kit are described in this part.

2.1. Natural Reagent from Guava Leaf (in Brief)

2.1.1. Production of the Natural Reagent Powder

Firstly, fresh guava leaves were washed and dried in a hot air oven at 60 $^{\circ}$ C for 24 h. The dried leaves were then ground to powder and sieved to obtain 10–20 mesh powder. Finally, a ready-to-use natural reagent powder was obtained and packed in a tea bag and sealed to an air-tight opaque package to avoid exposure to air and light as shown in Figure 1a. The natural reagent powder from guava leaves for the iron assay was so-called Nat Pow Iron (G).



Figure 1. (a) Nat Pow Iron (G) sachet; (b) natural-reagent iron assay kit.

2.1.2. Preparation of the Natural-Reagent Iron Assay Kit

The natural-reagent iron assay kit (Figure 1b) consists of (i) a sachet of Nat Pow Iron (G), (ii) a vial for extracting the natural reagent, (iii) a vial containing the iron(III) standard, (iv) a vial containing buffer solution, (v) a 96 wells microplate as a reaction platform and (vi) instructions for the reagent preparation as shown in Supplementary Material: Figure S1, which can be accessed by scanning the QR code on a sachet of Nat Pow Iron (G). The chemicals and conditions used were optimized as reported previously [6].

2.1.3. Quality Control of the Nat Pow Iron (G)

The quality control of the Nat Pow Iron (G) was performed using high performance liquid chromatography (HPLC). Every lot of the Nat Pow Iron (G) was sampled for HPLC analysis of active ingredients before packing as the reagent for the iron assay kit. The analysis method is described elsewhere [16]. The major components in the natural reagent extract were phenolic compounds, namely gallic acid, catechin and ellagic acid, as illustrated in Figure 2. Users could check the quality of the Nat Pow Iron (G) by scanning the QR code on the Nat Pow Iron (G) sachet for the HPLC fingerprint of the reagent. The reagent data sheet is also shown in the Supplementary Material: Figure S1.



Figure 2. HPLC chromatograms of mixed phenolic compound standards and natural reagent from guava leaf. The peaks are (a): gallic acid; (b): gallocatechin; (c): epigallocatechin; (d): catechin; and (e): ellagic acid.

2.1.4. Using a Smart Device as a Detector

A dark purple complex is the product of the reaction between iron (III) and phenolic compounds in the natural reagent. Its color intensity is directly proportional to the amount of the iron (III). Thus, the amount of iron (III could be measured by detecting the difference in the color intensity of the reaction product. To promote equitable quality education (SDG 4-Quality education), simplifying the detection using a smart device with a built-in camera, i.e., a smartphone or a tablet, was done. A photograph of the product solutions in the well-microplate was taken by a student's smart device. The color value of each solution was analyzed using an image processing program on a computer or an application on the smart device such as ImageJ application, Chemical Sensor System by TRF and I-ANALY-S-T (CSS by TRF & I-ANALY-S-T).

2.2. Implementation of the Natural-Reagent Iron Assay Kit

The natural-reagent iron assay kit was demonstrated at six national universities in Thailand (Figure 3): (i) Faculty of Pharmacy, Chiang Mai University, (ii) Faculty of Science, King Mongkut's University of Technology Thonburi, (iii) Faculty of Science and Engineering, Kasetsart University- Chalermphrakiat Sakon Nakhon Province Campus, (iv) Faculty of Science, Khon Kaen University, (v) School of Science, Mae Fah Luang University and (vi) School of Science, University of Phayao. The use of the developed assay kit was recommended in the procedure manual as shown in Supplementary Material: Figure S2. However, it could be customized by users and course instructors to achieve their goals or the objectives of the experiment and to suit different groups of learners at each university with different working environments. Some components of the kits could be replaced by either suitable or available equipment such as liquid measuring and handling tools.



Figure 3. Implementation of the assay kits as a learning material for a green chemical analysis topic in 6 universities: (i) Chiang Mai University (20 students); (ii) King Mongkut's University of Technology Thonburi (25 students); (iii) Kasetsart University — Chalermphrakiat Sakon Nakhon Province Campus (11 students); (iv) Khon Kaen University (10 students); (v) Mae Fah Luang University (12 students); and (vi) University of Phayao (25 students).

Satisfaction level was evaluated using a self-administered questionnaire. The questionnaire was composed of nine issues with a 5-point Likert scale (1 = least satisfied, 2 = less satisfied, 3 = moderately satisfied, 4 = very satisfied, 5 = most satisfied). The issues covered (i) clearness and comprehensibility of the procedure manual, (ii) simplicity of the operation, (iii) suitability of the equipment, (iv) student's comprehensibility in the principle of the iron assay, (v) student's ability in transferring knowledge learned from this experiment to other students, (vi) operator safety, (vii) environmentally friendliness, (viii) feasibility of application for sampling-site chemical analysis and (ix) overall satisfaction with this experiment. The satisfaction was evaluated by 40 students enrolled in two chemistry subjects from two faculties, i.e., Instrumental Methods of Chemical Analysis Laboratory I (CHM363) course (20 students) at the Faculty of Science, King Mongkut's University of Technology Thonburi, Bangkok and Quality Control for Food and Cosmetics (461571) course (20 students) at the Faculty of Pharmacy, Chiang Mai University, Chiang Mai. The difference of the satisfaction scores evaluated by students between the two groups was analyzed using independent *t*-test, with a p < 0.05 as significant difference. It should be noted that there was no control group of students who used a standard traditional method for iron analysis.

Moreover, we randomly selected students and divided them into three groups of four for the focus group discussion, asking about the advantages, difficulties, limitations, satisfaction and suggestions for improvement. The focus group discussion for each group took approximately 15–20 minutes and the content analysis was conducted.

3. Results

3.1. Implementation of the Developed Assay Kit in On-Site Classes

The iron assay kit using the Nat Pow Iron (G) as a natural reagent was distributed to the six universities for implementation in a hands-on experiment of chemistry courses (Figure 3). As an example, Supplementary Material: Video S1 demonstrated the use of the developed assay kit in a chemistry laboratory at the Department of Chemistry, Faculty of Science, King Mongkut's University of Technology Thonburi. Because the procedure was designed at the beginning to be flexible, the instructors in each university were able to adjust the procedures, including adding steps and changing the provided equipment to available lab equipment, to achieve the specific learning outcomes. For example, to measure the volumes of solutions, the instructors may use different types of measuring equipment, e.g., autopipette, disposable medical syringe, or dropper. The autopipette was used at all six universities because it provided an accurate and precise volume. Four universities additionally used the disposable syringe, while one university used the dropper, both calibrated with the autopipette, to measure the volumes of the solutions. The Supplementary Material: Figure S3 illustrates the example of the laboratory instruction modified from the provided procedure manual.

Despite the fact that students used various smart devices with different operating systems (iOS and Android) and different camera quality, the assay accuracy, as measured by analysis of known-concentration samples, was not significantly different (statistical analysis by independent *t*-test, p = 1.00). Moreover, although the procedure manual provided in the assay kit recommended using a smart device as the detection device, the image processing application/program could actually be varied. This flexibility promoted students to learn the importance of chemical data analysis, especially the accuracy and precision of the analysis, which expressed the reliability and reproducibility of the analytical data. In all, three universities used the in-house image processing mobile application; Chemical Sensor System by TRF and I-ANALY-S-T (CSS by TRF & I-ANALY-S-T for Android-APK download) [17], whereas the other three universities used different mobile applications, including Pixel Picker [18], ImageJ [19], Color Detector [20] and Color Grab [21].

3.2. Evaluation of the Utilisation of the Developed Assay Kit as a Learning Material in Chemistry Courses

The responses from all 40 students regarding their satisfaction with the experiment using the natural-reagent iron assay kit as a learning material are shown in Figure 4. Overall, the satisfaction scores evaluated by 40 students from the chosen universities (two universities, 20 students each) showed that the average score for every question was higher than 4.00 (full score = 5.00) points, meaning very satisfied, with the average satisfaction on the natural-reagent iron assay kit of 4.55 ± 0.64 . The three issues that had the highest scores were the operator safety (4.78 ± 0.42), the environmental friendliness (4.70 ± 0.46) and the suitability of the equipment (4.63 ± 0.54), respectively. The issue with the lowest score (4.20 ± 0.61) was the student's ability in transferring knowledge learned from the experiment to other students. The score of the student's ability in transferring knowledge given by applied science students in the university U1 was surprisingly below 4.00 points, meaning moderately satisfied. Nevertheless, the satisfaction scores evaluated by pharmacy students (U1) were lower than those by chemistry students (U2) with the significant differences (statistically tested by independent *t*-test, *p* < 0.05) in all issues, except issue 6—Operator safety (*p* = 0.06).

However, the results from the focus group discussion showed that all students (12 students) were very satisfied with the green chemical analysis using the developed assay kit. Most students were impressed with the innovation in green chemical analysis. Some students said that "It was a novel laboratory investigation that used less volume of chemicals, and the chemical reagent was produced from a natural bio-resource. I think it is very good and also safe for students".



Figure 4. Mean satisfaction on the experiment using the natural-reagent iron assay kit as a learning material evaluated by students (n = 40).

4. Discussion

4.1. Implementation of the Natural-Reagent Iron Assay Kit in Chemistry Education

An academic application of the developed natural-reagent iron assay kit was demonstrated in chemistry education at six universities in Thailand and was used as learning material for an experiment of natural and applied science courses. Although the provided procedure manual in the developed assay kit clearly directed the use, it remained flexible enough to be modified according to the desired learning outcomes of the courses. However, all six universities followed the core procedure to achieve the completion of the expected chemical reaction. The modification was made only in the steps of solution preparation. Using droppers and disposable syringes, calibrated with autopipettes, allowed students to learn with creativity and open-mindedness. They would be able to adapt any kind of volume-measuring equipment available in their laboratory or place to measure accurate and precise volume by appropriately calibrating with standard volumemeasuring equipment.

In the current digital and mobile world, it could not be denied that students must possess one of the smart devices as a tool for learning and communication. The employment of smart devices as detection devices led to simplicity and even equity in education, supporting SDG 10 (Reduced inequalities). Instead of a spectrophotometer, which might be unavailable in some laboratories due to its cost, a smart phone could obviously detect the color intensity using its high-quality camera. The results showed that various smart devices with different operating systems and different camera quality did not exhibit a significant difference in the accuracy of the assay. Thus, every student would be able to analyze his or her experimental data and learn how to analyze analytical data, which is the seventh step in the chemical analytical process. Students were excited and mentioned that "The experiment was easy and fun"; It's an exotic lab. It's very interesting; I didn't think I could use my mobile phone for the experiment, and it was very interesting".

The experiment report could also be created in Microsoft Excel spreadsheet and submitted to the instructor via email or other online methods. This increased the convenience for both the students and the instructors and importantly created a paperless culture, the green mind. Moreover, as the instructors at a university created a cloud space, i.e., Google spreadsheet, for sharing the analytical results during the class, the results from students could be discussed at real-time in the class. Students were encouraged to critique as a result of this.

4.2. Evaluation of the Utilisation of the Developed Assay Kit as a Learning Material in Chemistry Courses

Because all of the students had prior experience using the traditional spectrophotometric method for other analyses, no control group was used in this study. The students were able to compare their prior experience with the developed assay kit. Overall, the evaluation results illustrated a high level of satisfaction (average overall score 4.53 ± 0.60) in the utilization of the developed assay kit as a learning material in chemistry courses. Hence, employing the assay kit in chemistry education is promising. However, the statistical t-test, comparing the satisfaction scores evaluated by pharmacy (U1) and chemistry students (U2), indicated the pharmacy students' satisfaction scores were significantly lower than those of the chemistry students in all aspects, except the operator safety. The reason was that the pharmacy students were concerned that the assay kit would not be able to be used in pharmaceutical analysis, which needs high levels of precision and accuracy following FDA regulations and pharmaceutical pharmacopoeia. The satisfaction scores for the operator safety evaluated by students in U1 and U2 did not significantly differ. It demonstrated that the assay kit was safe for pharmaceutical and chemistry research. Therefore, the green chemical analysis may be used as an alternative analysis in the pharmaceutical industry, but may be unable to be utilized as the primary analysis. However, some pharmacy students were impressed by the use of natural extract as an environmentally friendly reagent for iron analysis, the simplicity of the analytical procedure and the use of smartphones as the detection device instead of a spectrophotometer.

The questions in the satisfaction survey were categorized, for further discussion, in four aspects: usability of the developed assay kit (questions 1, 2 and 3), learning achievement (questions 4 and 5), green chemistry (questions 6 and 7) and portability (question 8) and were discussed as follows.

4.2.1. Usability of the Developed Assay Kit

The results showed the students were very satisfied with using the assay kit in terms of the clearness and comprehensibility of the procedure manual (score 4.55 ± 0.64), the simplicity of the operation (score 4.33 ± 0.57) and the suitability of the equipment (score 4.48 ± 0.60), indicating the simplicity and user friendliness of the assay kit. Some students expressed their impression as follows:

"It was my first time discovering that natural reagents could detect iron, and was a lot easier than I imagined, with no need of expensive and complicated laboratory equipment or a teacher's assistance. I am capable of doing all tasks on my own."

Nonetheless, while the majority of students encountered no problems, some did. For example, because of the transparency of the solutions and the small size of the microvolume well plate, they found it difficult to locate the wells and drop the solutions into the wells. They stated that "We got dizzy when we had to drop the solution into the well because it was so small".

Therefore, the developed assay kit could definitely be employed as a learning material for natural and applied chemistry courses. The developed assay kit showed its flexibility so that educators could modify the procedures in order to achieve their learning outcomes of the course.

4.2.2. Learning Achievement

Most students said in the focus group session that "The experiment can be linked to the lesson and make the lesson more interesting and memorable", which agreed with the evaluation of the developed assay kit. The average score of the issue "student's comprehensibility in the principle of the iron assay" was 4.43 ± 0.64 , meaning very satisfied. However, the average score for the issue "Student's ability in transferring knowledge learned from this experiment to other students" was the lowest score (4.20 \pm 0.61), but its interpretation was very satisfactory. According to the Learning Theory Pyramid, students will gain approximately 90% of their knowledge by teaching others, resulting in a deeper understanding and long-term memory, as opposed to learning through traditional passive methods such as lecture and reading, which only gain 5-30% of the knowledge. Furthermore, some responses from the participating students via an online survey showed that they were able to learn new things related to green chemical analysis, such as how to colorimetrically determine iron using a natural reagent, accuracy and precision, perform green analytical chemistry and obtain information technology (IT) skills. Therefore, hands-on experiments with simple but intense learning material, such as the developed assay kit, helped promote deeper understanding and long-term memory quite well. Notably, students may not have the opportunity to practice standard laboratory skills such as measuring the volumes of solutions with pipettes and preparing solutions in volumetric flasks. However, these constraints came with more benefits in terms of sustainability.

4.2.3. Green Chemistry

As mentioned earlier, the average satisfaction scores for the green chemistry aspect, i.e., the issues "Operator safety" and "Environmental friendliness", were the highest scores (4.78 ± 0.42 and 4.70 ± 0.46 points, respectively). It indicated that students were deeply concerned with the hazards of chemicals that could harm the operator's and the environmental health. It could be implied that utilization of natural reagents, based on Thai local wisdom, in chemistry education had raised students' awareness of "Responsible consumption and production"—SDG 12, which is concerned with responsible use of natural resources to avoid harmful effects to the environment.

This proposed green chemistry experiment achieved our research group's aim to meet SDG 6—Clean water and sanitation and SDG 14—Life below water successfully. It helped ensure the availability and sustainable management of water sanitation (SDG 6) by reducing the hazardous wastewater released to water resources. Consequently, it conserved the oceans, sea and marine resources (SDG 14). Furthermore, not only does using natural reagents extracted from natural resources in Thailand reduce environmental hazards, but it also reduces inequality within and among countries (SDG 10—Reduced inequalities) by producing cost-effective natural reagents in Thailand, as well as minimizing the relatively high-cost chemical reagents imported from abroad, which some organizations rely on. Referring to the interview of Emeritus Professor in analytical chemistry, Prof. Dr. K. Grudpan, published in a national newspaper [22], the chemical reagents imported from abroad could cost four times more than those produced by using the bioresources in Thailand. Using natural extracts as natural reagents for chemical analyses could save up to 10 million THB and save time according to the duration of long-distance transportation (up to six months) [23].

Nevertheless, because natural reagents are non-toxic and produced from a locally available bio-resource, their use for chemistry promotes the green chemistry principles 5—safer solvents and auxiliaries and 7—use of renewable feedstocks, respectively. This approach also raised awareness about the importance of utilizing green chemistry and natural reagents in experiments to save the environment and ensure sustainable education and development. This agrees with target 4.7 of SDG 4, stating "By 2030, ensure that all learners acquire the knowledge and skills needed to promote sustainable development".

4.2.4. Portability and Future Application

The satisfaction score for the issue "Feasibility of application for sampling-site chemical analysis" was high (4.63 \pm 0.54). It meant that students found the developed assay kit convenient and highly feasible for sampling-site iron analysis. Students also expressed that the developed assay kit did not require expensive and complicated scientific instruments and, thus, could be easily carried to the site. In addition, unlike the traditional analysis method for iron assay, all steps of the analysis using the developed assay kit did not have to be performed in a laboratory because there were fewer hazardous chemicals and harmful procedures.

In our further studies, the developed assay kit could be implemented in an online chemistry laboratory class. As a result of the COVID-19 pandemic, all on-site academic activities were suspended and online learning became a feasible alternative. Science educators and also students may have difficulty with laboratory lessons because students may not have the necessary equipment at home and could not afford to obtain it. Therefore, educators still need to design efficient and effective classes to continue achieving the learning outcomes of the courses. Higher education in chemistry requires hands-on laboratories to allow students to develop laboratory and 21st-century learning skills, i.e., learning and innovation, information, media and technology and life and career. During the first wave of the COVID-19 outbreak, an educator in our research team constructed an online chemistry laboratory using the developed assay kit as a learning tool [22,23]. The developed assay kit was sent to students' homes so that all students enrolled in the course could conduct the analysis in an online laboratory class at their homes. This study will be presented as "Lab-at-Home" in the next series of the development of the natural-reagent iron assay kit based on Thai local wisdom.

5. Conclusions

Much Thai local wisdom, for over hundreds of years, has used Thai natural resources to test or analyze for poisons or toxins. Our research group has been working on utilizing this valuable wisdom to develop green chemical analysis as a green innovation. Moreover, the developed green chemical analysis method was implemented as a learning tool for chemistry education. This work presented an example of developing green innovation adapting the wisdom of detecting iron element in water using ground guava leaf extract to analyze iron(III) ion in a laboratory. The guava leaf extract was served as a natural reagent for this purpose. A set of a sachet of the ground and dried guava leaf powder (Nat Pow Iron (G)), an iron(III) standard solution, a buffer solution, a 96-wells microplate and a procedure manual were packed as a natural-reagent iron assay kit and used for a chemistry experiment of chemistry classes in six Thai universities. The detection of colored complex, the product of the chemical reaction, was simplified using smart devices with built-in cameras, instead of a relatively high-cost spectrophotometer. The instructors of the courses could modify the procedure to achieve the learning outcomes of the courses. The results showed that students were remarkably satisfied with the natural-reagent iron assay kit in terms of the usability of the developed assay kit, learning achievement, green chemistry and portability. Utilizing the developed assay kit supported SDG 4 (quality education), SDG 6 (clean water and sanitation), SDG 10 (reduced inequalities), SDG 12 (responsible consumption and production) and SDG 14 (life below water) and promoted sustainable chemistry education.

Supplementary Materials: The following are available online at https://www.mdpi.com/article/10 .3390/su132011147/s1, Figure S1: instruction for the reagent preparation, Figure S2: provided proce-

dure manual, Figure S3: example of the laboratory instruction modified from the provided procedure manual, Video S1: use of natural-reagent iron assay kit in the Instrumental Methods of Chemical Analysis Laboratory I (CHM363) course at Department of Chemistry, Faculty of Science, King Mongkut's University of Technology Thonburi in academic year 2019.

Author Contributions: Concept and design: K.G., M.V., C.S. and K.K. (Kanokwan Kiwfo); Experiments and onsite interview: M.V., C.S., K.K. (Kanokwan Kiwfo), P.P. (Pathinan Paengnakorn), K.K. (Kullapon Kesonkan), C.Y., P.I.N.A. and D.C.; Data curation and analysis: M.V., C.S., K.K. (Kanokwan Kiwfo), P.P. (Pathinan Paengnakorn), K.K. (Kullapon Kesonkan), C.Y., P.I.N.A., N.K., D.C. and S.C.; Statistical analysis and interpretation: S.S., D.C., S.C. and P.P. (Piyatida Panitsupakamol); Writing—original draft preparation: K.K. (Kanokwan Kiwfo), P.P. (Piyatida Panitsupakamol), S.S., P.P. (Pathinan Paengnakorn), D.C., S.C., C.S. and M.V.; Writing—review and editing: S.S., D.C., S.C., P.P. (Piyatida Panitsupakamol) and K.G.; Funding Acquisition: K.G. and C.S. All authors have read and agreed to the published version of the manuscript.

Funding: This research was funded by the Center of Excellence for Innovation in Analytical Science and Technology (I-ANALY-S-T) and Cluster of Excellence on Biodiversity-Based Economics and Society (B.BES-CMU), Chiang Mai University, Chiang Mai, 50200, Thailand. This research was also supported by Distinguished Research Professor Award in "Green Innovation in Chemical Analysis with Local Wisdom" (Grant No. DPG6080002; K. Grudpan), Thailand Research Found (TRF), Thailand.

Institutional Review Board Statement: Ethical review and approval were waived for this study because the study was conducted in the established educational settings, involving normal instructional practices that were unlikely to have a negative impact on students' learning.

Informed Consent Statement: Informed consent was not applicable because this study was waived for ethical approval.

Data Availability Statement: The data presented in this study are available on reasonable request from the corresponding author.

Acknowledgments: Six universities, namely Chiang Mai University (Chalermpong Saenjum), King Mongkut's University of Technology Thonburi (Monnapat Vongboot), Kasetsart University-Chalermphrakiat Sakon Nakhon Province Campus (Saiphon Chanpaka), Khon Kaen University (Rodjana Burakham), Mae Fah Luang University (Kanchana Watla-iad), University of Phayao (Lucksa-goon Ganranoo), are acknowledged for their contributions in the implementation of the developed assay kit in chemistry courses. The authors are grateful to the Department of Pharmaceutical Sciences, Faculty of Pharmacy, Chiang Mai University and the Department of Chemistry, Faculty of Science, King Mongkut's University of Technology Thonburi, for their supports and assistance. We also acknowledge Thomas McManamon for language editing and proofreading the article.

Conflicts of Interest: The authors declare no conflict of interest.

References

- 1. Keith, L.H.; Gron, L.U.; Young, J.L. Green Analytical Methodologies. Chem. Rev. 2007, 107, 2695–2708. [CrossRef] [PubMed]
- Anastas, P.T.; Warner, J.C. Green Chemistry: Theory and Practice; Oxford University Press: Oxford, UK; New York, NY, USA, 1998; ISBN 0-19-850234-6.
- Koel, M.; Kaljurand, M. Application of the Principles of Green Chemistry in Analytical Chemistry. Pure Appl. Chem. 2006, 78, 1993–2002. [CrossRef]
- Tobiszewski, M.; Mechlińska, A.; Namieśnik, J. Green Analytical Chemistry: Theory and Practice. Chem. Soc. Rev. 2010, 39, 2869–2878. [CrossRef] [PubMed]
- Sustainable Chemistry OECD. Available online: https://www.oecd.org/chemicalsafety/risk-management/sustainablechemistry. htm (accessed on 8 September 2021).
- Settheeworrarit, T.; Hartwell, S.K.; Lapanatnoppakhun, S.; Jakmunee, J.; Christian, G.D.; Grudpan, K. Exploiting Guava Leaf Extract as an Alternative Natural Reagent for Flow Injection Determination of Iron. *Talanta* 2005, 68, 262–267. [CrossRef] [PubMed]
- I-ANALY-S-T CMU. Green Innovation in Chemical Analysis with Local Wisdom: EP1 from Local to Global. 2020. Available online: https://www.youtube.com/watch?v=Kka-OAzOigY (accessed on 1 September 2021).
- Siriangkhawut, W.; Ponhong, K.; Grudpan, K. A Green Colorimetric Method Using Guava Leaves Extract for Quality Control of Iron Content in Pharmaceutical Formulations. *Malays. J. Anal. Sci.* 2019, 23, 595–603.

- Pinyou, P.; Kradtap Hartwell, S.; Jakmunee, J.; Lapanantnoppakhun, S.; Grudpan, K. Flow Injection Determination of Iron Ions with Green Tea Extracts as a Natural Chromogenic Reagent. *Anal. Sci. Int. J. Jpn. Soc. Anal. Chem.* 2010, 26, 619–623. [CrossRef] [PubMed]
- Jaikrajang, N.; Kruanetr, S.; Harding, D.J.; Rattanakit, P. A Simple Flow Injection Spectrophotometric Procedure for Iron(III) Determination Using Phyllanthus Emblica Linn. as a Natural Reagent. Spectrochim. Acta A Mol. Biomol. Spectrosc. 2018, 204, 726–734. [CrossRef] [PubMed]
- Insain, P.; Khonyoung, S.; Sooksamiti, P.; Lapanantnoppakhun, S.; Jakmunee, J.; Grudpan, K.; Zajicek, K.; Kradtap Hartwell, S. Green Analytical Methodology Using Indian Almond (*Terminalia Catappa L.*) Leaf Extract for Determination of Aluminum Ion in Waste Water from Ceramic Factories. *Anal. Sci. Int. J. Jpn. Soc. Anal. Chem.* 2013, 29, 655–659. [CrossRef] [PubMed]
- 12. Tontrong, S.; Khonyoung, S.; Jakmunee, J. Flow Injection Spectrophotometry Using Natural Reagent from Morinda Citrifolia Root for Determination of Aluminium in Tea. *Food Chem.* **2012**, *132*, 624–629. [CrossRef] [PubMed]
- 13. Grudpan, K.; Hartwell, S.K.; Lapanantnoppakhun, S.; McKelvie, I. The Case for the Use of Unrefined Natural Reagents in Analytical Chemistry A Green Chemical Perspective. *Anal. Methods* **2010**, *2*, 1651–1661. [CrossRef]
- Grudpan, K.; Hartwell, S.K.; Wongwilai, W.; Grudpan, S.; Lapanantnoppakhun, S. Exploiting Green Analytical Procedures for Acidity and Iron Assays Employing Flow Analysis with Simple Natural Reagent Extracts. *Talanta* 2011, 84, 1396–1400. [CrossRef] [PubMed]
- Lapanantnoppakhun, S.; Tengjaroensakul, U.; Mungkornasawakul, P.; Puangpila, C.; Kittiwachana, S.; Saengtempiam, J.; Hartwell, S.K. Green Analytical Chemistry Experiment: Quantitative Analysis of Iron in Supplement Tablets with Vis Spectrophotometry Using Tea Extract as a Chromogenic Agent. J. Chem. Educ. 2020, 97, 207–214. [CrossRef]
- Saenjum, C.; Pattananandecha, T.; Nakagawa, K. Detection of Antioxidant Phytochemicals Isolated from Camellia Japonica Seeds Using HPLC and EPR Imaging. *Antioxidants* 2020, 9, 493. [CrossRef] [PubMed]
- CSS by TRF & I-ANALY-S-T for Android APK Download. Available online: https://apkpure.com/css-by-trf-i-analy-s-t/com. imagecolorpicker (accessed on 10 September 2021).
- Pixel Picker–Image Color Picker. Available online: https://apps.apple.com/us/app/pixel-picker-image-color-picker/id9308043 27 (accessed on 10 September 2021).
- Schneider, C.A.; Rasband, W.S.; Eliceiri, K.W. NIH Image to ImageJ: 25 Years of Image Analysis. Nat. Methods 2012, 9, 671–675. [CrossRef] [PubMed]
- Color Detector—Color Picker. Available online: https://apps.apple.com/us/app/color-detector-color-picker/id1478504229 (accessed on 10 September 2021).
- 21. SI, U.T. Color Grab (Android). Available online: https://color-grab.en.uptodown.com/android (accessed on 10 September 2021).
- CMU Researcher Creates Green Innovation in Chemical Analysis from Century-Old Wisdom of the Fang People (In Thai:นักวิจัย มช.สร้างนวัตกรรมสี่เขียวในการวิเคราะห์ทางเคมี จากภูมิปัญญาบรรพชนชาวฝาง). Available online: https://www.matichon.co.th/region/news_ 2748857 (accessed on 10 September 2021).
- Outstanding Research Professor Develops Green Innovation for Chemical Analysis Based on the Fang People's Century-Old Wisdom. (In Thai: ตาสตราจารย์วิจัยดีเด่นสร้างนวัตกรรมสีเซียว วิเคราะห์ทางเคม็จากภูมิปัญญาบรรพชนฝาง). Available online: https://tsri.or.th/th/ news/content/601/นวัตกรรมสีเซียวในการวิเคราะห์ทางเคม็กบัฏูมิปัญญาท้องถิ่น (accessed on 10 September 2021).





Article Lab-at-Home: Hands-On Green Analytical Chemistry Laboratory for New Normal Experimentation

Chonnipa Yeerum ¹, Piyanat Issarangkura Na Ayutthaya ¹, Kullapon Kesonkan ¹, Kanokwan Kiwfo ^{2,3}, Siripat Suteerapataranon ², Piyatida Panitsupakamol ⁴, Pathinan Paengnakorn ^{2,5}, Dujrudee Chinwong ^{2,6}, Surarong Chinwong ^{2,6}, Chalermpong Saenjum ^{2,7,*}, Monnapat Vongboot ^{1,2,*} and Kate Grudpan ^{2,3}

- ¹ Department of Chemistry, Faculty of Science, King Mongkut's University of Technology Thonburi, Bangkok 10140, Thailand; chonnipa.yee@mail.kmutt.ac.th (C.Y.); piyanat.tp@gmail.com (P.I.N.A.); kullapon.kes@mail.kmutt.ac.th (K.K.)
- ² Center of Excellence for Innovation in Analytical Science and Technology for Biodiversity-Based Economic and Society (I-ANALY-S-T_B.BES-CMU), Chiang Mai University, Chiang Mai 50200, Thailand; kanokwan_kiwfo@cmu.ac.th (K.K.); siripat.sutee@gmail.com (S.S.); pathinan.p@cmu.ac.th (P.P.); dujrudee.c@cmu.ac.th (D.C.); surarong@gmail.com (S.C.); kgrudpan@gmail.com (K.G.)
- ³ Department of Chemistry, Faculty of Science, Chiang Mai University, Chiang Mai 50200, Thailand
- ⁴ Department of Pharmacy, Maharaj Nakorn Chiang Mai Hospital, Chiang Mai 50200, Thailand; piyatida.panit@cmu.ac.th
- ⁵ Biomedical Engineering Institute, Chiang Mai University, Chiang Mai 50200, Thailand
- ⁶ Department of Pharmaceutical Care, Faculty of Pharmacy, Chiang Mai University, Chiang Mai 50200, Thailand
- ⁷ Department of Pharmaceutical Sciences, Faculty of Pharmacy, Chiang Mai University, Chiang Mai 50200, Thailand
- * Correspondence: chalermpong.s@cmu.ac.th (C.S.); sumalee.tan@kmutt.ac.th (M.V.)

Abstract: The COVID-19 pandemic has forced analytical chemistry educators in Thailand to change methods of teaching and learning to new normal ones. Higher education has faced additional challenges because of a lack of hands-on experiments and an increasing number of students in foundation chemistry courses being hindered from practicing skills. This work aimed to develop a Lab-at-Home (LAH) for new normal, analytical chemistry experimentation. The LAH implemented a hands-on green chemistry experiment, i.e., colorimetric determination of iron using non-hazardous reagents (supporting Sustainable Development Goal (SDG) 12-responsible consumption and production). The LAH was sent to students at their location before the synchronous class, where the instructors were prompt to supervise. Thus, this supports SDG4-quality education and SDG10-reduced inequalities. The learning outcome achievements, i.e., the analytical characteristics and colorimetry principles comprehension, as well as the ability to perform data analysis, were evaluated by a quiz and laboratory report. LAH satisfaction was assessed by questionnaire and focus group discussion. The learning outcomes were successfully achieved, although students who performed the experiment individually received higher scores than those who did in groups. Students were very satisfied with the LAH as a tool for new normal experimentation, yet some students faced a poor Internet connection during the synchronous online class.

Keywords: Lab-at-Home; new normal experimentation; green analytical chemistry; hands-on remote learning; higher education; sustainability development

1. Introduction

During the COVID-19 pandemic, the forced closure of workplaces since 2019 (including universities, where thousands of faculty staff and students meet regularly), were undeniable. Consequently, higher education methods have suddenly changed from onsite to remote learning, in order to avoid the spread of the coronavirus among people. Educators, especially in the sciences, have had the most frustrating experiences because

Citation: Yeerum, C.; Issarangkura Na Ayutthaya, P.; Kesonkan, K.; Kiwfo, K.; Suteerapataranon, S.; Panitsupakamol, P.; Paengnakorn, P.; Chinwong, D.; Chinwong, S.; Saenjum, C.; et al. Lab-at-Home: Hands-On Green Analytical Chemistry Laboratory for New Normal Experimentation. *Sustainability* **2022**, *14*, 3314. https:// doi.org/10.3390/su14063314

Academic Editor: Larry J. Grabau

Received: 8 February 2022 Accepted: 8 March 2022 Published: 11 March 2022

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/).

95

they had to design remote learning lessons for theoretical and practical classes, aiming to achieve scientific learning outcomes. Laboratory practice is essential in chemistry education because graduates in chemistry need laboratory skills for their careers, i.e., the abilities to conduct experiments, analyze experimental data, and interpret the results.

Kelley has recently reviewed articles reporting the laboratory learning in chemistry and closely related disciplines during the COVID-19 pandemic [1]. She aimed to determine the findings about the types of adjustments made to laboratory curricula and the immediate effect of these adjustments on students. The 91 reviewed articles illustrated the experiences of sudden transmission of teaching and learning methods from January 2020–June 2021 in a variety of institutions in and outside the US, e.g., Australia, Germany, Mexico, Slovakia, Spain, and the UK. Types of laboratory curricular adjustments that were diverted, for example, include (1) adapting experiments to be hands-on, at-home activities with household/school/commercial kit supplies, followed by some form of analysis (29.7%); (2) distributing procedures and non-interactive videos/pictures of experiments, from which students might or might not collect data themselves (50.5%); and (3) using digital simulations, models, augmented reality/virtual reality (AR/VR), games, or interactive videos, from which students might or might not collect simulated/real data themselves (45.1%). The findings evidenced that hands-on experience may improve technical performance (i.e., analyzing data, proposing a procedural step) but did not necessarily impact non-technical performance, such as writing a lab report and answering conceptual questions. Lower comprehension of data and procedures, as well as greater discomfort or poorer quality of interpretation, was found among students without hands-on experience compared with students assigned to perform the hands-on laboratory experiments. Moreover, students commented that performing hands-on work was more helpful for learning than remote alternative activities; however, doing hands-on experiments in a face-to-face laboratory was required. Therefore, to develop students' technical skills efficiently during the closure of the university, chemistry education curricular adjustments with hands-on laboratories were undeniable.

The diversity of hands-on, at-home laboratories for chemistry and related fields, i.e., analytical chemistry, organic chemistry, inorganic chemistry, biochemistry, general chemistry, and chemical engineering, has been reported [2–10]. Since we could not avoid using chemicals in the chemistry laboratory, the chemicals used in the designed hands-on, at-home laboratories must be safe for students and their family members as well as friendly to the environment, so that those chemicals could be discarded as household waste. Therefore, the chemicals used in those proposed laboratories were either kitchen chemicals or chemicals available at pharmacies that were disposable after use.

Analytical chemistry laboratories, especially for second-year-up undergraduate students, aim to encourage students to learn the principles of the analytical process, the steps for performing an analysis, i.e., (1) define the problem, (2) select an analytical method, (3) obtain a representative sample, (4) prepare the sample for the analysis, (5) perform any necessary separations, (6) perform the measurement, and (7) calculate the results and report [11] the analytical characteristics, e.g., accuracy, precision, calibration graph, etc., as well as the principles of instrumental analysis, for example, colorimetry, advanced spectrophotometry, chromatography, and electrochemistry. To meet the learning outcomes in analytical chemistry during the closure of universities, the instructors of analytical chemistry subjects have to design hands-on laboratories that students could perform at home. It should, again, be emphasized that the chemicals used must be less toxic, or even non-toxic, and disposable as household waste. Nevertheless, in general, chemical analyses require chemical reagents to react with analytes, resulting in chemical reaction products that are detected using an analytical method. Thus, the hands-on analytical chemistry experiments for at-home laboratories have been challenging to design. Natural extracts are alternative environmentally friendly chemical reagents that were proposed for a variety of green chemical analyses [12]. However, to date, only one article reported the use of natural extracts for hands-on, at-home analytical chemistry laboratories. Caraballo et al.

proposed using natural extracts, i.e., curcumin and red cabbage extracts, as indicators for an acid-base equilibrium experiment, in which students would learn the main concepts of acid—base equilibrium, buffer capacity, and titration curves, as well as titrimetric analysis [7]. However, the natural extracts they used were exploited as indicators and were indirectly involved in the main acid-base reactions.

Our previous work suggested that using the natural extracts from guava leaves as a reagent for colorimetric determination of iron worked successfully [13]. The phenolic compounds in guava leaf extract played a spot role as the reagent binding to or chelating with iron ions (Fe(III)), resulting in a dark purple complex that could be detected colorimetrically. The use of guava leaf simple extract as the natural reagent for iron assay was developed further as a natural-reagent iron assay kit that was implemented in on-site analytical chemistry laboratories at six Thai universities [14]. The dried and ground guava leaf powder was packed in a tea bag and sealed in an airtight opaque package that was convenient to use. Other nonhazardous reagents, i.e., Fe(III) stock standard solution and acetate/acetic acid buffer solution, both with mild concentrations, as well as the necessary equipment including a vial for extracting the natural reagent and a 96-well microplate as a reaction platform, were provided in the kit. Detecting the color complex, the product of the reaction could be performed using a smart device (a smartphone or a tablet) with a built-in camera. Since the intensity of the color of the product solution indicated the concentration of the Fe(III) ion, compared with a standard calibration graph, the photo of the reacted sample/standard solution was simply processed to obtain the color intensity using an image processing program on a computer or an application on a smart device. Therefore, the use of the natural-reagent iron assay kit is not only promising for hands-on, at-home laboratories but also for promoting sustainability in analytical chemistry education, supporting the Sustainable Development Goals (SDGs) identified by the United Nations (UN) in 2015, namely good health (SDG3), quality education (SDG4), reduced inequalities (SDG10), and responsible consumption and production (SDG12).

Herein, we present the development of a Lab-at-Home (LAH) box set, as the second article in the series to develop the natural-reagent iron assay kit based on Thai local wisdom, for new normal, analytical chemistry experimentation. This work aimed to design and implement the LAH for the remote learning of an analytical chemistry laboratory, i.e., colorimetry experiment, for foundation chemistry students at the Department of Chemistry, Faculty of Science, King Mongkut's University of Technology Thonburi (KMUTT), Bangkok, Thailand. Students (group/individual) received the LAH box set about one week before the synchronous experiment class via Zoom, where the instructors and teaching assistants (TAs) were prompt to advise and answer questions. The student learning outcome achievements including the comprehension of analytical characteristics and colorimetry principles, as well as the ability to analyze statistical data, were evaluated using a post-lab quiz and laboratory report. The satisfaction of the LAH used for new normal, analytical chemistry experimentation was also assessed using questionnaires and focus group discussions.

2. Materials and Methods

2.1. LAH Box Set

The LAH box set was adapted from the natural-reagent iron assay kit that was implemented as a tool in an analytical chemistry experiment (determination of iron by colorimetric method) for undergraduate science students at six Thai universities, as reported previously [14]. Non-toxic chemicals were provided in the LAH box set, similar to the iron assay kit. The guava leaf extract was used as a natural reagent for complexing with the Fe(III) ion. A series of working Fe(III) standard solutions with mild concentrations were provided. A weak acid in vinegar (acetic acid) was used to prepare the acetic/acetate buffer solution, also provided in the box set. A 96-well plate was exploited as a reaction platform. However, instead of the micropipet used in the previous assay kit for measuring the volumes of the solutions, disposable syringes were employed because they are low cost and more readily available in Thailand. In addition, a series of dye solutions were given to practice the calibration graph construction from the data obtained using a smart device camera, followed by processing with an image analyzing application/program.

Therefore, the LAH box set consisted of eight items (Figure 1), i.e., (i) a sachet of Nat Pow Iron (guava leaf powder), (ii) vials containing Fe(III) standard solutions (with different concentrations) and a sample solution, (iii) a bottle of acetic/acetate buffer solution, (iv) vials containing dye solutions with various concentrations, (v) a glass bottle for extracting the natural reagent, (vi) a 96-well plate, (vii) a 10-mL disposable syringe, and (viii) 1mL disposable syringes. To reduce the inequalities (supporting SDG10) and control the analytical parameters, the LAH box set with equal quantities and qualities of chemicals and materials was sent to all students.



Figure 1. LAH box set: (i) a sachet of Nat Pow Iron, (ii) vials containing Fe(III) standard and sample solutions, (iii) a bottle of acetic/acetate buffer solution, (iv) vials containing dye solutions, (v) a natural reagent extracting bottle, (vi) a 96-well plate, (vii) a 10-mL disposable syringe, and (viii) 1-mL disposable syringes.

2.2. The Design of the LAH Experiment

The LAH was especially designed to suit the foundation analytical chemistry experiment at the undergraduate level in Thailand. In general, the foundation course has a large number of students, for example, 800 students per course, in 20 sections (40 or more students/section) at the Department of Chemistry, Faculty of Science, KMUTT, with a tendency to have more students. A face-to-face on-site laboratory for the large foundation course would consume a large amount of chemicals and budget, require numerous laboratory facilities, and generate a significant amount of chemical waste. The LAH was created to reduce the number of people encountering COVID-19 during the pandemic, and also to address the challenges of large foundation courses using online synchronous experimentation allowing the experiment to be conducted conveniently and safely at a student's residence. All students could conduct the experiment synchronously via a virtual meeting application/program such as Zoom, Microsoft Teams, and Google Meet, and/or a LINE broadcast, where the instructor and TAs were ready to give advice and answer questions. Moreover, other online tools, including Mentimeter, Microsoft Office, and Google Drive, were applied to share ideas and analytical data, facilitating rapid communication, data transmission, and traceability.

For colorimetry experiments, detecting the color intensity is, obviously, important. The LAH experiment used a built-in camera of a smart device (a smartphone/a tablet) as a

detector. A photo of the solutions in the 96-well plate was taken as one shot. The intensity (Red-Green-Blue or RGB value) of the color of each solution was evaluated using an image processing application/program, i.e., Color Grab (Loomatix, Haifa, Israel) for Android or Pixel Picker for iOS.

In addition, the LAH was designed to encourage the learning of colorimetry principles and analytical characteristics as well as to practice statistical data analysis skills. The LAH employed the natural reagent because of its ease of use, safety, and waste management, in which the goals of green chemical analysis and sustainability were achieved.

2.3. Course Characteristics

The LAH experiment was implemented in an analytical chemistry experiment for undergraduate students in two sections at the Department of Chemistry, Faculty of Science, KMUTT, in the second semester of academic year 2020 (2nd/2020), from March to April 2021, and the first semester of academic year 2021 (1st/2021), from August to September 2021. Altogether, 38 and 65 students were enrolled in sections of the 2nd/2020 and 1st/2021 semesters, respectively. During the 2nd/2020 semester, classrooms and laboratories were closed due to the moderate level of COVID-19 pandemic in Thailand, but students were permitted to stay at the university dormitories for online remote learning. Furthermore, no more than five students were allowed to meet face-to-face. Thus, the students in sections during the 2nd/2020 semester were divided in ten groups (three to four students/group), and each group conducted the experiment at the dorms using one LAH box set. During the 1st/2021 semester, the university was closed due to the severity of the pandemic, forcing students to leave the dormitories and stay at home. In this latter case, students performed the experiment individually at home. It should be noted that the basic knowledge in chemistry of these two student groups was comparable. Both groups of students had completed previous chemistry courses (General Chemistry and General Chemistry Laboratory).

2.4. Teaching and Learning Method

The goal of the designed teaching and learning approach in this work was to limit the risk of COVID-19 infection and spread, and to solve the challenges of a large foundation course that could occur in the near future, as previously noted. As a result, the proposed method was designed for 100% remote foundation chemistry education, as well as achieving students' learning outcomes.

The experiment employing the LAH was entitled "Determination of iron by modern green chemical analysis employing a natural reagent and with a smartphone", with three objectives, requiring students to be able to (1) explain colorimetry principles and analytical characteristics, e.g., accuracy and precision, and apply the knowledge to perform quantitative analysis, (2) use a smart device for colorimetric detection and analytical data acquisition for chemical analysis, and (3) briefly describe the importance of the green analytical chemistry (GAC) and the benefits of the novel and modern green chemical analysis.

The experiment was divided in two parts: part 1—Determination of a dye solution concentration and part 2—Colorimetric determination of iron. Part 1 was designed for practicing color detection using a smart device, so that students would be familiar with the use of a smart device as an analytical detector as well as the concept of colorimetric detection, before performing the iron analysis. In part 2, students would learn how to perform colorimetric determination and the green analytical chemistry concept. Moreover, students would learn how to handle analytical data and assess analytical characteristics from both parts.

Figure 2 illustrates the stepwise method of the experiment. An online lab briefing session via Zoom was held two weeks before the experiment date to discuss the methods of the online class using a synchronous model, guidelines of the experimental data and report submission, to introduce the concept of the experiment and to discuss the GAC concept and principles related to this experiment. Then, one week before the day of the experiment, LAH box sets were sent to students at their home addresses and the group representative
at his/her dormitory, in cases of individual and group assignments, respectively. Students were assigned to self-study the concept and experiment instructions and write up the experiment plan. Students were required to submit the experiment plan to the instructor via Google Drive at least five days before the experiment date, for the instructor's approval.

On the experiment date, the students performed the experiment synchronously at their respective locations using Zoom, while instructors and TAs were available online to advise and answer questions as needed by the students. The students were required to activate the video mode on the device connected to Zoom to show their activities during the LAH experiment. The students engaged in a hands-on experiment at home. After the experiment, the analytical data were immediately submitted online via Google Doc/spreadsheet, whereas the experiment report submission due date was one week after the experiment. Finally, another synchronous session was conducted via Zoom to discuss the analytical results and the comprehension of colorimetry principles, analytical characteristics, and GAC principles, to take the post-lab quiz, and to evaluate the use of the LAH.



Figure 2. Stepwise method of the LAH experiment.

2.5. Experimental Procedures

The instruction of the experiment (see Supplementary Material: File S1) was included in the LAH box set, so that students could self-study the procedure before the experiment date. The experiment was designed for two parts, with specific objectives as detailed in Section 2.4.

Part 1-Determination of a dye solution concentration

Firstly, the given dye solutions with various known concentrations and a blind sample solution were dropped in designated positions of the 96-well plate (Figure 3). Secondly, a photograph of the plate was taken using the lighting technique that students preferred or found convenient, and the color intensities of the solutions were evaluated using an application on a smart device. Finally, a calibration graph for each row (B, C, D, E, F, and G, Figure 3) of the 96-well plate was constructed by plotting the color intensities against the concentrations of the dye standard solutions. Supplementary Material: Video S1 shows a demonstration of determining the concentration of a dye solution. The colorimetric principle according to the RGB channel was applied. A proper color channel for the

measurement was determined based on the slope and the correlation coefficient (R^2) of the calibration graph. The influence of the lighting conditions was discussed.

Part 2-Colorimetric determination of iron

The natural reagent containing polyphenolic compounds was, firstly, extracted from the provided guava leaf powder in a sachet (Nat Pow Iron (G)) by adding 60 mL hot (60–100 °C) clean water in the extraction bottle, already containing the sachet. Secondly, 9 mL of the extract was brought to the buffer bottle containing the buffer solution, and the natural reagent was obtained. Thirdly, the standard Fe(III) solutions, with a series of concentrations and some known concentration samples, were dropped in the 96-well plate at the designated wells (Figure 3). The natural reagent was, fourthly, dropped into the standard and sample solutions. The dark-purple polyphenolic-Fe(III) complex was obtained with different color intensities, proportional to the amount of Fe(III). Finally, for each row, the color intensity against the concentration of Fe(III) was plotted as a calibration graph. The Fe(III) concentrations in the samples were evaluated using the calibration graph. Supplementary Material: Video S2 shows a demonstration of colorimetric determination of iron. Some parameters affecting the analytical results were discussed.



Standard Dye/Fe(III) Ounknown sample

Figure 3. Designated positions for dropping solutions in the 96-well plate, where S is standard, U is unknown, B–G are the rows, the numbers after S and U are the number of standard and unknown solutions, respectively, and the numbers after the rows B–G are the column numbers.

2.6. Evaluation and Statistical Analysis of Student Outcomes

Post-lab quizzes as well as lab report scores were used to evaluate the student learning outcome achievements including comprehending the analytical characteristics and colorimetry principles, and the ability to perform statistical data analysis. The LAH experiment used for remote learning was evaluated using a satisfaction survey with a 5-point Likert scale (1 = least satisfied, 2 = less satisfied, 3 = moderately satisfied, 4 = very satisfied, 5 = most satisfied) and focus group discussion. The student learning outcome achievements and satisfaction with the LAH experiment evaluated by students assigned to perform the experiment as a group (see Supplementary Material: Video S3) were compared with those assigned individually (see Supplementary Material: Video S4) to perform the experiment, using independent *t*-test, with a p < 0.05 as significant difference. Bonferroni correction was used to adjust the p-value because of multiple comparisons (10 correlations). The significance level or the *p*-value (0.05) was then adjusted to 0.05/10 = 0.005.

3. Results and Discussion

The LAH was designed for large foundation chemistry courses with a large number of students, such as the General Chemistry course at KMUTT, which has approximately 800 students/semester. In this work, the LAH was introduced to sessions for feasibility planning. Students performed the experimentation individually or in groups at their location using the LAH under the guidance of the instructor and/or TAs. This led to quality education (SDG4) and reduced inequalities (SDG10). The students were able to practice analytical chemistry skills using a hands-on experiment during the COVID-19 pandemic, and they were able to experience green chemistry practice.

3.1. Design of LAH Box Set

The LAH design changed some equipment of the natural-reagent iron assay kit, as reported previously [14]. The reagents/materials provided in the LAH box set were safe for users and the environment, as well as readily available locally, not only for economic reasons, but also for the convenience of the box set preparation. The equipment in the LAH box was lower cost and more readily available in stores/pharmacies, such as disposable syringes to measure the volume of solution, instead of using an autopipet, and a glass bottle to extract the natural reagent that could be reused. The reagents used to determine Fe(III) were the same as those used in the previous study. The dye solutions with a series of concentrations, including blind concentration, were included in the newly designed LAH to determine dye concentration. Students could thus practice using their smart devices to take photographs of the solutions in the well plate and evaluate the color intensity of solutions. The LAH box cost only 200 THB (6 USD) per set (110 THB (3.3 USD) for reusable equipment and 90 THB (2.7 USD) for chemicals and reagents). In addition, the LAH was the first green chemical analysis tool, especially for chemistry laboratory, in Thailand that could be easily and safely performed at home because the reagent was extracted from guava leaves, nonhazardous chemicals were down-scaled, the waste was easily eliminated and was deemed environmentally friendly.

3.2. Learning Process

The learning process planned for the LAH experiment began with an online laboratory briefing session using Zoom, followed by a hands-on experiment using the LAH box set, and finally, an online discussion session. Students received the LAH box set with the laboratory instruction so that they studied the concept and procedure before the online briefing session. On the experiment date, they streamed themselves using Zoom while performing the experiment alongside their friends, while the instructor and TAs observed and were available for advising and answering questions (see Supplementary Materials: Videos S3 and S4). In this way, students felt relieved that they could ask the instructor or TAs for help whenever they needed it, as one student commented: "When we had a serious issue, we could turn on the microphone and ask the instructor. I didn't feel nervous or anxious at all." Since Thai students are typically hesitant to criticize and express their opinions, prior-normal experimentation included a discussion session to encourage them. When transforming to new-normal experimentation, the synchronous session could not be neglected. The results showed that in the online discussion session, students shared and exchanged their opinions with friends and the instructor. Critical thinking skills could be improved in this session, as one student stated: "Sharing mistakes and exchanging ideas with friends helped us understand more and gain a broader perspective." The proposed learning process could be applied in the future to large foundation chemistry courses, which may not have sufficient laboratory space for 800 students. Although the proposed process used synchronous sessions to support the development of Thai students' learning skills, asynchronous virtual laboratories could be used in some cases, such as when students were mature and/or had sufficient learning skills.

3.3. LAH Experimentation

The analytical chemistry experimentation using the LAH as a tool was divided in two parts. Part 1 determined the dye solution concentration using modern green chemical analysis and a smart device. The concept of colorimetric determination, as well as analytical data analysis, was introduced. Students practiced taking photos of the solutions in the 96-well plate using the camera of a smart device and evaluating the RGB color intensities of the solutions by an image processing application. From the focus group discussion, students were impressed using a smart device with an image processing application for analytical data processing and judged it promising for future chemical analysis due to its simplicity. The effects of the color channel on the color intensity evaluation and the lighting condition were studied. Finally, students plotted the calibration graph and determined the analytical characteristics, including accuracy and precision.

3.3.1. RGB Color Intensities

TechTerms.com (accessed on 10 January 2022) defined that "the RGB refers to three hues of light that can be mixed together to create different colors [15]. Combining red, green, and blue light is the standard method of producing color images on screens, such as TVs, computer monitors, and smartphone screens." Levels of R, G, and B can each range from 0% to 100% of full intensity. Each level is represented by the range of decimal numbers from 0 to 255 (256 levels for each color) [16]. The RGB level of the photograph of the solution was determined by Color Grab for Android and Pixel Picker for iOS systems. Figure 4a,b illustrate the examples of the images taken under two different lights. One was taken in a student's room with a fluorescent bulb (Figure 4a), while the other was taken in a room with a white-light LED lamp (Figure 4b). Students read the R, G, and B levels of the solutions in the assigned wells and constructed the calibration graphs, plotting the intensity of each color against the concentration of dye solutions. Figure 4c,d illustrate the examples of the calibration graphs using the R, G, and B levels of standard dye solutions at positions B3 to B7.

According to the sensitivity (the slope of the graph), the linearity (the correlation coefficient, R²), and the accuracy and precision of the determination, students chose the suitable light color (R, G, or B) for further study. For instance, from Figure 4c under the fluorescent lamp, the sensitivity obtained by plotting the green light intensity (y = -4.3x + 180.6) was higher than that obtained by plotting the blue (y = -2.2x + 172.0) and red (y = -0.1x + 186.6) light intensity, respectively.

Due to the wide use of standard RGB (sRGB) in, for example, digital photography, graphics, multimedia, and Internet imaging, the image is probably encoded in sRGB colors. The JPEG image that the students took using their smart device camera was opened with ImageJ or Java. The pixel values in the resulting data array were R, G, and B components [17]. As mentioned earlier, R, G, and B have integer values from 0 to 255. The different shades of pink have specific integer values of R, G, and B. For example, the RGB integer values for pink, light pink, and hot pink are RGB(255,192,203), RBG(255,182,193), and RGB(255,105,180), respectively [18]. Notably, the integer value of R is 255 for all shades of pink. This could explain the lower slope of the calibration graph obtained by plotting the red intensity. The highest linearity was found when the green light intensity was applied. Interestingly, the highest slope of the calibration graph obtained by plotting the green intensity could have resulted from the largest difference of the G values. Similar findings were obtained under the white-light LED lamp (Figure 4d). In this case, the calibration graph using the green light intensity was selected.



Figure 4. Examples of photographs from experiment part 1 taken (**a**) in a student's room having a fluorescent lamp and (**b**) under a white-light LED lamp, and the calibration graphs plotted the RGB levels of the solutions at positions B3–B7 in the black squares of (\mathbf{a} ,(\mathbf{b}) obtained under (\mathbf{c}) a fluorescent lamp and (\mathbf{d}) a white-light LED lamp against the dye concentration, respectively.

3.3.2. The Effect of the Light Condition

Each student or group used a different light condition depending on their preference or convenience. Some examples of light sources were white-light LED lamp, fluorescent lamp, and the sun (resulting in natural light). To discuss the effect of the light condition, students shared their results during the online discussion session, and compared the accuracy and precision of the determination using the calibration graph of the green light intensity from each row of well plates and dye concentrations. Altogether, 10 groups and 65 individual students, each with six calibration graphs from six rows in the well plate, yielded 450 calibration graphs. However, 130 calibration graphs were not linear due to the effect of the light. Consequently, the concentration range of dye in the blind sample was 1 to 8 ppm (true value = 5 ppm), with the average and standard deviation (SD) of 5 and 2 ppm (n = 320), respectively. The overall precision of the determination was low (%RSD = 36). The comparison indicated that the colorimetric determination using a smart device as a detector had been greatly influenced by the light condition. In theory, the sRGB was developed with the goal of creating a precisely specified color space for computer-based and display-oriented applications. This means that precise specifications of factors, such as the white reference point and the ambient lighting conditions, are important [3]. From the results, students would be aware that light condition affects the precision and accuracy of the color intensity when performing the colorimetric determination using a smart device as a detector in the future.

Aside from the light condition effect, the small size of the wells on the reaction plate was another cause of non-linear calibration graphs. When dropping solutions, a student said they had trouble finding the assigned wells on the plate, resulting in the solutions being

dropped in the wrong order. Students had to focus intensely on the experiment. However, working from home could involve interruptions by family members and household pets; one student mentioned that their grandmother was so concerned for their safety that she came to see them frequently. These constraints should be kept in mind for future work.

3.3.3. Colorimetric Determination of Iron

In part 2 of the experiment, the dark purple polyphenolic-Fe(III) complex was obtained as the product of the chemical reaction. Its intensity depends on the concentration of Fe(III) ion. Higher Fe(III) concentrations produce higher color intensities. Figure 5 illustrates the examples of images of the polyphenolic-Fe(III) complex solutions in the well plate under different lights and calibration graphs plotting the green color intensity against the Fe(III) concentration. The concentration of Fe(III) in the blind sample in each row was determined using a calibration graph of the same row. For example, the Fe(III) concentration of the sample at positions B8 to B10 was determined using the calibration graph plotting the color intensities of the standards at the position B3–B7. Unsurprisingly, the Fe(III) concentration was strongly affected by light condition during the photographing. Altogether, 127 nonlinear calibration graphs resulted in 323 linear calibration graphs determine Fe(III) concentration. The range of Fe(III) concentration determined by the students was 1 to 8 ppm (n = 323), with an average, SD, and %RSD of 5 ppm, 2 ppm, and 38%, respectively. The students again noticed the influence of light condition and RGB colors on colorimetric determination. They learned the principles of colorimetry as well as analytical characteristics. Students' comprehension was assessed by quiz scores and lab reports.

3.4. Learning Outcome Achievement

At the end of the experiment, students submitted the raw analytical data, calibration graphs, and concentrations of the dye and Fe(III) in blind samples using Google Drive. The students then worked in groups of three to four students to write the lab report. The report must be submitted using Google Drive within one week after the experiment. Students took the post-lab quiz in the online discussion session using Zoom (Zoom Video Communications, Inc., San Jose, USA). To evaluate the learning outcome achievements, the post-lab quiz assessed for the student's comprehension of colorimetry principles, analytical characteristics, statistical analysis of the analytical data, and the green analytical chemistry concept.

3.4.1. Post-Lab Quiz Scores

The average post-lab quiz scores of students who conducted the LAH experiment individually and in groups were 7.00 \pm 1.54 (n = 65) and 6.68 \pm 1.12 (n = 38), respectively, out of 10 points. The difference between the scores of students who performed the LAH experiment individually and in groups was without statistical significance (p = 0.951). When these scores were compared with those of students performing the colorimetry experiment at the university in the first semester of 2019 (August to November 2019), before the COVID-19 pandemic (6.18 ± 1.68 points, n = 79), the average score of students performing the LAH experiment in groups did not significantly differ (p = 0.283). However, the average score of students performing the LAH individually was significantly higher (p = 0.005). Notably, before the COVID-19 pandemic, students performed the colorimetry experiment in groups. This suggested that individual rather than group work could be used to improve student comprehension. It contrasted with the findings reported by Rawas et al. (2020), that the group-based activities resulted in better test scores compared with the design comprising individual activities for face-to-face flipped classroom [19]. However, results agreed with studies by Achuthan et al. (2021), reporting that providing and enabling a learning environment promoting effective individualized learning and comprehension of experimental concepts and skills could improve educational outcomes in remote laboratories [20].



Figure 5. Examples of the photographs from part 2 of the experiment taken under (**a**) fluorescent lamp and (**b**) white-light LED lamp, establishing calibration graphs in (**c**), under (**a**), and for rows B and G as well as calibration graphs in (**d**) under (**b**) for rows B and G.

3.4.2. Lab Report Scores

To write a lab report, students worked in groups of three to four, discussing the experiment method, evaluating the analytical results, and drawing conclusions. Thus, the number of reports composed by students who performed the experiment individually (65 students) and in groups (38 students) was 18 and 10, respectively. The lab report was used to assess (1) the understanding of the influences of the RGB color and lighting conditions as a result of using a smart device as a colorimetry detector, (2) the ability in applying colorimetry principles to determine dye and Fe(III) concentrations, and (3) the ability to apply the statistical data analysis principle to explain the analytical results. Overall report scores of the students using the LAH individually (23.33 \pm 2.39 points, n = 18) and the students who performed the colorimetry experiment at the university before the COVID-19 pandemic (23.53 \pm 2.24 points, n = 20) were significantly higher than those of the students who conducted the LAH experiment in groups (20.00 \pm 2.22 points, n = 10) with *p* = 0.01 and < 0.001, respectively. It showed that individualized laboratory practice was just as effective as face-to-face laboratory practice. This was consistent with the studies of Rawas et al. [19] and Achuthan et al. [20], above.

The scores of students performing the LAH experiment individually were compared with those in groups for each learning outcome achievement (Table 1). Students who worked individually and in groups received similar scores, without any statistically significant difference, for their understanding of the influences of the RGB color and lighting conditions, as well as the ability in applying colorimetry principles to determine dye and Fe(III) concentrations. Unexpectedly, students who performed the experiment individually had statistically higher scores for their ability in applying the statistical data analysis principle to explain the analytical results than those who performed the experiment in groups although their basic knowledge of chemistry was likely comparable. To statistically analyze the chemical analysis data, students in groups had to collaborate with peers having different learning styles. The university's semi-closure due to the COVID-19 pandemic during the second semester of academic year 2020 (March-April 2021) resulted in fewer student meetings and fewer face-to-face activities and, therefore, a lack of social and collaborative skill development for students. Moreover, in a report on the importance of group work in mathematics, Koçak et al. (2009) discovered that the efficiency of group work increases with the intensity of the communication that is established among the group members, which is free of teacher-centered studies. The group work that is carried out through a planned and programmed activity and healthy communication among group members are the most important factors. When students are not directed toward a goal, and without planning in place, group work will be nonbeneficial [21].

Table 1. Comparison of the scores of students who performed the LAH experiment individually and in groups assessed by learning outcome achievements.

Learning Outcome Achievement	Score (Mean \pm 5 Who Performed the	p Value	
5	Individually (n = 18)	In Groups (n = 10)	
Understanding of the influences of the RGB color and the lighting conditions (4 points)	3.56 ± 0.51	3.50 ± 0.85	0.830
Ability to apply colorimetry principles to determine dye and Fe(III) concentrations (4 points)	2.33 ± 0.86	1.80 ± 0.68	0.103
Ability to apply the statistical data analysis principle to explain the analytical results (10 points)	8.11 ± 1.88	6.7 ± 1.16	0.041

3.5. Evaluation of Using the LAH for the Analytical Chemistry Experiment

The use of the LAH for the analytical chemistry experiment, i.e., colorimetric determination of Fe(III), was evaluated using a satisfaction survey. Students using the LAH both individually and in groups for the experimentation were asked to rate the satisfaction level of ten issues, using a Likert scale from 1 to 5 (1 = least satisfied, 2 = less satisfied, 3 = moderately satisfied, 4 = very satisfied, 5 = most satisfied). In general, both groups of students were very satisfied with the use of the LAH in all aspects, as shown in Figure 6, with satisfaction scores greater than 3.80. The satisfaction levels rated by students who experimented individually were slightly higher than those rated by students who experimented in groups. When a statistical independent t-test with Bonferroni correction was applied, it was found that the satisfaction scores in five aspects, namely items 2, 3, 4, 6, and 7, rated by students experimenting individually were significantly higher than those by students experimenting in groups (p < 0.005), details as shown in Table 2. For the other aspects, namely items 1, 5, 8, 9, and 10, there were trends showing the higher scores among individual experimenting than experimenting in groups were without statistical significance (p > 0.005). A student performed the experiment individually expressed her impression that "I enjoy that I can practice working alone because in the future, when I finish my undergraduate degree and go to work, I won't have anyone to assist me with my work".

The overall satisfaction levels rated by the individual and group experiments were 4.35 ± 0.80 and 3.95 ± 0.87 , respectively, meaning very satisfied. Many expressed, for example, "I've never had any experience like this before. This LAH could be the first of its kind in the world." and, "Because we see what happens, the hands-on experiment allows us to gain a deeper understanding of concepts. Unlike watching a demonstrated experiment, real-world experience leads to long-term memories." indicating students' impressions of the LAH.

- Group Experiment
- Individual Experiment



Figure 6. Satisfaction in the use of the LAH for the hands-on experiment at home.

Comparing the issues, students were most satisfied with the harmlessness of chemicals in the LAH, exhibiting satisfaction levels of 4.80 ± 0.40 and 4.45 ± 0.50 rated by students working individually and in groups, respectively. A student commented during the focus group discussion, "I was worried about the hazards of chemicals because the elderly and children were residing in the house, and my mother had a congenital disease. After reading the instructions and hearing from the instructor during the online lab briefing session, I discovered that the reagents were safe because one was guava leaf extract and

the two others were a kitchen ingredient (vinegar) and a mineral (iron) at much diluted concentrations. So, I was relieved. My mother was also concerned when the LAH box set arrived, wondering if it would cause any harm. I explained that the material was made from guava leaves and that the chemicals used were extremely low in concentration, even lower than the vinegar we used at home. My mother understood and agreed to let me conduct the experiment at home." Students were impressed not only with the use of non-hazardous chemicals, such as acetic acid, which is less concentrated than the vinegar used in the kitchen, and a natural reagent extracted from guava leaves, but also with the ease of the non-hazardous chemical waste disposal. Moreover, the equipment and containers were reusable. This encouraged students to apply the green analytical chemistry concept in their future work and also supported SDG12 (responsible consumption and production), which is concerned with responsible use of natural resources to avoid harmful effects to the environment.

Table 2. Satisfaction of LAH use rated by students performing the experiment individually and in groups.

Issue	Satisfaction Level (Mean \pm SD) Rated by Students Performing the Experiment		<i>p</i> Value
_	Individually (n = 65)	In Groups (n = 38)	
1. Clearness and comprehensibility of the procedure manual	4.58 ± 0.56	4.24 ± 0.68	0.006
2. Harmlessness of chemicals in the LAH, compared with those in other laboratories	4.80 ± 0.40	4.45 ± 0.50	< 0.001
3. Availability and simplicity of the equipment	4.65 ± 0.57	4.08 ± 0.75	< 0.001
4. Advantage of using a smart device with an image processing application for analytical data processing	4.66 ± 0.51	4.16 ± 0.72	0.001
5. The interestingness of online analytical data and report submission	4.23 ± 0.70	3.95 ± 0.80	0.064
6. Knowledge and experience in new modern chemical analysis gained	4.72 ± 0.52	4.13 ± 0.78	<0.001
7. Knowledge in the principle of green chemistry	4.72 ± 0.48	4.18 ± 0.61	< 0.001
8. Comprehensibility in the principle of the analytical characteristics according to hands-on experiment	4.35 ± 0.62	4.13 ± 0.70	0.099
9. Ease of selecting a location for the experiment	4.29 ± 0.74	3.82 ± 0.90	0.005
10. Overall satisfaction of the LAH	4.35 ± 0.80	3.95 ± 0.87	0.018

On the other hand, the least satisfactory issue for the students performing the LAH experiment individually was the interests in online analytical data and report submission (4.23 \pm 0.70), whereas for the students performing the LAH experiment in groups, the least satisfactory issue was the ease of selecting a location for the experiment (3.82 \pm 0.90). For students working individually, the low satisfaction score of those interested in online submissions could be due to issues with internet connection at their location. Some students reported that their Zoom connection was frequently lost due to a poor Internet signal. Students performing group work had to find one location in the university with sufficient space for experimentation and where the group members could meet easily. As a result, they may have experienced difficulty doing so. However, using LAH for analytical chemistry practice, either individually or in groups, could help develop the expected learning outcomes of the experiment.

3.6. Sustainability

This work has developed a hands-on analytical chemistry experiment that students in higher education could easily and safely conduct at their home on their own. Using locally

available equipment/materials in the LAH box set has somehow ensured equal access for all foundation chemistry students to affordable, supporting inclusive and equitable education. The results also showed that students could develop an understanding of the principle of colorimetry, as well as analytical chemistry skills. It indicated that the LAH ensured quality education. The high scores of the laboratory reports showed that students' lifelong learning skills, such as problem solving, critical thinking, communication, and collaboration, were improved. Thus, the LAH supports sustainable development in quality education (SDG4) to "ensure inclusive and equitable quality education and promote lifelong learning opportunities for all" and, particularly, Target 4.3 that states "by 2030, ensure equal access for all women and men to affordable and quality technical, vocational and tertiary education, including university."

Analytical chemistry laboratories typically require accurate and precise equipment/ instruments, which are relatively expensive. Some Thai universities may be unable to provide sufficient and efficient equipment/instruments, limiting students' opportunities to gain real-world experience with them and practice good chemistry skills. The inexpensive LAH box set means that Thai universities could probably afford to provide each student with one set of the LAH for self-learning anywhere, including in a foundation course having a large number of students, which would help to reduce educational inequalities in Thailand and elsewhere. As a result, Thai universities could be one of the country's mechanisms for ensuring equal opportunity and reducing outcomes inequalities (Target 10.3 of SDG10—Ensure equal opportunity and reduce inequalities of outcome, including by eliminating discriminatory laws, policies, and practices and promoting appropriate legislation, policies, and action in this regard).

Nevertheless, the impacts of the LAH use on human health and the environment were the primary factors that we considered as we designed the experimentation. All the chemicals in the box set must be safe for students and their family. The waste must be easily managed and environmentally friendly. The coloring reagent is most important in colorimetry and is required in high amount. Thus, a nonhazardous natural extract of guava leaves was used as the coloring reagent in the determination of iron. The other chemicals provided in the LAH box set were also nonhazardous. The diluted acetate buffer for controlling the pH of solutions could be prepared from kitchen vinegar and its salt. Fe(III), which is one of the naturally occurring elements and required for plants [22], was prepared as the standard solutions at relatively low concentrations. The dye used for part 1 in the experiment was food grade. The LAH use, therefore, encourages students and faculty to aware of the GAC concept and supports SDG12, Target 12.2-by 2030, achieve the sustainable management and efficient use of natural resources, and Target 12.4—by 2020, achieve the environmentally sound management of chemicals and all wastes throughout their life cycle, in accordance with agreed international frameworks, and significantly reduce their release to air, water, and soil in order to minimize their adverse impacts on human health and the environment.

Sustainability, defined by the United Nations Brundtland Commission in 1987 [23] as "meeting the needs of the present without compromising the ability of future generations to meet their own needs", could be developed incrementally by all of us. Since we are chemists and educators, we have developed a tool that would meet the needs of chemistry in the present while also encouraging future generations to meet their own.

4. Conclusions

Not only was the university closed during the continuous COVID-19 pandemic, but also a large number of the students in the foundation chemistry course in Thailand stimulated us to develop LAH experimentation. The LAH box set adapted the natural reagent iron assay kit, reported in our previous work, aiming to increase the local availability and reduce the cost of the materials. Thus, the university could produce sufficient LAH box sets for large numbers of students. It supported quality education (SDG4) and reduced inequalities (SDG10). The LAH reagents and materials were nonhazardous, making it safe to perform the experiment at home and also support responsible consumption and production (SDG12). The method of the LAH was so simple that students could perform the experiment by themselves, in this case under the supervision of the instructor/TAs. However, a poor Internet connection might constitute a constraint of the synchronous online sessions, in which case asynchronous sessions could be used instead. When using a smart device as a colorimetric detector, students should be aware of the effects of the color channels for color intensity evaluation and light conditions for photographing. With the proposed LAH hands-on experimentation, the students' learning outcomes for the colorimetric determination of Fe(III) experiment, i.e., students' comprehension of colorimetry principles, chemistry principles involving determining Fe(III) concentration, analytical characteristics, statistical analysis of the analytical data, and the green analytical chemistry concept, could be successfully achieved. Although students experimenting individually rated higher satisfaction scores than those experimenting in groups, both were very satisfied with using the LAH in all ten aspects.

Supplementary Materials: The following supporting information can be downloaded at: https: //www.mdpi.com/article/10.3390/su14063314/s1, File S1: instruction of the LAH experiment (in Thai); Video S1: demonstration of the determination of dye concentration using the LAH box set; Video S2: demonstration of the determination of iron concentration using the LAH box set; Video S3: a group of students performing the hands-on LAH experiment; Video S4: students individually performing the LAH experiment. References [12,13,24–30] are cited in the supplementary materials.

Author Contributions: Conceptualization, K.G., M.V., C.Y., P.I.N.A., K.K. (Kullapon Kesonkan) and K.K. (Kanokwan Kiwfo); methodology, M.V., C.Y., P.I.N.A., K.K. (Kullapon Kesonkan) and K.G.; investigation, M.V., C.Y., P.I.N.A. and K.K. (Kullapon Kesonkan); data curation and analysis, M.V., C.Y., P.I.N.A., K.K. (Kullapon Kesonkan), C.S., S.S., D.C., S.C. and P.P. (Piyatida Panitsupakamol); writing—original draft preparation, S.S., M.V., C.Y., P.I.N.A., K.K. (Kullapon Kesonkan), K.G., K.K. (Kanokwan Kiwfo) and P.P. (Pathinan Paengnakorn); writing—review and editing, S.S., M.V., C.Y., P.I.N.A., K.K. (Kullapon Kesonkan), K.G., C.S., D.C. and S.C.; funding acquisition, C.S. and K.G. All authors have read and agreed to the published version of the manuscript.

Funding: This research was supported by the Center of Excellence for Innovation in Analytical Science and Technology for Biodiversity-based Economic and Society (I-ANALY-S-T_B.BES-CMU), Chiang Mai University, under the Distinguished Research Professor Award in "Green Innovation in Chemical Analysis with Local Wisdom" (Grant No. DPG6080002; K. Grudpan), Thailand Research Fund (TRF), Thailand.

Institutional Review Board Statement: The study was conducted in accordance with the guidelines of the Declaration of Helsinki and exempted by the Human Ethics Committee of the Faculty of Pharmacy, Chiang Mai University (Ethics Approval Number: 005/2021/Exe), issued on 15 September 2021.

Informed Consent Statement: Not applicable.

Data Availability Statement: The original contributions generated for this study are included in the article; the data presented in this study are available on request from the corresponding author.

Acknowledgments: The authors would like to express their gratitude to the Center of Excellence for Innovation in Analytical Science and Technology for Biodiversity-based Economic and Society (I-ANALY-S-T_B.BES-CMU), Chiang Mai University, Thailand. We gratefully acknowledge the support from the Alexander von Humboldt Foundation. We are thankful to Tanyuta Prajongsangsri for her assistance with the focus group discussion. We are grateful to the Department of Chemistry, Faculty of Science, King Mongkut's University of Technology Thonburi, and Phannee Rattanachaisit, their support and assistance.

Conflicts of Interest: The authors declare no conflict of interest.

References

- Kelley, E.W. LAB Theory, HLAB Pedagogy, and Review of Laboratory Learning in Chemistry during the COVID-19 Pandemic. J. Chem. Educ. 2021, 98, 2496–2517. [CrossRef]
- Destino, J.F.; Cunningham, K. At-Home Colorimetric and Absorbance-Based Analyses: An Opportunity for Inquiry-Based, Laboratory-Style Learning. J. Chem. Educ. 2020, 97, 2960–2966. [CrossRef]
- Schultz, M.; Callahan, D.L.; Miltiadous, A. Development and Use of Kitchen Chemistry Home Practical Activities during Unanticipated Campus Closures. J. Chem. Educ. 2020, 97, 2678–2684. [CrossRef]
- Andrews, J.L.; de Los Rios, J.P.; Rayaluru, M.; Lee, S.; Mai, L.; Schusser, A.; Mak, C.H. Experimenting with At-Home General Chemistry Laboratories During the COVID-19 Pandemic. J. Chem. Educ. 2020, 97, 1887–1894. [CrossRef]
- Maqsood, S.; Kilpatrick, S.M.; Truong, C.D.; Lefler, S.R. Analysis of Amylase in the Kitchen: An At-Home Biochemistry Experiment for the COVID-19 Pandemic. J. Chem. Educ. 2021, 98, 858–865. [CrossRef]
- Doughan, S.; Shahmuradyan, A. At-Home Real-Life Sample Preparation and Colorimetric-Based Analysis: A Practical Experience Outside the Laboratory. J. Chem. Educ. 2021, 98, 1031–1036. [CrossRef]
- Caraballo, R.M.; Saleh Medina, L.M.; Gomez, S.G.J.; Vensaus, P.; Hamer, M. Turmeric and RGB Analysis: A Low-Cost Experiment for Teaching Acid–Base Equilibria at Home. J. Chem. Educ. 2021, 98, 958–965. [CrossRef]
- Pitre, D.; Stokes, S.; Mlsna, D. Development of an At-Home Chemistry Lab Experience for Survey of Organic Chemistry Students. J. Chem. Educ. 2021, 98, 2403–2410. [CrossRef]
- Ibarra-Rivera, T.R.; Delgado-Montemayor, C.; Oviedo-Garza, F.; Perez-Meseguer, J.; Rivas-Galindo, V.M.; Waksman-Minsky, N.; Alejandro Perez-Lopez, L. Setting Up an Educational Column Chromatography Experiment from Home. J. Chem. Educ. 2020, 97, 3055–3059. [CrossRef]
- Larriba, M.; Rodríguez-Llorente, D.; Cañada-Barcala, A.; Sanz-Santos, E.; Gutiérrez-Sánchez, P.; Pascual-Muñoz, G.; Álvarez-Torrellas, S.; Águeda, V.I.; Delgado, J.A.; García, J. Lab at Home: 3D Printed and Low-Cost Experiments for Thermal Engineering and Separation Processes in COVID-19 Time. *Educ. Chem. Eng.* 2021, 36, 24–37. [CrossRef]
- 11. Christian, G.D.; Dasgupta, P.K.; Schug, K.A. Analytical Chemistry, 7th ed.; John Wiley and Sons, Inc.: Hoboken, NJ, USA, 2014; ISBN 978-0-470-88757-8.
- 12. Grudpan, K.; Hartwell, S.K.; Lapanantnoppakhun, S.; McKelvie, I. The Case for the Use of Unrefined Natural Reagents in Analytical Chemistry—A Green Chemical Perspective. *Anal. Methods* 2010, 2, 1651–1661. [CrossRef]
- Settheeworrarit, T.; Hartwell, S.K.; Lapanatnoppakhun, S.; Jakmunee, J.; Christian, G.D.; Grudpan, K. Exploiting Guava Leaf Extract as an Alternative Natural Reagent for Flow Injection Determination of Iron. *Talanta* 2005, 68, 262–267. [CrossRef] [PubMed]
- Kiwfo, K.; Yeerum, C.; Issarangkura Na Ayutthaya, P.; Kesonkan, K.; Suteerapataranon, S.; Panitsupakamol, P.; Chinwong, D.; Paengnakorn, P.; Chinwong, S.; Kotchabhakdi, N.; et al. Sustainable Education with Local-Wisdom Based Natural Reagent for Green Chemical Analysis with a Smart Device: Experiences in Thailand. *Sustainability* 2021, 13, 11147. [CrossRef]
- 15. RGB (Red Green Blue) Definition. Available online: https://techterms.com/definition/rgb (accessed on 10 January 2022).
- What Is RGB (Red, Green, and Blue)?—Definition from WhatIs.Com. Available online: https://whatis.techtarget.com/definition/ RGB-red-green-and-blue (accessed on 10 January 2022).
- Burger, W.; Burge, M.J. Principles of Digital Image Processing. In Undergraduate Topics in Computer Science; Springer: London, UK, 2009; ISBN 978-1-84800-194-7.
- Pink Color Code-RGB Pink Color. Available online: https://www.rapidtables.com/web/color/pink-color.html (accessed on 6 January 2022).
- Rawas, H.; Bano, N.; Alaidarous, S. Comparing the Effects of Individual Versus Group Face-to-Face Class Activities in Flipped Classroom on Student's Test Performances. *Health Prof. Educ.* 2020, *6*, 153–161. [CrossRef]
- Achuthan, K.; Raghavan, D.; Shankar, B.; Francis, S.P.; Kolil, V.K. Impact of Remote Experimentation, Interactivity and Platform Effectiveness on Laboratory Learning Outcomes. Int. J. Educ. Technol. High. Educ. 2021, 18, 38. [CrossRef]
- Koçak, Z.F.; Bozan, R.; Işık, Ö. The Importance of Group Work in Mathematics. Procedia-Soc. Behav. Sci. 2009, 1, 2363–2365. [CrossRef]
- Schmidt, W.; Thomine, S.; Buckhout, T.J. Editorial: Iron Nutrition and Interactions in Plants. Front. Plant Sci. 2020, 10, 1. [CrossRef] [PubMed]
- 23. Nations, U. Sustainability. Available online: https://www.un.org/en/academic-impact/sustainability (accessed on 1 March 2022).
- 24. Armenta, S.; Garrigues, S.; de la Guardia, M. Green Analytical Chemistry. Trends Anal. Chem. 2008, 27, 497–511. [CrossRef]
- Galuszka, A.; Migaszewski, Z.; Namiesnik, J. The 12 Principles of Green Analytical Chemistry and the Significance Mnemonic of Green Analytical Practices. *Trends Anal. Chem.* 2013, 50, 78–84. [CrossRef]
- Lapanantnoppakhun, S.; Tengjaroensakul, U.; Mungkomasawakul, P.; Puangpila, C.; Kittiwachana, S.; Saengtempiam, J.; Hartwell, S.K. Green Analytical Chemistry Experiment: Quantitative Analysis of Iron in Supplement Tablets with Vis spectrophotometry Using Tea Extract as a Chromogenic Agent. J. Chem. Educ. 2020, 97, 207–214. [CrossRef]
- 27. Yeerum, C.; Wongwilai, W.; Grudpan, K.; Vongboot, M. Green Assay of Anionic Surfactant via Ion-association with Methylene Blue Sorbedon Polyurethane Foam Monolithic Rod and Using a SmartphoneH. *Taianta* **2018**, *190*, 85–88. [CrossRef]
- Grudpan, K.; Hartwell, S.K.; Wongwilai, W.; Grudpan, S.; Lapanantnoppakhun, S. Exploiting Green Analytical Procedures for Acidity and Iron Assays Employing Flow Analysis with Simple Natural Reagent Extracts. *Taianta* 2011, 84, 1396–1400. [CrossRef]

- Kiwfo, K.; Wongwilai, W.; Sakai, T.; Teshima, N.; Murakami, H.; Paengnakorn, P.; Phojuang, K.; Kotchabhakdi, N.; Grudpan, K. Simple Natural Material Based Microfluidic Platforms with a Smartphone Detection Employing Natural Reagent for Acidity Assay. J. Flow Inject. Anal. 2020, 37, 9–12.
- Center of Excellence for Innovation in Analytical Science and Technology (I-ANALY-S-T), Green Innovation in Chemical Analysis with Local Wisdom: EP1 from Local to Global. Available online: https://www.youtube.com/watch?v=Kka-OAzOigY&t=3s (accessed on 20 February 2021).



Article



Engaging Environmental Sciences Students in Statistics through an Inclusive Experience in a Spanish University

Nieves Aquino-Llinares and Pilar Moreno-Navarro *

Department of Economics, Quantitative Methods and Economic History, Pablo de Olavide University, 41013 Seville, Spain; naqulli@upo.es

* Correspondence: mpmornav@upo.es

Abstract: In Spain, the issue of people with disability's access to adapted educational material is still unresolved in the university context. Other insufficiently addressed issues comprise actions to include students with intellectual disability in university classrooms, and the awareness-raising and sensitisation of undergraduate students regarding disability. These deficiencies persist despite the known benefits of these types of initiatives for all the agents involved. For this reason, we carried out an inclusive experience at the Pablo de Olavide University, specifically in the statistics subject. Educational resources were adapted, inclusion activities were conducted with students with intellectual disability, and we were in charge of awareness-raising and sensitisation of undergraduate students. The present paper describes the experience as well as its evaluation, which was performed using a survey. Furthermore, the work compares the achievement of students with a more engaging system that incorporates inclusive teaching versus one that does not. The results, which were statistically analysed, report high levels of satisfaction for all the involved agents, as well as improvements in the academic performance of the students. Recommendations directed towards both teaching staff and educational authorities are also provided on how to promote inclusion in universities and more specifically inclusion in science. These suggested educational policies aim to ensure inclusive and equitable quality education and promote lifelong learning opportunities for all, based on the Sustainable Development Goal 4 of the 2030 Agenda for Sustainable Development.

Keywords: accessibility; awareness; disability; higher education; inclusive science; statistics; teaching support

1. Introduction

According to the World Health Organization, about 15% of the world's population (i.e., more than one billion people) has some form of disability. People with disability thus constitute the largest minority in the world and numbers are increasing with advances in medicine and the growth and aging of the population. That is why the fourth goal of the 2030 Agenda for Sustainable Development aims to ensure inclusive and equitable quality education and promote lifelong learning opportunities for all [1]. On the other hand, the latest 2017–2018 academic year CRUE report states that 14,930 students with some type of disability are studying for bachelor's degrees in Spanish public universities, of which 12% (1798) are studying in Andalusian universities [2]. Society cannot look the other way in the face of such an obvious reality. Universities play a fundamental role in the training and incorporation of students with disability, a crucial pillar in the improvement of the welfare state.

On 13 December 2006, the International Convention on the Rights of Persons with Disability was approved in New York. It provides measures, of both non-discrimination and positive action, that states must implement to ensure that disabled people enjoy their rights on equal terms with other people. Spain was one of the first countries to ratify this international treaty [3], whose article 24, subsection 5 holds: "Member States shall ensure

Citation: Aquino-Llinares, N.; Moreno-Navarro, P. Engaging Environmental Sciences Students in Statistics through an Inclusive Experience in a Spanish University. *Sustainability* **2022**, *14*, 8352. https://doi.org/10.3390/ su14148352

Academic Editor: Larry J. Grabau

Received: 15 May 2022 Accepted: 6 July 2022 Published: 7 July 2022

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). that persons with disability have general access to higher education, vocational training, adult education and lifelong learning without discrimination and under equal conditions than the others. To this end, Member States shall ensure that reasonable accommodations are made for persons with disability". That is why Spanish public universities must comply with these precepts. In this sense, it is important to highlight that current Spanish regulations reserve 5% of places in each university degree for people with disability (see article 26 of Royal Decree 412/2014, of 6 June 2014 [4]). The incorporation of a quota for people with disability into the laws that regulate access to university represented a great advance in the fight for the educational and social integration of this group. It was a clear example of positive action policies towards people with disability. Recently, Royal Decree 822/2021, of 28 September 2021 [5] extended the reserve quota in force for undergraduate studies to official university master's degrees. In addition, it reserves the places available for students with disability who attend the extraordinary calls for access to university, until 5% of the reserve quota of the total number of places offered in the degree in question is reached. However, there is still a long way to go, as demonstrated by García-González et al. [6], who identified barriers to higher education for students with disabilities in Spain.

Through higher education, Spanish public universities play a crucial role in training professionals, leading to a more accessible, egalitarian and fair society. They also empower people with disability and promote their talents. The Pablo de Olavide University (UPO) is strongly committed to supporting people with disability and is working towards an increasingly accessible and inclusive university, where universal accessibility is a reality in the classroom—not merely a future goal. In fact, articles 6.4 and 133.1 of the UPO Statutes include these fundamental principles as basic rules of organisation and operation (see Decree 298/2003, of 21 October 2003 [7]). Moreover, in 2019, the UPO Vice-Chancellor's Office for Culture and Social Commitment promoted the Second UPO Accessibility and Inclusion Plan for Functional Diversity [8] to regulate actions aimed at UPO people with functional diversity and to guarantee equal opportunities during their stay at the institution. This plan is inspired by the "Independent Living Movement" which promotes the values of integral formation, autonomy, empowerment and responsibility.

The presence of students with disability and specific educational support needs in university classrooms is a reality today. Since 2020, these students have benefited from a good practice guide [9], directed towards facilitating their access to university and preventing first-year university drop-out. University teaching staff must guarantee accessible and inclusive training, so that the necessary tools to guarantee competence acquisition are available to all students, regardless of their learning difficulties. Unfortunately, this scenario does not always play out, and daily students with some type of disability face varying degrees of academic barriers. The Fifth Study on the Degree of Inclusion of the Spanish University System regarding the Realities of People with Disability [10] examined the ways to improve the inclusion of people with disability at university. It concluded that efforts should be mostly oriented towards the adaptation of resources that allow quality distance education. The reason is that 22% of university students with disability considered that the main way to improve their inclusion at the university was to adapt the resources to study from home. This was followed by 14% of students with disability who referred to the elimination of architectural barriers that still existed for people with mobility difficulties. This lack of resources has an additionally negative impact in the case of the statistics subject, due to the idiosyncrasy of the discipline and the scarce amounts of teaching support materials adapted to functional diversity. Aware of these difficulties, the authors of this work undertook the challenge to create specific audiovisual content on descriptive statistics, adapted to people with hearing and visual impairment.

It has been found, moreover, that providing an inclusive education to students with intellectual disability in higher education is an emerging challenge [11]. This latter study confirmed that, from the students' perspective, university environments could be suitable for students with intellectual disability. The creation of inclusive higher education pro-

grammes should therefore be encouraged [11]. In this sense, the Training Programme for Employment and Autonomous Life of People with Intellectual Disability (FEVIDA) was established at the UPO in 2017. It is a social innovation university programme designed to train 17 young people between the ages of 18 and 29 to improve their possibilities of employment and autonomous life. During the 2021–2022 academic year, these students with disability are and will be receiving 30 credits worth of training in practical, humanistic and professional subjects on the university campus. Extracurricular activities, activities shared with undergraduate and postgraduate students of official studies, as well as other actions that take place within the university community, complete the Own Title. This fifth edition includes an innovative experience aimed at promoting the autonomous life of these students with functional cognitive diversity. They will spend a week of community immersion, staying at the Flora Tristán university residence. Further information on FEVIDA can be found both in [12] in its classic version, and in [13], in its version adapted for easy reading. The authors of this work, convinced of its potential, incorporated the statistics subject of both the Degree in Environmental Sciences and the Dual Degree in Environmental Sciences and History-Geography at UPO in the FEVIDA's fifth edition.

Despite the need of the current society of inclusive higher education experiences that inspire and motivate other university teachers, there is still not enough work addressing this challenge. However, the study of the inclusion of people with disability at university has been developed with more intensity in recent years in different parts of the world, as can be found in [11,14–24]. Unfortunately, this trend is not as exciting if we focus on the field of science in general and statistics in particular. Nevertheless, in Spain emerging groups of researchers work to make the science accessible to all kind of people regardless of their age, social status, functional diversity and academic level [25–27]. In this sense, both the adapted teaching material and the inclusive workshop prepared by the authors are proposals where the processes of knowledge transfer of statistics are aimed at people with disability. The main goal of these inclusive actions is to make statistics accessible to all audiences without regard to their abilities; at the same time students with disability are encouraged to undertake scientific careers and science teachers to make their work more inclusive.

The present paper describes how this inclusive experience unfolded, detailing the adapted teaching support material elaborated, as well as the inclusive activities shared by undergraduate and FEVIDA students. All these activities took place in an environment of awareness, sensitisation and respect regarding disability and functional diversity within the context of a science subject. A total of 43% of the undergraduate students enrolled in the subject participated in a questionnaire to evaluate the integration experience. The questionnaire data and student academic results were statistically analysed. The intended outcome was the promotion, protection and guarantee of the full and equal enjoyment of all human rights and fundamental freedoms of all persons with disability, as well as the promotion of respect for their inherent dignity (article 1 of the Convention on the Rights of Persons with Disabilities [28]).

2. Materials and Methods

During the first semester of the 2021–2022 academic year, we conducted a series of actions in the statistical subject of the Degree in Environmental Sciences and Dual Degree in Environmental Sciences and History-Geography at the UPO. These actions were designed to make this subject more accessible to students with disability, to incorporate students with some type of intellectual disability, as well as to sensitise the whole class about an increasingly diverse and enriching social environment. Therefore, they sought to achieve the four primary objectives detailed below.

 Improve the acquisition of competences in the statistics subject of all students, including those with some type of disability, with a special emphasis on hearing and visual impairment. To reach this objective, new audiovisual material specially developed for statistics was created, which was in turn adapted to people with hearing and visual disability, that facilitated studying the subject and acquiring competencies. In other words, not only the learning process of statistics students was eased, but also everyone was included, since some of the students' special needs were addressed.

- 2. Promote the inclusion of people with disability at the UPO, especially students with intellectual disability, raising awareness of the need to integrate people with some type of disability into the classroom. A theoretical-practical descriptive statistics session was held, adapted to their special needs in an inclusive way, where the rest of the classmates participated very actively. This allows the UPO graduates to promote respect for diversity, equity and equality when they integrate into the labour market in the future.
- Make university students aware of diversity, including people with disability, generating experiences that bring this reality closer to students, raising awareness about the importance of educational inclusion, promoting an education based on respect, solidarity and tolerance.
- 4. Analyse whether the students' average grade improves when the subject is inclusive. If the mark in the different tests increases when the subject is inclusive, the belief that all the actors benefit from an inclusive subject will be strengthened.

To reach these objectives, the authors applied statistical techniques as Pearson's chisquare test for the independence of two qualitative variables, or Fisher's exact test when convenient; likewise, to study the difference between the means, the normality test, the non-parametric technique of Mann-Whitney's and median difference test were studied.

The previous objectives were based on the assumption that both teachers and students prepared materials and activities in a clear and simple language, using short sentences, including images or pictograms, avoiding negative sentences and metaphors and providing easy reading. According to the Une 153101:2018 EX standard, easy reading is a tool that allows us to make texts and documents understandable for everyone. The easy reading method collects a set of guidelines and recommendations related to the writing of a text, the design/layout of documents, and the validation of their comprehensibility, in order to make information accessible to people with reading comprehension difficulties. The easy-to-read materials prepared by the teachers were not only designed for people with disability, but also for people with low educational levels, social problems, language difficulties and immigrants.

Specific activities were developed for the three first objectives and are detailed in the following subsections; meanwhile the fourth objective is addressed in Section 3.

2.1. Creation of Teaching Support Materials Adapted to Functional Diversity

This subsection explains how the authors of this work developed training pills on basic concepts of the subject that were accessible to students with hearing and visual impairment. The purpose of these resources was to work on the competencies required in the descriptive statistics block of the statistics subject taught in the Degree in Environmental Sciences and Dual Degree in Environmental Sciences and History-Geography at the UPO. The teaching or training pills were each of the small parts into which the content was divided in the "microlearning" methodology, which has great advantages in both formal and non-formal education. These teaching pills are much easier to assimilate and are taught through different means, among which the use of new technologies stands out.

The audiovisual material was made accessible to people with hearing disability through subtitling and translation into sign language. Subtitling consists of simultaneously displaying written text and images on the screen. The problem with sign language translation is that each country or region has a different language. This makes it extremely difficult to generate universally accessible audiovisual material. While subtitling is valid for all Spanish-speaking countries, the translation into sign language needs to be adapted to each country or region. The audiovisual material developed by the teachers combined subtitling with sign language translation by an official interpreter. In this way, a person with both hearing disability and low reading abilities could enjoy the statistics subject contents on equal terms with the rest of the students. Teachers had to collaborate with the sign language interpreter to develop their work adequately, for example, by advancing the common technical vocabulary used in order to anticipate the translation into sign language.

In addition, the material developed contained contextualised examples in environments the students are familiar with, which facilitates learning. Likewise, the material addressed descriptive statistics contents very useful in real life, so it also helped people with some mathematical knowledge and a desire to learn.

Specifically, the teachers elaborated the three following teaching pills adapted to the needs of people with hearing disability:

- The arithmetic mean [29], which is the most widely used descriptive statistic in real life. The arithmetic mean or average, or simply the mean or average if the context is clear, is the sum of a set of results divided by the total number of results. By delving into its advantages, disadvantages and different forms of calculation, we better understand its potential. Specifically, the average is theoretically defined, simple examples are given, the pros and cons of its use are analysed and some of its properties are shown. A simple illustration of the weighted arithmetic mean was also defined and presented. Finally, two exercises were given to practice the average for discrete and continuous statistical variables, as well as another exercise to work on the weighted arithmetic mean.
- The median [30], which is a broadly recognised descriptive centralisation statistic, not influenced by atypical values. The median is the midpoint at which no more than half the values are above or below, that is, the median is the value separating the higher half from the lower half of a data sample. The material began with the definition of the median, followed by an explanation of its calculus when the data are not grouped in frequency tables (distinguishing the cases when the total is an odd or even number). An explanation was also given on how the median is calculated for data grouped into frequency tables, differentiating between discrete and continuous statistical variables. The advantages and disadvantages of this descriptive central tendency statistic together with several examples were shown as well.
- The mode [31], which is a widely used descriptive centralisation statistic. The mode is
 the most frequent value found in a set of data values. This learning capsule begins
 with the definition of the mode and continues with the explanation of its calculus,
 distinguishing whether the statistical variable is qualitative, discrete quantitative, or
 continuous quantitative. Various examples of the three types of variables were presented, including one where the distribution was bimodal. The strengths, weaknesses
 and properties of this central tendency descriptive statistic were also analysed.

The three training pills contained a support annex that included the complete transcription of the videos and materials, as well as the exact minute of the recording, in order to support all the students in their learning. Furthermore, these annexes are being translated for the ONCE Scientific Editor (EDICO), so that they will be accessible to blind or severely visually impaired students. EDICO is the first Accessible Mathematics Editor that allows blind or severely visually impaired students to follow maths, statistics, physics or chemistry classes, thanks to the transcription into Braille code. That is why these teaching support materials are also adapted to students with visual disability.

It is important to note that students who are blind or have low vision use a software, such as JAWS (Job Access With Speech), which reads everything on the computer screen aloud. However, these screen reader software packages have the limitation of not reading formulas, scientific signs or mathematical signography, an aspect to be covered by EDICO. Consequently, there is a gap in the teaching support material adapted to students with visual impairment in the field of STEM (science, technology, engineering and mathematics) subjects, a gap that increases as one ascends through the different educational levels.

These new teaching support materials were designed to allow students with disability to undertake the studies in which they were enrolled under the same conditions as the rest of the students. They aimed at promoting full inclusion, in accordance with equal opportunity and non-discrimination principles. The goal was thus that all students reach their maximum personal, intellectual, social and emotional development, and that the objectives established in the current law were attained.

Finally, these audiovisual documents have been included in UpoTV, which is UPO's multimedia repository. In this way, these statistics training pills are open to all people who require them, regardless of whether they have disability or whether they are part of the UPO university community or not. This aspect is an important milestone: indeed, such specific, as well as transversal statistical material is almost impossible to find on the internet (whether free of charge or not). To guarantee the technical quality of these training pills, the teachers sought the advice of the UPO Library Multimedia Laboratory.

2.2. Inclusion of Students with Intellectual Disability in University Classrooms

This subsection describes the two-hour workshop on the statistics subject of the Degree in Environmental Sciences and Dual Degree in Environmental Sciences and History-Geography. It took place during the first semester of the 2021–2022 academic year and was adapted to the special needs of people with intellectual disability in the FEVIDA programme. All those enrolled in the subject in both degrees, i.e., a total of 121 students, participated in the FEVIDA workshop as assistants.

At the beginning of the semester, teachers presented the project to the undergraduate and FEVIDA students, providing a Google form in which they registered in the workshop in order to actively participate. Finally, 45 undergraduate students and 11 FEVIDA students signed up to the workshop, out of the 121 students enrolled in the subject and 17 enrolled in FEVIDA. The activity was very well received by the undergraduate students: 1 out of every 4 students in the subject became actively involved; and of these, 73% answered the assessment survey. This allowed us to gather their specific opinions about the activity, and make future improvements, by enhancing the detected strengths and correcting the weaknesses.

Throughout the semester, several face-to-face and virtual sessions took place between the teachers and the undergraduate students via the Blackboard Collaborate Ultra platform. These meetings addressed not only the topics to focus on, but also the way in which to work on them: enhancing FEVIDA students' autonomy, encouraging their active participation, avoiding infantilisation, etc. Following a number of meetings that were held outside class hours, the teachers and the undergraduate students agreed on a series of topics for the inclusive statistics workshop, detailed below.

- Topic 1: arithmetic mean. At the beginning of the workshop, the undergraduate students presented some theoretical notions. FEVIDA students learnt to work out the average, for example, the average mark of several exams. Undergraduate students prepared a variety of highly practical exercises that required the active participation of FEVIDA students. Two concrete examples of mean drawbacks were also presented: one to explain that if the variable is discrete, the average might not belong to the variable's set of values, and another to show that the arithmetic mean is sensitive to extreme values.
- Topic 2: household economy. Undergraduate students prepared some activities where FEVIDA students worked on monthly/annual expense concepts, instilling the monthly savings necessary to meet annual expenses such as car insurance. Various cases of real spending were introduced where FEVIDA students had to manage several house expenses based on a salary or pension.
- Topic 3: discounts. FEVIDA students learned how to calculate the cost of discounted items. Specifically, undergraduate students explained how to compute the price of an item with a 25% or 30% discount, what it meant when a second item was 50% or 70% off, how to calculate the cost of products in the case of promotions such as 2×1 or 3×2 , etc.
- Topic 4: games of chance. Undergraduate students introduced the concept of probability calculus based on the Laplace rule. The probabilities were calculated for games of chance with dice, cards and the Christmas lottery. This latter example was chosen because the lottery is highly traditional and popular in Spain.

- Topic 5: statistics in social networks. FEVIDA students learnt to interpret the data provided by social networks on "likes", statistics of photos associated with questions, etc. The practical activity consisted of taking a photo of everyone at the beginning of the class, uploading it to social networks, and analysing the percentage of responses to the questions posed at the end of the class. To carry out this activity, all participants gave their consent for the publication or dissemination of the image.
- Topic 6: Kahoot. Kahoot is known to be a highly motivating method for evaluating and reviewing content. The quiz contained questions related to the contents explained and worked on during the class. FEVIDA students could use the mobile calculator for the resolution. By default, Kahoot allocates 20 s to each question, but undergraduate students adjusted it to 240 s to avoid stressing the FEVIDA students.

The inclusive statistics workshop was eminently practical, dynamic and participatory for both undergraduate and FEVIDA students. In fact, the undergraduate students explained some basic statistics concepts to their FEVIDA classmates adopting a pragmatic perspective. The teachers had a supervising role during the workshop. The result of this experience was twofold: making students with intellectual disability more familiar with statistics, and bringing undergraduate students closer to the realities of people with disability.

2.3. Disability Awareness

Disability awareness, which is part of an inclusive culture, promotes positive attitudes of appreciation, respect, solidarity and tolerance towards disability. Carrying out disability awareness activities encourages coexistence, develops empathy and favours the acceptance of people with disability.

In addition to the preparation of the FEVIDA workshop described in the previous section, examples, exercises and problems in the field of disability were frequently used during the statistics classes. Official statistics on people with disability were also analysed. In this transversal way, the teachers tried, through positive and inclusive language, to make their students gain awareness of the daily difficulties faced by people with disability and their strength at overcoming them. Through these activities, the teachers also encouraged their students to appreciate the abilities and achievements of people with disability. Naturally, in all the developed activities, exercises and examples about statistics were also combined within the context of environmental problems, as well as the conservation and protection of the environment and nature, as is typical of the degree in which the subject is taught.

These activities are essential. Indeed, according to the Universia Foundation [10], less than 31% of students recognise having received sensitisation sessions on people with disability. In addition to promoting awareness about disability across classes, universities should promote training courses on disability aimed at undergraduate students, and they should be recognised as free credits.

3. Results

The evaluation process was similar to any proper academic and educational activity. On the one hand, the undergraduate students were evaluated as specified in the subject's General and Specific Teaching Guides. In addition, undergraduate students prepared assessment exercises for FEVIDA students to perform independently at home. They were delivered to teachers' lockers within a set deadline. The responsibility of delivering homework on time to the teachers' lockers implied that FEVIDA students acquired typical undergraduate student routines, which helped them to feel autonomous and empowered. Moreover, the teaching staff was subject to the student satisfaction surveys about the teaching activity, that is, a teaching staff evaluation process that is institutionally conducted by the UPO.

At the end of the first semester of the 2021–2022 academic year, students enrolled in the statistics subject of the Degree in Environmental Sciences and Dual Degree in Environmental Sciences and History-Geography at the UPO were invited to take a survey through Google Forms, available in the virtual classroom. The survey was anonymous and participants were assured that their responses would not be identifiable. The questionnaire data were analysed using the statistical package IBM SPSS Statistics 27.0 (IBM, New York, NY, USA). The survey questions that undergraduate students asked about the activity can be found in the Appendix A.

Of note, undergraduate students evaluated this experience through Google Forms. Specifically, 52 undergraduate students filled out the survey: 42% men and 58% women. All students who actively participated in the workshop showed the highest levels of satisfaction with the activity.

Although it was female students who were primarily involved in preparing the FEVIDA workshop activities (almost 2 out of 3 participants in the preparation of exercises were female), no statistical relationship was found between gender and active participation in the workshop activities (Fisher *p*-value = 0.569). Indeed, the workshop was well received by male and female students alike. In fact, the workshop participants' degree of satisfaction was notable, both that of the attendees and that of the students who organised the activities: 100% indicated being satisfied with the workshop experience and 9 out of 10 students said they were very satisfied or completely satisfied.

However, the first, unfortunate surprising fact was that 63% of undergraduate students said that their level of knowledge of the different types of disability was insufficient. Only 2% claimed that they had very good knowledge of the different types of disability. That is why different educational institutions at all levels should promote integration activities for students with functional diversity, since the benefits generated for all parties are remarkable. It would also be interesting for educational institutions to promote training on the different types of disability, in order to promote general social awareness of the limitations and challenges faced by these individuals on a daily basis.

On the other hand, the survey showed that more than 70% of the students did not doubt the sensitivity of people with disability and only 25% believed that they became difficult to deal with over time. Only a minority thought that older people, due to their age, were people with disability. Indeed, only 13% expressed this opinion, and a large majority of the students surveyed, 62%, were aware that in the future they could develop a disability.

The study also indicated that 81% of those surveyed considered that people with disability were not sufficiently protected in Spain. The perception that not enough was being done was palpable after analysing the results. Almost 9 out of 10 students affirmed that insufficient actions were being taken by the authorities to further integrate people with disability, and 87% said that more money should be spent to remove physical barriers in order to make life easier for people with physical disability.

The inequality suffered by people with disability was confirmed once again: 3 out of 4 students surveyed perceived the lack of real-life opportunities for people with disability who shared the same skills, aptitudes or diplomas as persons without disability.

At this point, we asked whether activities such as the FEVIDA workshop served to raise awareness about the different daily life realities in a society. The answer was resoundingly positive, given that 100% of the students confirmed that dealing with people with disability had helped them to understand another social reality, to gain awareness of the situation of others, and to be conscious that some people in society needed help. Small day-to-day problems were diminished after dealing with people with disability, since after having followed the FEVIDA workshop, 4 out of 5 students believed the problems of their daily life seemed less important. Almost all students became aware that dealing with people with disability had made them to improve as people, and that a functional diversity team enhances everyone's learning.

Having exhaustively analysed the impact of an active participation in the workshop on undergraduate students, we immediately realised that the training activity was a success. Not only had it served to improve the competence worked on in the group work subject, but also all the students affirmed that it taught them to manage time correctly, to help each other, and to progress in the group's objectives through the appropriate execution of the role of each.

The workshop undoubtedly made undergraduate students reflect on disability, although this was not the first time they had done so. A total of 83% stated that they had already reflected on the matter on other occasions.

Finally, undergraduate students were asked about the teachers' underlying motives in setting up such a workshop, involving students with intellectual disability in the statistics subject. The answers obtained are illustrated in Figure 1.





The purpose most referred to by the students attending the workshop was "making the subject more inclusive", followed by "showing the difficulties that people with disability face every day". The undergraduate students also frequently selected the motives of promoting the rights of people with disability in state universities and broadening the understanding of other realities. Finally, it should be noted that among all the purposes, the least mentioned was "giving a positive image of people with disability", which confirms the students' degree of understanding of the purpose of the FEVIDA workshop.

The workshop coordinators highlighted the following elements: the resounding success of the activity; the excellent reflection exercise that took place in the classroom with all the actors involved; and the need to work towards a more inclusive university, where functional diversity is a reality, leading to enriching synergies between students and teachers in the classroom.

In addition, upon data analysis, it is very important to highlight that disability is not a gender issue. No statistical evidence was found relating gender to any of the opinions and perceptions expressed by undergraduate students. In the study's pertinent questions, the answers of male and female students were greatly similar in all cases.

In order to cover the proposed fourth objective of detecting a change or improvement in the average grade when the subject has been inclusive, the average grades of the three exams carried out throughout the semester were analysed to evaluate the students of the Degree in Environmental Sciences. To this goal, the data of two academic years were taken, the 2020–2021 course where there were no students with disability in the classroom, and the 2021–2022 course where the subject was inclusive and the FEVIDA workshop was held. The data showed a significant increase in two of the three exams carried out in the course with the inclusive subject, as shown in Table 1. Both the data distribution and the median of the analysed distributions were statistically different in two of the three evaluation exams. Specifically, the average mark of the first exam increased by almost 13%, from 5.59 to 6.3 points. On the other hand, the mark of the knowledge exam using the SPSS statistical programme further increased the average score in the inclusive course, almost 40%, from 3.64 to 5.05 points.

	Course 2020–2021		Inclusive Course 2021–2022		<i>p</i> -Value
	Mean	Standard Deviation	Mean	Standard Deviation	
First exam score	5.59	1.82	6.30	2.26	0.006 */0.039 **
Second exam score	4.56	2.21	4.55	2.65	NS */NS **
Third exam score (SPSS)	3.64	2.10	5.05	2.63	0.002 */0.047 **
Number of students		115		110	

Table 1. Scores of the statistics exams of the Degree in Environmental Sciences.

* Non-parametric Mann-Whitney U test. ** Test for median difference. NS = Not significant.

Last but not least, upon completion of the experience, the teachers also reflected on the need for university teachers to receive specialised training on how to make their subjects and classes more inclusive. That is why we concluded that the training unit of any public university should promote educational training and innovation processes for teachers in the field of functional diversity, since faculty members require training and information on inclusive education to meet the needs of students with disability [16]. In addition, this reflection also stemmed from the low percentage of teachers-only 14%-who had received training and guidance on the care of students with disability [10]. According to the report "Persons with Disability in the Andalusian University System" [32], Andalusian university teachers lack training on how to tutor students with disability. Nearly half the tutors (44%) stated they had not received any specific training; some tutors had received instructions or recommendations from the relevant university bodies or from the teaching team (17%), of the others, none (13%), and only 12% had taken training courses. Another motive is the fact that 57% of students with disability perceived that their teachers did not know or did not have up-to-date knowledge of their needs [10]. The competent authorities should take note of this, especially as this is not the first time such requirements have become apparent [20].

4. Discussion

The materials prepared by the teachers included content and activities adapted to the students' diversity. Specifically, the materials were especially adapted to students with hearing and visual impairment. They also followed easy reading standards and were thus accessible to people with low educational levels, with social problems, language difficulties and immigrants. In addition, it was easier for students to control and manage their learning itinerary based on their ability or previous knowledge. Likewise, the material produced was all the more relevant since there is little or no public offer of contents that attend to diversity. Therefore, the didactic materials elaborated are very useful, because they responded to a weakness detected in the teaching-learning of the statistics subject addressed to people with disability. Furthermore, this lack of adapted teaching support materials—that affects all fields of knowledge—is more significant in the field of statistics. Indeed, the absence of adapted material must be added to the necessary prior knowledge of mathematics, as well as the level of abstraction required to understand the subject. There is still, however, a long way to go to achieve true equal opportunities for people with disability in education.

In addition, the inclusion of FEVIDA students in a statistics class of undergraduate students marked a before and after in the teachers' personal and professional lives. The experience made it possible to promote the inclusion of people with intellectual disability at the UPO, raising awareness of the need to integrate people with some type of disability in the classroom. Undergraduate students gained awareness of a diverse society in which all people, regardless of their condition, should have the same opportunities to face life, with all its obstacles and goals. At the same time, FEVIDA students understood that they were an active component of the citizenry, and their talents contributed to a more homogeneous development and progress of the community. In this sense, the research confirms that inclusive education presents high levels of teacher and undergraduate student satisfaction, in addition to improving the academic performance of students. The academic

managers of the FEVIDA programme also reported high scores for students with intellectual disability's satisfaction. This fact corroborates the results obtained in [11]; that is, university environments can be suitable for the education of students with intellectual disability and, therefore, the creation of inclusive higher education programmes should be encouraged. That is why the authors suggest generalizing and replicating the successful experiences at the UPO regarding the inclusion of students with disability in the educational system, as recommended in [24].

On the other hand, authorities should develop laws and strategies encouraging university faculties to elaborate inclusive materials in their subjects and to incorporate students with disability in their classes. In the development of these educational policies, the competent authorities cannot lose sight of the fact that the existence of adapted teaching support material decreases as we advance through the different educational levels, being more pressing in the case of STEM subjects. This involves an added difficulty for students with disability studying at university in the scope of STEM. In fact, students with disability tend to study degrees related to engineering and architecture and sciences to a lesser extent than the general population; in contrast, they study in a much higher percentage of arts and humanities degrees, with lower labour insertion rates than other branches [33]. The authors have shown that it is possible to approach and explain the statistics subject to students with disability in a valuable way for this group, banishing prejudices shared by teachers, students and society in general about the distrust in the democratisation of science [27]. Therefore, educational authorities should promote inclusive science teaching (science without barriers) and encourage teachers to develop teaching support materials accessible to all.

Faculty members should be aware of the need to guarantee quality and inclusive educational materials; in particular, statistics subject resources, for all people with disability. Obviously, these adapted teaching support materials should be openly accessible and free of charge. It is necessary to raise awareness and sensitise the university community (teaching and research staff, administration and services staff, and students) to promote the inclusion of students with disability at university. This educational recommendation working to raise the awareness was already suggested in the inclusion of deaf and hard-of-hearing students at university [24].

Finally, university teachers should undergo training in inclusive education for people with disability. These courses should meet diverse objectives and cover varying degrees of complexity and themes. Universities must have teachers with inclusive profiles who are interested in continuous training [17]. In fact, González-Castellano et al. [17] showed that the more accessible and inclusive the universities are, the more continuous training teachers have and therefore the more interest they have in continuing training. Ortiz Colón et al. [19] also confirm the need for training in special needs processes to enable university teaching staff to participate in an inclusive model and reveal the teaching staff do not consider themselves sufficiently prepared to provide an educational response to students with a disability. In a similar sense, Carballo et al. [16] state inclusive faculty members recommend that other teachers become informed about disability, develop a good and close relationship with their students and value their abilities, not their limitations.

5. Conclusions

Among the most relevant conclusions that we can draw from the research work carried out is the lack of previous inclusive experiences and adapted teaching material in mathematics and statistics subjects in the university context. The deficit of integration experiences is much more significant in the case of students with intellectual disability. All this entails a gap in the protocols and procedures to support university professors in their teaching work in these subjects. In this sense, the authors propose activities that have proven successfully with students with mild intellectual disabilities. In addition, several audiovisual materials adapted to various disabilities have been proposed by the authors to understand the different theoretical concepts of statistics.

This contrasts with the fact that the proliferation of free educational videos of any statistical content has grown exponentially over the past decade. Students today have at their disposal a large amount of audiovisual material to review the contents explained in class. However, it is almost impossible to find material adapted to the needs of people with disability. This evidence, together with the fact that the statistics subject is taught in the first year of many university degrees, led us to focus on this line of research and to pursue the publication of statistics subject has been made more accessible to everyone, and we progressed towards a more egalitarian university, one that is aware of the enrichment that comes with student diversity.

The initiative exposed above aims to raise awareness in the educational community generally, and the university community in particular. It sought to respond to the lack of educational materials adapted to people with disability, especially in the area of the STEM subjects. We suggest furthermore that the National Agency for Quality Assessment and Accreditation (ANECA)—responsible for the evaluation process of various university teaching staff members—includes specific assessment criteria on the actions and measures it promotes to develop the inclusion and equal opportunities of university students with disability. That is, the educational authorities should positively value university teachers who develop adapted resources and conduct inclusion and awareness activities in their classes. Under these conditions, university teachers would be further incentivised to make their subjects more inclusive.

The experience was highly gratifying for all the people involved: the FEVIDA and undergraduate students and the teachers. In fact, teachers are hoping to repeat the experience in the coming academic year and to incorporate improvements, new topics of interest, etc. Teachers are convinced that these types of activities allow everyone to grow, in the broadest sense of the term. Students with disabilities represent a positive professional challenge for the faculty, leading to new learnings and greater professional satisfaction [22]. Therefore, we will pursue our efforts towards the achievement of a more accessible and inclusive university. University lecturers should, however, receive training on inclusive education [11]. In addition, students with disabilities emphasised the need to provide training and/or information on disability [21], a reflection we mentioned in Section 3.

The role of public Spanish universities is a key in promoting the talents of people with disability. Therefore, they must continue to develop policies to achieve real and effective equality of opportunities for people with disability in the higher education environment. This goal reflects a societal need but it also corresponds to the fourth objective to transform our world; that is, to guarantee inclusive, equitable and quality education and promote lifelong learning opportunities for all, featuring in the 2030 Agenda for Sustainable Development and approved by the United Nations Organization in 2015 (United Nations Organization) [1].

6. Limitations and Future Research

Among the limitations that the authors have found in their research we highlight the lack of previous inclusive experiences as well as adapted material for the statistics subject in the university context. These drawbacks, together with the scarcity of teacher training in students with special needs or disability, have made authors self-taught in the field of inclusive statistics, marking a new path that is expected to serve as a model for other university teachers.

Moreover, the authors lack the academic authorities' recognition for implementing inclusive activities as well as preparing material adapted to students with disability, because this type of experience involves a large amount of invisible work by teachers.

This study also includes some limitations that entail challenges for future works. The research has been carried out in a specific subject and university context, so it would be necessary to conduct a comparative analysis of subjects of the same nature in different branches of knowledge and universities. At the same time, it would be interesting to

accomplish studies differentiating according to the type of student disability. Furthermore, to develop a more inclusive higher education, the authors have decided as future research to create a workshop on statistics in the Degree in Computer Engineering in Information Systems, adapted to people with disability, where the use of information and communications technology is the main actor to promote the acquisition of basic statistical concepts. The realisation of this inclusive workshop will involve, among other aspects, the analysis of the accessibility of different computer applications for mobile devices and tablets, differentiating according to the type of disability or special need.

Author Contributions: Conceptualization, N.A.-L. and P.M.-N.; Data curation, N.A.-L. and P.M.-N.; Formal analysis, N.A.-L. and P.M.-N.; Investigation, N.A.-L. and P.M.-N.; Methodology, N.A.-L. and P.M.-N.; Project administration, N.A.-L. and P.M.-N.; Resources, N.A.-L. and P.M.-N.; Software, N.A.-L. and P.M.-N.; Supervision, N.A.-L. and P.M.-N.; Validation, N.A.-L. and P.M.-N.; Visualization, N.A.-L. and P.M.-N.; Writing—original draft, N.A.-L. and P.M.-N. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Institutional Review Board Statement: Ethical review and approval were waived for this study because it was non-interventional. Confidentiality was maintained by responses being completely anonymous and only aggregated data are presented.

Informed Consent Statement: Informed consent was obtained from all subjects involved in the study.

Data Availability Statement: The data presented in this study are available on request from the corresponding author. The data are not publicly available due to privacy or ethical restrictions.

Acknowledgments: The authors are very grateful to the reviewers for their valuable and constructive comments.

Conflicts of Interest: The authors declare no conflict of interest.

Appendix A

Survey questions undergraduate students

- 1. Have you actively participated in the FEVIDA workshop organised in the subject of statistics of the Degree in Environmental Sciences and the Dual Degree in Environmental Sciences and History-Geography?
 - Yes, I have taken part in a working group.
 - No, I only attended the workshop.
 - I have not participated in the preparation of the activities nor have I attended the workshop.
- 2. Please specify whether you are male/female.
- 3. How satisfied are you with the FEVIDA workshop (1/2/3/4/5), where 1 is "not at all satisfied" and 5 is "very satisfied".
- 4. Evaluation of cooperative teamwork (needs improvement/good/very good):
 - Have we all learned?
 - Did we use our time efficiently?
 - Did we finish the job on time?
 - Have we helped each other?
 - Have we made progress in our group goals?
 - Have we each fulfilled our mission?
 - Did everyone correctly exercise their individual role?
- 5. How much do you know about the different types of disability?
 - I know almost nothing.
 - I do not know enough.
 - I know quite a lot.
 - I know a lot.

- 6. Do you think that people with disability are sufficiently protected in Spain? Yes/No.
- Please read the following statements carefully and tell us how much you agree or disagree (strongly disagree/quite disagree/quite agree/totally agree):
 - My life is easier than that of a person with disability.
 - People with disability are very sensitive.
 - Older people are people with disability.
 - People with disability become difficult to deal with.
 - I may also become a person with a disability.
 - More actions should be taken to further integrate people with disability into society.
 - More money should be invested in bringing down the physical barriers that make life more difficult for people with physical disability.
- 8. What opportunities do you believe people with disability have to get a job, training or promotion compared to people without any type of disability and with the same skills, aptitudes or diplomas? Fewer opportunities/The same opportunities/More opportunities.
- 9. Do you think that dealing with people with disability can help you to ... ? Yes/No.
 - To understand another social reality.
 - To be more aware of other peoples' situations.
 - To give less importance to small problems.
 - To understand that some people need help.
 - To improve as a person.
- 10. Did your participation in this project make you reflect seriously for the first time on the reality of people with disability? Yes/No.
- 11. Please indicate the purposes of this project (multiple answer).
 - To improve the understanding of issues related to disability.
 - To promote the rights of people with disability.
 - To give a more positive image of people with disability.
 - To show the difficulties faced by people with disability in their daily lives.
 - To make the statistics subject more inclusive.
 - Other.

References

- 1. United Nations Organization. Sustainable Development Goals. Available online: https://www.un.org/sustainabledevelopment/ (accessed on 29 June 2022).
- La Universidad Española en Cifras 2017/2018. CRUE Universidades Españolas. Available online: https://www.crue.org/wpcontent/uploads/2020/02/UEC-1718_FINAL_DIGITAL.pdf (accessed on 29 June 2022).
- INSTRUMENTO de Ratificación de la Convención Sobre los Derechos de las Personas con Discapacidad, Hecho en Nueva York el 13 de Diciembre de 2006. BOE núm. 96, de 21 de Abril de 2008. Available online: https://www.boe.es/boe/dias/2008/04/21 /pdfs/A20648-20659.pdf (accessed on 29 June 2022).
- 4. Ministerio de Educación, Cultura y Deporte. BOE núm. 138, de 7 de Junio de 2014. Real Decreto 412/2014, de 6 de Junio, por el que se Establece la Normativa Básica de los Procedimientos de Admisión a las Enseñanzas Universitarias Oficiales de Grado. Available online: https://www.boe.es/buscar/pdf/2014/BOE-A-2014-6008-consolidado.pdf (accessed on 29 June 2022).
- 5. Ministerio de Universidades. BOE núm. 233, de 29 de Septiembre de 2021. Real Decreto 822/2021, de 28 de Septiembre, por el que se Establece la Organización de las Enseñanzas Universitarias y del Procedimiento de Aseguramiento de su Calidad. Available online: https://www.boe.es/boe/dias/2021/09/29/pdfs/BOE-A-2021-15781.pdf (accessed on 29 June 2022).
- 6. García-González, J.M.; Gutiérrez Gómez-Calcerrada, S.; Solera Hernández, E.; Ríos-Aguilar, S. Barriers in higher education: Perceptions and discourse analysis of students with disabilities in Spain. *Disabil. Soc.* **2021**, *36*, 579–595. [CrossRef]
- Comunidad Autónoma de Andalucía. BOE núm. 306, de 23 de Diciembre de 2003. Decreto 298/2003, de 21 de Octubre, por el que se Aprueban los Estatutos de la Universidad Pablo de Olavide, de Sevilla. Available online: https://www.boe.es/boe/dias/ 2003/12/23/pdfs/A45772-45810.pdf (accessed on 29 June 2022).
- II Plan de Accesibilidad e Inclusión de la Diversidad Funcional, Universidad Pablo de Olavide, 2019–2022. Available online: https://www.upo.es/cms1/export/sites/upo/upsc/diversidad-funcional/documentos/II-Plan-de-Accesibilidad-e-Inclusion-de-la-Diversidad-Funcional-en-la-UPO.pdf (accessed on 29 June 2022).

- RED SAPDU. Guía de Buenas Prácticas para la Transición, el Acceso y la Acogida del Alumnado con Necesidad Específica de Apoyo Educativo en los Estudios Universitarios. Madrid. Fundación ONCE. Con la Colaboración de CRUE-Asuntos Estudiantiles. 2020. Available online: https://www.crue.org/wp-content/uploads/2021/05/GUI%CC%81A-SAPDU_DIGITAL.pdf (accessed on 29 June 2022).
- Fundación Universia. Universidad y Discapacidad. V Estudio sobre el Grado de Inclusión del Sistema Universitario Español Respecto de la Realidad de las Personas con Discapacidad. 2021. Available online: https://www.cermi.es/sites/default/files/ docs/colecciones/V_Estudio_Universidad_y_Discapacidad_2019_20.pdf (accessed on 29 June 2022).
- Rodríguez Herrero, P.; Izuzquiza Gasset, D.; Cabrera García, A. Inclusive education at a Spanish University: The voice of students with intellectual disability. *Disabil. Soc.* 2021, 36, 376–398. [CrossRef]
- Título Propio: Diploma de Extensión Universitaria en Formación para el Empleo y la Vida Autónoma y de las Personas con Discapacidad Intelectual. Universidad Pablo de Olavide. Available online: https://www.upo.es/postgrado/Diploma-de-Extension-Universitaria-Formacion-para-el-empleo-y-la-vida-autonoma-y-de-las-personas-con-discapacidad-intelectual/ (accessed on 29 June 2022).
- FEVIDA. Universidad Pablo de Olavide. Available online: https://www.upo.es/fevida-lectura-facil/ (accessed on 29 June 2022).
- Alhumaid, M.M.; Khoo, S.; Bastos, T. Self-Efficacy of Pre-Service Physical Education Teachers toward Inclusion in Saudi Arabia. Sustainability 2020, 12, 3898. [CrossRef]
- Aust, R. Disability in Higher Education: Explanations and Legitimisation from Teachers at Leipzig University. Soc. Incl. 2018, 6, 125–136. [CrossRef]
- Carballo, R.; Aguirre, A.; López-Gavira, R. Social and Juridical Sciences faculty members' experiences in Spain: What to do to develop an inclusive pedagogy. *Disabil. Soc.* 2021, in press. [CrossRef]
- González-Castellano, N.; Cordón-Pozo, E.; Pueyo-Villa, S.; Colmenero-Ruiz, M.J. Higher Education Teachers' Training in Attention to SEN Students: Testing a Mediation Model. *Sustainability* 2021, 13, 4908. [CrossRef]
- Muyor-Rodríguez, J.; Fuentes-Gutiérrez, V.; De la Fuente-Robles, Y.M.; Amezcua-Aguilar, T. Inclusive University Education in Bolivia: The Actors and Their Discourses. Sustainability 2021, 13, 10818. [CrossRef]
- Ortiz Colón, A.M.; Agreda Montoro, M.; Colmenero Ruiz, M.J. Toward Inclusive Higher Education in a Global Context. Sustainability 2018, 10, 2670. [CrossRef]
- Osborne, T. Not Lazy, Not Faking: Teaching and Learning Experiences of University Students with Disabilities. Disabil. Soc. 2019, 34, 228–232. [CrossRef]
- Polo Sánchez, M.T.; Aparicio Puerta, M. Perceptions of university students with disabilities in Spain: Ideas and beliefs about attitudes towards their inclusion. *Disabil. Soc.* 2021, in press. [CrossRef]
- Sandoval, M.; Morgado, B.; Doménech, A. University students with disabilities in Spain: Faculty beliefs, practices and support in providing reasonable adjustments. *Disabil. Soc.* 2021, 36, 730–749. [CrossRef]
- Shevlin, M.; Kenny, M.; Mcneela, E. Participation in Higher Education for Students with Disabilities: An Irish Perspective. *Disabil.* Soc. 2004, 19, 15–30. [CrossRef]
- Yousif, N.B.A.; Yousef, E.M.; Abdelrahman, R.M. The Social and Psychological Effects of Inclusive Education of Persons with Hearing Disability in Society: A Field Study at the Disability Resource Centre (Sharjah University). Sustainability 2021, 13, 12823. [CrossRef]
- Capeáns Garrido, R.; Refojo Seronero, C.; López García, C.; Camacho Bolós, P. Towards Inclusive Science Communication: Reflections and Successful Actions; Fundación Española para la Ciencia y la Tecnología: Coruña, Spain, 2022. Available online: https://www.fecyt.es/ es/publicacion/hacia-una-comunicacion-inclusiva-de-la-ciencia-reflexiones-y-acciones-de-exito (accessed on 29 June 2022).
- Gibaja, J.F.; Higuera, S.; Remolins, G.; Mozota, M.; Guzmán, V.; Nieto, A.; Oms, X.; Palma, O.; Cubas, M.; González, A.; et al. *Ciencia Inclusiva, Ciencia Sin Barreras. Memòria Viva: Publicación del proyecto Patrimoni del Programa de Extensión Universitaria;* Universitat Jaume I de Castelló: Castellón de la Plana, Spain, 2019; Volume 11, pp. 41–50. Available online: https://issuu.com/ peu2009/docs/uji_memoriaviva11_2019_cas (accessed on 29 June 2022).
- Mozota Holgueras, M.; Gibaja, J.F. Una Ciencia Democrática es una Ciencia Inclusiva. Conversation 2021. Available online: https://theconversation.com/una-ciencia-democratica-es-una-ciencia-inclusiva-162685 (accessed on 29 June 2022).
- Convention on the Rights of Persons with Disabilities. Available online: https://www.un.org/disabilities/documents/ convention/convoptprot-e.pdf (accessed on 29 June 2022).
- Aquino Llinares, N.; Moreno Navarro, P. Estadística para Tod@s: Adaptación a la Diversidad Funcional. La Media Aritmética. 2021. Available online: https://upotv.upo.es/video/6144556dabe3c6fd1b8b456a (accessed on 5 July 2022).
- Aquino Llinares, N.; Moreno Navarro, P. Estadística para Tod@s: Adaptación a la Diversidad Funcional. La Mediana. 2021. Available online: https://upotv.upo.es/video/616d32e5abe3c6d4348b456b (accessed on 5 July 2022).
- Aquino Llinares, N.; Moreno Navarro, P. Estadística para Tod@s: Adaptación a la Diversidad Funcional. La Moda. 2021. Available online: https://upotv.upo.es/video/6149f4eeabe3c654618b4567 (accessed on 5 July 2022).
- 32. González-Badía Fraga, J.; Cazorla Martín, B.; Gutiérrez Hidalgo, N. Las Personas con Discapacidad en el Sistema Universitario Andaluz. Informe de Situación. Consejería de Igualdad, Políticas Sociales y Conciliación de la Junta de Andalucía. 2022. Available online: https://www.juntadeandalucia.es/organismos/igualdadpoliticassocialesyconciliacion/areas/discapacidad/ publicaciones.html (accessed on 29 June 2022).
- 33. Universidad Nacional de Educación a Distancia. Estudio sobre la Empleabilidad de los Estudiantes y Titulados Universitarios con Discapacidad y sus Necesidades Formativas y de Aprendizaje para una Inclusión en Empleos de Calida. 2015. Available online: https://www.uned.es/universidad/inicio/institucional/unidis/publicaciones.html (accessed on 29 June 2022).



Article

Multinomial Cross-Sectional Regression Models to Estimate and Predict the Determinants of Academic Performance: The Case of Auditor Accountant of the Pontifical Catholic University of Valparaíso

Hanns de la Fuente-Mella ¹, Ricardo Campos-Espinoza ², Nelson Lay-Raby ³,*, Omar Lamelés-Corvalán ³, Mario Pino-Moya ² and Reynier Ramírez-Molina ⁴

- ¹ Instituto de Estadística, Facultad de Ciencias, Pontificia Universidad Católica de Valparaíso, Valparaíso 2340031, Chile; hanns.delafuente@pucv.cl
- ² Escuela de Comercio, Facultad de Ciencias Económicas y Administrativas, Pontificia Universidad Católica de Valparaíso, Valparaíso 2340025, Chile; ricardo.campos.e@pucv.cl (R.C.-E.); mario.pino@pucv.cl (M.P.-M.)
- ³ Facultad de Educación y Ciencias Sociales, Universidad Andres Bello, Viña del Mar 2531015, Chile; o.lameles@gmail.com
- ⁴ Departamento Ciencias Empresariales, Universidad de la Costa, Barranquilla 080002, Colombia; rramirez13@cuc.edu.co
- * Correspondence: nelson.lay@unab.cl

Abstract: The debate on the primary cross-curricular skills or fundamental competencies that must be improved in higher education has increased in the last few years. This is especially important in the new distant learning environments, which bring new challenges to the educational process. Econometric models have been designed to explain the students' academic performance, which has been measured using their qualifications average, the number of failed subjects, passed subjects, and withdrawn subjects, and the level of progress, among other indicators, to try to understand the influence of variables such as students' self-esteem, reading comprehension, English proficiency level, and performance in a mathematics-related subject on the students of accountant auditor program from Pontificia Universidad Católica de Valparaiso. Students were asked to fill in a questionnaire to collect data on the psychological and pedagogical variables, while the socio-economic and sociodemographic data were collected from the university. The results have shown that the most significant variables in the development level of this skill type are socio-demographic and socio-economic characteristics. Some of the psychological and pedagogical variables that have, to a lesser degree, some influences are self-regulation in the learning process and the self-perception of anxiety levels. Lastly, some recommendations to intervene in the students' learning process are presented with the objective of achieving a higher level of development in this type of competences.

Keywords: multinomial logistic regression; academic performance; econometric models

1. Introduction

1.1. Academic Performance in Chile and the World

Addressing a study variable such as the academic performance of students in these times becomes a challenge without falling into value judgments or subjectivities due to the social impact that this topic causes in the life of a human being and, as such, in the university student specifically. In addition, different life situations have a notorious impact on their academic training, on the development of their skills, and how these are ted at work. All this marks differentiating qualities that will make human talent authentic and distinctive, which is a sign that generates value to its environment through its contributions or actions, reducing risks that alter the productive life of its services. According to the research carried out by Véliz, Dorner, and Sandoval [1], the results indicate that one of the variables that must

Citation: de la Fuente-Mella, H.; Campos-Espinoza, R.; Lay-Raby, N.; Lamelés-Corvalán, O.; Pino-Moya, M.; Ramírez-Molina, R. Multinomial Cross-Sectional Regression Models to Estimate and Predict the Determinants of Academic Performance: The Case of Auditor Accountant of the Pontifical Catholic University of Valparaíso. *Sustainability* 2022, 14, 9232. https:// doi.org/10.3390/su14159232

Academic Editor: Larry J. Grabau

Received: 9 July 2022 Accepted: 26 July 2022 Published: 28 July 2022

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). be worked on in university life is academic self-efficacy, which would contribute to favoring academic performance and the processes of adaptation to higher education.

In this sense, academic performance today is affected by environmental, social, and economic factors [2] that undermine the effectiveness of sustainability and sustainability in the global world [3,4], altering the raison d'être of educational organizations in search of learning assurance and of generic and technical skills that help students to exercise their professional role in the market with ethical and quality indicators. People are immersed in a bubble wherein they recreates their lives, overloaded with information, stimulated by external and internal factors, and exposed to constant changes, which has caused the best of the universities in Latin America to revolve around new proposals for flexible academic programs [5–7] and their to curricula adapt to the requirements of the environment. In this approach, the center of everything is learning that develops the student's skills for their daily work; that is, it prepares the student for life, with a humanizing, recursive, and holistic approach, which allows the student to obtain tools to put in practice in their personal and work area [8–10].

To the above, we must add the effects of environmental deterioration and biological damage, which have caused universities in countries such as Argentina [11,12], Mexico [13,14], Brazil [15–17], and Chile [18,19] to generate new models for the creation of educational organizations, new teaching practices have been implemented that maximize the academic performance of the students, and learning spaces (laboratories) have been recreated that allow them to experience new forms of learning and knowledge where constructivism and constructionism have value have been put into practice. At the same time, these new practices promote initiatives for the internationalization of the curriculum, promoting global cultural experiences, and new learning styles are generated, impacting pedagogical models. In contrast, new communication tools and digital platforms are being used that generate new connectivity alternatives, aiming at the freedom of education and the free transit of knowledge. For Ramírez, Villalobos, Lay, and Herrera [20], the media allow the appropriation of knowledge.

Due to social problems in Chile, academic performance has undergone dizzying changes, where young talent has had to leave their classes to join the labor area informally, seeking to improve their quality of life. Due to these notable cases, it becomes essential to obtain practical skills through training that allows students to opt for job improvements, which later are rewarded in personal and family well-being. The 21st century will offer unprecedented resources for the circulation and storage of information and communication. It will pose a double demand on education that, at first glance, may seem almost contradictory: education must massively and efficiently transmit an increasing volume of evolutionary theoretical and technical knowledge, adapted to the cognitive civilization, because they are the bases of the competencies of the future [21].

The purpose of the research is to identify and measure the impact of the variables affecting the increase in the academic performance of the students of the accounting auditor program of the Pontificia Universidad Católica de Valparaíso to assess the finding that allows conclusions to be drawn about the study variable, so that contributions can be generated for the study for the advancement of science, technology, and innovation. Thus, in this research, our questions are how variables such as reading comprehension, level of English, academic self-esteem, general self-esteem, theoretical value, theoretical value, nutrition, student entry, math test, self-updating, graduation, middle school score, religious value, and exercise affect the academic performance through the design of an econometric model that will allow for the measuring and quantifying of said effect.

Some previous studies have examined how to understand and overcome learning difficulties. Most researchers analyze their research data using multiple regression analyses but do not consider learning difficulties as the dependent variable in qualitative data [10,22,23]. Therefore, this article presents a further analysis of previous studies. Multinomial logistic regression (MLR) analysis is a suitable model to solve this problem. That is because the students' tests (on their perceptions) are answers based on alternatives (categorical or qualitative). Additionally, the students' characteristics are nominal and ordinal data. In contrast, other studies have used MLR with different objectives, such as [24–27]. However, the approach to the problem faced is not the same as that of those investigations.

This paper is organized as follows. Section 1 presents the introduction, which includes a literature review. The materials and methods are presented in Section 2. The results are shown in Section 3. Section 4 presents the discussion and conclusions of this research.

1.2. Competencies in Higher Education

Over time, the evolution of institutions has been glimpsed in terms of strengthening the skills of their students. In previous centuries, all those physical skills and strength were required due to the jobs that demanded an effort on the part of the student. Man, however, with various industrial revolutions, has changed this. Currently, organizations focus their attention on developing technical skills [28–30]. This is due to the need for humanized capital in the new global era, with essential skills for executing administrative and operational processes according to the corresponding soft areas. The educational institutions that innovate their curriculum [31–33] and bet on the training of students by challenges have been oriented to the solution of problems that the productive industry sector demands, with a holistic vision, creativity, and innovation, for the generation of value and competitiveness against the challenges of the environment, forming comprehensive leaders with skills that enhance the work of companies.

This demand has been linked to the progress of information technologies, expanding access to tools, knowledge, and networks for the expansion of meaningful learning that brings students closer to natural spaces that allow them to make decisions and be guided by what they do. This can be seen in the virtualization of multiple processes and training, work, and interaction activities through simulators, leading the participant to develop skills by making logical, rational, and analytical decisions that strengthen their work. All of this originates innovative strategies that promote the recovery and reactivation of the nation's economy and contribute to a sustainable development that minimizes the devastating effects of the environment, being focused on the responses of the human being as a strategic partner that generates value for companies following their skills acquired at the university.

Competencies in higher education refer to the set of skills, aptitudes, attitudes, and knowledge an individual has. For Fuentes, Moreno-Murcia, Rincón-Tellez, and Silva-García [34], competencies are the grouping of behaviors, measurable and perceptible, that can be evaluated in a person. Hence, it is necessary to determine which skills are required to execute activities. According to Carrión-Martínez, Fernández-Martínez, Pérez-Fuentes, and Gázquez-Linares [35], the outstanding segments for the grouping of skills and abilities are guided by cognitive traits (logical thinking and theoretical knowledge), followed by cognitive traits, social traits (communicative skills), and psychosocial traits (motivational skills). When referring to generic skills, we speak of those universal skills that must be acquired regardless of the study area. These allow for the carrying out of simple but essential processes for developing and improving local economies.

Research carried out by Echeverría-Ramírez and Mazzitelli [36], Preciado-Serrano, González, Colunga-Rodríguez, Vázquez-Colunga, Esparza-Zamora, Vázquez-Juárez, and Obando-Changuán [37] and Sagredo, Etchepare, Mendizabal, and Wilson [38] make it clear that generic competencies in higher education emphasize general attributes and abilities for the development of activities such as analytical and critical thinking, conflict resolution, communication skills, and teamwork. However, these are situational, therefore it is imperative to analyze the current and future situations to master the required skills. In contrast, technical skills in the educational context are those skills and abilities obtained by carrying out activities or, failing that, by experience. Technical or complex skills are all those qualities and knowledge acquired through experience in developing activities. It should be noted that these technical skills are more specific to the sector or type of industry to work.

Nowadays, social, economic, political, environmental, and biological factors have notoriously influenced human beings to reinvent themselves, adapting to new environmental changes. These changes force us to enhance the competence of the living being through dynamic capabilities to systemically solve the problems that human talent itself recreates, made up of opportunities and threats. For this reason, the articulation of competencies in higher education is relevant, where students are trained with abilities, knowledge, and skills that allow them to respond to this new global era that is increasingly volatile and competitive. According to the investigations of Jiménez, Pérez, and Gómez [39] and León, Mendoza, and Gilar [40], there are social, economic, and environmental factors in universities that have a notorious impact on student performance. For this reason, it is evident to generate innovative initiatives that enhance the learner's skills to mitigate the adverse effects of these factors that threaten the development of learners' skills.

According to Sukier, Ramírez, Parra, Martínez, Fernández, and Lay [41], companies must change their models of strategic management of human talent, adapting to the dynamic challenges of the environment. These models are increasingly unstable, aggressive, and turbulent, condemning people to reinvent themselves and rely on tools to consolidate their goals and improve their skills. In recent months, the multiple effects caused by the COVID-19 virus have been evidenced. At current times, at the global level, the slowdown in economic growth, reduction in income, loss of profitability and economic insolvency, high level of unemployment, and the disappearance of companies have been observed [42–44]. At the same time, society has feared physical contact, which characterizes human beings. The Chilean economic context has not been different from the global one [45].

In the following section, we describe how the measurement of certain competencies were used for the development of this research.

1.3. Meauserement of Competencies

The instruments used to measure the skills of higher education students were the following six tests: Psychosocial Motivations Scale (MPS) by J. L. Fernández (1987) [46], Gordon Allport's values study test (1968–1972) [47], Lifestyle Profile (PEPS-I) by Nola Pender (1996) [48], Wonderlic by Eldon F. Wonderlic (1936) [49], ICT Questionnaire in Personal Training (REATIC) [50], Stanley Coopersmith's self-esteem inventory (1967) [51]. Below is a summary of each of the tests used.

Psychosocial Motivations Scale (MPS) by J. L. Fernández (1987) [46]. The instrument was designed to measure the motivation system of individuals, published by TEA Ediciones in 1987. The scale seeks to appreciate the individual's motivational system's differential structure and functional dynamics (emotional, cognitive, and situational) and predict the person's future behavior. Said evaluation is of six factors and five components of psychosocial motivations in the work context. The scale factors are acceptance and social integration, social recognition, self-concept, self-development, power, security. Its reliability coefficient is 0.53–0.83. It consists of three parts, with a total of 173 items.

In its first part, a true or false answer is given to each item for the levels of Activation, Expectation, and Execution. In its second part and, in turn, the third, it is answered using an evaluation scale of the elements chosen by the individual in the Incentives and Satisfaction of each item. The rating ranges from 1 ("some"), 2 ("normal"), 3 ("a lot"), 4 ("very much").

Allport Values. Gordon Allport's values study test (1968–1972) [47] applied his study of values inspired by the types of men of Stranger (1964), and defines the ways of life of man in six interests or basic motives in personality: the social, political, aesthetic, economic, theoretical, and religious. It consists of two parts; in the first one, there are statements or questions with two alternative answers. In the second part, situations are presented, followed by four possible attitudes or responses, where the individual must classify these responses in the order of personal preference, noting the score according to the degree of preference. Finally, the test classifies each interest style as high, average, and low.

Lifestyle Profile (PEPS-I) by Nola Pender (1996) [48]. The questionnaire quantitatively measures the level of the individual's lifestyle. It is made up of 48 Likert-type items, which are subdivided into the dimensions of nutrition (6 items), exercise (5 items), responsibility in health (10 items), stress management (7 items), interpersonal support (7 items), and self-updating

(13 items). This questionnaire asks the respondent to identify how often they have practiced each statement in the last 30 days and to circle the answer that best reflects their current lifestyle. Each item has a minimum score of 1 ("never") and a maximum of 4 ("always"), adding a minimum overall score of 48 and a maximum of 192. The scores define high lifestyles (132–192), medium (108–131), and low (48–107) according to 0.75 standard deviations above or below the mean score.

Wonderlic test by Eldon F. Wonderlic [49]. This instrument has 50 logical reasoning and problem-solving questions distributed from least to significant difficulty. It is made up of three subtests: vocabulary, arithmetic reasoning, and spatial reasoning. It presents a correlational validity between teaching and academic achievement between r = 0.30 and r = 0.80, with a reliability coefficient between 0.73 and 0.95. The Wonderlic Personnel Test (WPT) is a general intelligence test used in personnel selection and vocational guidance. It has an execution time of 12 min for the resolution of the test. The scores define each student's work and academic and training potential, distributed in six ranges below 10 points, between 10 and 15 points, between 16 and 20 points, between 21 and 23 points, between 24 and 27 points, and above 28 points. It also has a correction table according to the age of the person evaluated.

ICT Questionnaire in Personal Training (REATIC), the De Moya questionnaire [50], comprises 60 items through which the greater or lesser degree of knowledge, use, and existing assessment of information technologies can be established and communicated in university education. Therefore, the purpose of the instrument is to determine the conditions of a student's management and the positive assessment of the implementation of ICT in their academic and personal life. The test is divided into four subgroups: I know (item 1 to 14), I use (15 to 28), I consider ICT (item 29 to 44), and I use ICT according to learning style (item 45 to 60).

Stanley Coopersmith's self-esteem inventory in 1967 [51]. This instrument aims to identify the evaluation that a person makes and maintains about himself, indicating the degree to which he considers himself capable, competent, and successful. It has 25 items on a dichotomous scale (yes–no), separated into three areas, i.e., General Self, Social, and Family, establishing the following intervals for a maximum score of 100 points. Low level of self-esteem (0 to 24 points), Medium–low level of self-esteem (25 to 49 points), Medium–high level (50 to 74 points), and High-level (75 to 100 points). Its reliability is 0.79, based on the Tarazona study.

2. Materials and Methods

The analysis is carried out for the career of Accountant Auditor of the Pontifical Catholic University of Valparaíso (PUCV), Chile. It is a sample of 82 students of both genders, of which 34% are male and 66% are female. This research considered different ways of measuring the endogenous variable: average of grades; the number of approved courses; the number of failed courses; the number of retirees; the degree of advancement; the minimum qualification; the maximum qualification; and the standard deviation of qualification [52]. Both the data for the endogenous and exogenous variables of the proposed models are obtained from each of the surveys. Given the above, and following the proposed objectives, the analysis aims to establish the determining factors of the training competencies of the students of the Accountant Auditor at the PUCV, Chile, and their impact on the students' academic performance.

The instruments used to collect the information come from the different tests that the students of the Accountant Auditor career take upon entering the university. This is how students develop various tests, where the general self-esteem tests corresponding to the Coopersmith self-esteem inventory, academic self-esteem, reading comprehension, and a diagnostic test of English [51–55] stand out, considering the leveling given by the Council of the Common European Framework of Reference for the Languages (CEFR). For this research, the results obtained by each student participating in the study have been collected
directly from the Directorate of Institutional Analysis and Strategic Development (DAIDE) of the Pontifical Catholic University of Valparaíso.

Both types of data have been collected (among 2016 and 2021) by different platforms and systems in the PUCV and a unique code has been used to identify the data of each student. A protocol was established so that only one person knew the link between said code and the individualization of each student. This relationship data was not disclosed to anyone else. All this is due to the procedures for the private use of said data, respecting the individualization of the students. Similarly, when the students were surveyed, they had been informed of how the data would be used, protecting their identity. These tests were only been carried out if the participant granted their approval and consent.

The first stage of the study consists of carrying out an exploratory analysis of the variables under study to understand the behavior of each of the variables involved in the research. Subsequently, cross-sectional econometric models are used to establish the determining factors of the fundamental training competencies of the Auditor Accountant students. Cross-sectional data, or a cross-section of a study population, in econometrics, is a type of data collected by observing many subjects (such as individuals, firms, countries, or regions) at one point or period of time. The analysis might also have no regard for differences in time. Cross-sectional data can be used in cross-sectional regression, which is a regression analysis of cross-sectional data, which was conducted in this research. Analysis of cross-sectional data usually consists of comparing the differences among selected subjects, through which the assumptions associated with the problem of spherical disturbances are reviewed and evaluated and the other tests are proposed for its validation [56], performing the necessary transformations for its validation without affecting the economic nature of the model [57].

3. Results

This section presents the results of the econometric estimation of the mixed model, defined as an estimation of the academic performance of an accountant auditor student at the Pontifical Catholic University of Valparaíso, Chile.

Among the research results, determining factors were evidenced in the academic performance measured by the indicator of courses approved by students, which correspond to the scores obtained in evaluations carried out upon entering university, which sought to identify psychological elements such as levels of general and academic, or educational, self-esteem, such as performance in the English language diagnostic test or reading comprehension, that allow us to approach the objective of identifying elements that are related to the development of fundamental training competence in the students under study. Thus, the descriptive statistics of the variables used in the model are shown in Table 1

Table 1. Descriptive analysis variables of the research.

	Ν	Minimum	Maximum	Mean	Std. Deviation
Average of grades	82	4.17	6.62	5.42	0.65
Approved courses	82	0.50	1	0.83	0.14
Failed courses	82	0.00	0.23	0.057	0.076
Retirees courses	82	0.00	0.30	0.087	0.086
Degree of advancement	82	0.105	0.926	0.440	0.300
Minimum qualification	82	1.0	6.3	3.72	1.52
Maximum qualification	82	5.8	7.0	6.7	0.30
Standard deviation of Qualification	82	0.206	1.269	0.793	0.236
Theoretical value	82	1	5	3.93	0.890
Nutrition	82	7	24	15.66	4.054
Exercise	82	5	18	9.34	3.181
General self-esteem	82	28.0	104.0	67.92	24.192
Academic self-esteem	82	4.0	16.0	11.882	3.3802
Reading Comprehension	82	2.0	10.0	6.039	1.9896

N Minimum Maximum Mean Std. Deviation 82 9.0 74.0 28.302 14.8823 Level of English Maths test 82 1.023.013.360 5.3443 Self-updating 82 24 50 38.49 6.784 Religious value 82 13 45 25.86 7.236 Middle school score 82 5.2 6.8 6.070 0.3259

Table 1. Cont.

From the analysis of Table 1, we can see that the ranges of the exogenous variables are wide because they are related to the characterization of a heterogeneous sample, composed of students who are at different times in the curriculum of the career of the Accountant Auditor. When analyzing the standard deviation presented by some variables, this is confirmed since the high levels of dispersion with respect to their mean reinforce the idea of a heterogeneous sample, which is therefore representative of the population under study. For example, when considering variables such as the level of English of the students in the sample or the results obtained in the general self-esteem test, it is evident that these are characteristic aspects of the students themselves and that they influence the degree of development of the fundamental competencies.

Econometric Models

With the previously exposed variables, a cross-sectional econometric model has been developed. With the above, through the cross-sectional model, the intention is to assess the degree of development of fundamental competencies and their impact on the academic performance of students. The endogenous variables tested in this study are: average of grades; the number of approved courses; the number of failed courses; the number of retirees; the degree of advancement; the minimum qualification; the maximum qualification and the standard deviation of qualification. Moreover, the exogenous variables are: the results in the entrance tests of reading comprehension; level of English; academic self-esteem and general self-esteem. All discrete variables of real scale that constitute the determining factors in the degree of development of fundamental competencies.

To estimate the functional relationship of the model based on Equation (1), we propose a mixed model to explain the response variable Y in terms of a set of explanatory variables. The functional relationship of the model is defined as follows (see Equation (1)):

$$Y_{ij} = \beta_0 + \sum_{s=1}^{S} \beta_s X_{sij} + u_{i0} + e_{ij},$$
(1)

where Y_{ij} corresponds to the dependent variable measured by academic performance from each auditor accountant student *i* over different measures of academic performance *j*; β_0 corresponds to a baseline parameter of the model; and β is a parameter vector that corresponds to the parameters related to variables that affect the academic performance of students *i*. x_{ij} corresponds to the following variables: reading comprehension, level of English, academic self-esteem, general self-esteem, theoretical value, theoretical value, nutrition, student entry, maths test, self-updating, nutrition, student entry, maths test, self-updating, graduation, middle school score, religious value, exercise; from each auditor accountant student *i* over different measures of academic performance *j*. Finally, *i* = auditor accountant students and *j* = different measures of academic performance.

From the analysis of Table 2, we can see that multiple cross-sectional econometric models have been tested for each of the endogenous variables under study. Thus, in the case of the academic performance of students measured by average of grades, the best model is explained by the variables theoretical value and nutrition. In the case of the number of approved courses, the best model is explained by the variables general self-esteem, academic self-esteem, reading comprehension, and level of English; for the model

with endogenous variable failed courses, the best model is explained with the variables theoretical value, student entry, math's test, self-updating; for the measurement of academic performance through the retirees courses variable, the best model is explained by the variables general self-esteem and academic self-esteem; with the model measured through the endogenous variable degree of advancement, the variables that explain are graduation; in the case of measuring academic performance through the minimum qualification variable, the best model is explained by the variables religious value and student entry; for the measurement through maximum qualification of student, the model is explained by the variables exercise and religious value; and, finally, for the academic performance model measured through the standard deviation of qualification variable, the best model is explained by the variables.

Table 2. Results of the estimation of the cross-section model and its goodness of fit.

Average of grades				
	Model 1	Model 2		
R Square	0.321	0.321		
ANOVA (p-value)	0.003	0.000		
Variables	Theoretical value 0.280 (3.369)	Theoretical value 0.294 (4.045)		
		Nutrition -0.047 (-2.931)		
		Approved courses		
	Model 1	Model 2	Model 3	Model 4
R Square	0.301	0.568	0.639	0.708
ANOVA (p-value)	0.004	0.000	0.000	0.000
Variables	General self-esteem -0.003 (-3.218)	General self-esteem -0.003 (-4.715)	General self-esteem -0.003 (-4.946)	General self-esteem 0.003 (5.525)
		Academic self-esteem 0.019 (3.772)	Academic self-esteem 0.023 (4.529)	Academic self-esteem 0.022 (4.628)
			Reading comprehension 0.019 (2.082)	Reading comprehension 0.020 (2.426)
				Level of English -0.002 (-2.230)
		Failed courses		
	Model 1	Model 2	Model 3	Model 4
R Square	0.416	0.573	0.652	0.751
ANOVA (p-value)	0.000	0.000	0.000	0.000
Variables	Theoretical value -0.061 (-4.135)	Theoretical value -0.071 (-5.320)	Theoretical value -0.069 (-5.534)	Theoretical value -0.071 (-6.588)
		Student entry -0.055 (-2.906)	Student entry -0.066 (-3.661)	Student entry -0.066 (-4.218)
			Maths test -0.004 (-2.232)	Maths test -0.005 (-3.160)
				Self-updating -0.004 (-2.891)

 Table 2. Cont.

 Retirees courses

 Model 1
 Model 2

 0.446
 0.554

	Model 1	Model 2			
R Square	0.446	0.554			
ANOVA (p-value)	0.000	0.000			
Variables	General self-esteem 0.002 (4.400)	General self-esteem 0.002 (5.190)			
		Academic self-esteem -0.007 (-2.358)			
	Degree of advancement				
	Model 1				
R Square	0.414				
ANOVA (p-value)	0.000				
Variables	Graduation -0.130 (-4.116)				
Minimum qualification					
	Model 1	Model 2			
R Square	0.314	0.488			
ANOVA (p-value)	0.003	0.000			
Variables	Religious value -1.138 (-3.313)	Religious value -1.410 (-4.428)			
		Student entry 0.779 (2.797)			
		Maximum qualification			
	Model 1	Model 2			
R Square	0.493	0.583			
ANOVA (p-value)	0.000	0.000			
Variables	Exercise -0.065 (-4.834)	Exercise -0.077 (-5.671)			
		Religious value 0.196 (2.225)			
	Standard deviation of qualification				
	Model 1	Model 2			
R Square	0.214	0.356			
ANOVA (p-value)	0.017	0.006			
Variables	Middle school score -0.250 (-2.559)	Middle school score -0.216 (-2.358)			
		Theoretical value -0.083 (-2.252)			

The above information is presented in summary form in Table 3.

In the first model, we can see that the cognitive attitude favorably influences the grade point average, unlike the effect of concern for a healthy diet and a lifestyle that considers the importance of nutritional aspects in students; in the second model, we can see that higher levels of general self-esteem and academic self-esteem positively influence the number

of approved courses, as well as the results obtained in the reading comprehension tests, while knowledge of the English language has a negative effect on the endogenous variable; for the model that considers failed courses as an endogenous variable, the relationship of the exogenous variables with it is negative, since, the higher the cognitive attitude score achieved in the mathematics test and self-updating, the lower the expected number of failed courses obtained by students in the development of the study plan-the same impact generated by considering the type of entry of students to the career of Accountant Auditor; on the model whose endogenous variable is represented by the withdrawn courses, the general self-esteem positively affects this, while the higher the academic self-esteem, the lower the number of courses that a student is expected to be able to withdraw from during their career. For the model whose endogenous variable is defined by the minimum qualification, we note that the development of a religious life has a direct relationship with obtaining lower minimum marks, unlike the effect generated by the type of income. When the endogenous variable is defined by the maximum qualification, the religious value has a positive effect, which his quite the opposite of the effect of that generated by lifestyle when linked to sport. Finally, if we consider the model whose endogenous variable is the standard deviation of the grades, the theoretical value negatively influences this variable, that is, those students with a higher cognitive attitude present lower levels of dispersion in their grades with respect to the mean being more stable their qualifications.

Table 3. Best cross-sectional econometric model per variable.

Endogenous Variable	Explanatory Variables	
Academic performance of students measured (by average of grades)	Theoretical value and nutrition	
Number of approved courses	General self-esteem, academic self-esteem, reading comprehension and level of English	
Failed courses	Theoretical value, student entry, maths test, self-updating	
Academic performance (through the retirees courses variable)	General self-esteem and academic self-esteem	
Degree of advancement	Graduation	
Academic performance (by minimum qualification variable)	Religious value and student entry	
Maximum qualification of student	Exercise and religious value	
Academic performance model	Theoretical value	

Once the analysis of the resulting models has been carried out, we can see that the theoretical value, general self-esteem, and academic self-esteem are the most significant and influential variables when we search the determining factors of the training competencies of the students of the Accountant Auditor of the PUCV, Chile, measured through indicators of academic performance. In addition, it is possible to observe that the effect that these variables have on the endogenous variables is consistent. For example, the theoretical value positively affects the explanation of the performance of students (measured by average of grades), while it negatively affects the explanation of failed courses. Other variables that maintain the same type of consistency are academic self-esteem and religious value. Additionally, the knowledge and skills of the students, measured by theoretical value, maths test, reading comprehension, and self-updating, are consistent with the expected performance, since, to the extent that they are higher, it is expected that the degree of progress and approved courses are also approved.

4. Conclusions and Discussion

According to what was mentioned in the previous section in this study, it was possible to estimate and predict the determinants of academic performance for the students of the Auditor Accountant of the Pontifical Catholic University of Valparaíso career.

The results made it possible to show that the most significant variables in the level of development of this type of skill are related to socio-demographic and socio-economic characteristics. This is somewhat related to the case studied, which is found in the Latin American [58–61] country—in the process of economic development—of Chile, in which such characteristics are heterogeneous in the study population. In this sense, both theoretical and applied studies are clear in indicating how socioeconomic variables influence academic performance in the world [62–65], in Latin America, and in the Chilean case [66–69].

This is quite helpful in improving the material elements in the broader university context and at the national level, and could subsequently improve the acquisition of the skills detailed above.

From the results, it is possible to observe that, in the sample, the self-esteem of the students is one of the strongest and most significant factors in the different models used in this research. Students' prior knowledge level, measured by the grades they obtained before enrolling in the University, is a factor that significantly explains the dispersion of university academic performance. However, what is interesting is that the levels of reading comprehension and mathematics were less significant for the average academic performance. The above could mean that this baseline and prior knowledge only contribute to significant but decreasing baseline development. On the other hand, the self-esteem that students have of themselves is even more relevant.

At another level, the psychological and pedagogical variables had, to a lesser degree, some influence on self-regulation in the learning process and the self-perception of anxiety levels. From the psychology of learning, it has been widely documented how emotional self-regulation is important to obtain academic achievement [70–75]. The same is true for self-perception of anxiety levels [76–78]. Additionally, it can be commented that the measurements of both variables were taken in a confinement context due to the COVID-19 pandemic, thus that variable has been shown with a higher estimate. Coincidentally, both the social characteristics mentioned above and confinement have a large material component at the base of the learning of the students studied.

The identification of factors associated with academic performance can allow university authorities to design policies that improve the academic quality of undergraduate students. In specific terms, the authorities could strengthen university programs and activities that foster confidence in young people and create coaching programs. According to this research, this type of activity could contribute more to improving student learning than reinforcing activities with technical program content.

This research could serve as a starting point for a broader investigation. On the one hand, expanding the number of students analyzed by incorporating other cohorts of graduates, which would also make it possible to carry out various cross-sectional studies and identify possible changes in the predictor variables that could occur. In addition, this research could be developed in other degrees and universities, which would also allow comparative studies. Specifically, according to the data collected in this research, making controlled variations of more heterogeneous samples in the social conditions of the students could provide more significant evidence.

Moreover, it is expected that the conclusions reached by this research can serve as a basis for future analyses and/or modifications of the study plan, considering it essential to become a support for the accompaniment of students in this stage of adaptation, such as, in the first year, introducing students to the opportunities that arise when the person has perseverance in the study, and fostering loyalty to the institution, creating a good image and retention from responsible academic work.

In order to safeguard the learning and social process of young people who are immersed in university, it is imperative to carry out psychosocial interventions from the field of mainly prevention through an induction that facilitates the process of adapting to university; however, when detecting students who may be more likely to have mental health disorders, it is necessary to support them through two aspects, namely counseling or individual therapy and workshops for a targeted group of students at risk. Research indicates that the streams with better results are cognitive, behavioral, and mindfulness interventions, mainly in reducing stress in students; this could be a good path to start an intervention aimed at student assets.

It remains to be seen in future studies in this same population if the same estimates are maintained for the determined variables exposed in the proposed models. It is important for the analysis of this case that the measurement was made at a time of confinement due to the pandemic.

Notwithstanding the foregoing, the proposed models are important to be able to estimate the effects of the study variables on academic performance and a subsequent achievement of educational excellence.

Author Contributions: Data curation, R.C.-E.; Formal analysis, H.d.I.F.-M.; Investigation, N.L.-R.; Software, O.L.-C.; Writing—review & editing, M.P.-M. and R.R.-M. All authors have read and agreed to the published version of the manuscript.

Funding: Research work of H. de la Fuente-Mella was supported by grant Núcleo de Investigación en Data Analytics/VRIEA/PUCV/039.432/2020 from the Vice-Rectory for Research and Advanced Studies of the Pontificia Universidad Católica de Valparaíso, Chile.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: The data used to support the findings of this study are available from the corresponding author upon request.

Conflicts of Interest: The authors declare no conflict of interest.

References

- Véliz, A.; Dorner, A.; Sandoval, S. Relación entre autoconcepto, eficacia académica y rendimiento académico en estudiantes de salud de Puerto Montt, Chile. EDUCADI 2016, 1, 97–109. [CrossRef]
- 2. Mushtaq, I.; Khan, S.N. Factors affecting student's academic performance. Glob. J. Manag. Bus. Res. 2012, 12, 17–22.
- Alamri, M.; Almaiah, M.; Al-Rahmi, W. Social Media Applications Affecting Students' Academic Performance: A Model Developed for Sustainability in Higher Education. Sustainability 2020, 12, 6471. [CrossRef]
- Alam, M.M.; Ahmad, N.; Naveed, Q.N.; Patel, A.; Abohashrh, M.; Khaleel, M.A. E-learning services to archive sustainable learning and academic performance: An empirical study. Sustainability 2021, 13, 2653. [CrossRef]
- Cook, M.L. Politics of Labor Reform in Latin America: Between Flexibility and Rights; The Pennsylvania State University Press: University Park, PA, USA, 2010.
- Lamarra, N.F. Higher Education, Quality Evaluation and Accreditation in Latin America and MERCOSUR. Eur. J. Educ. 2003, 38, 253–269. [CrossRef]
- 7. Rama, C. University virtualisation in Latin America. Int. J. Educ. Technol. High. Educ. 2014, 11, 32–41. [CrossRef]
- Montané, A.; Llanes, J.; Calduch, I.; Hervás, G.; Méndez-Ulrich, J.L.; Muñoz, J. The social dimension in Higher Education. Design and implementation of an instrument for student analytics in a Latin American context. In *Culture, Citizenship, Participation. Comparative Perspectives from Latin America on Inclusive Education*, 1st ed.; Bon, A., Pini, M., Akkermans, H., Eds.; Pangea: Amsterdam, The Netherlands, 2019; Volume 2, pp. 133–158.
- Lent, R.W.; Taveira, M.D.C.; Figuera, P.; Dorio, I.; Faria, S.; Gonçalves, A.M. Test of the Social Cognitive Model of Well-Being in Spanish College Students. J. Career Assess. 2016, 25, 135–143. [CrossRef]
- Llanes, J.; Méndez-Ulrich, J.L.; Montané, A. Motivación y satisfacción académica de los estudiantes de Educación: Una visión internacional. Educ. XX1 2021, 24, 45–68.
- 11. Fracchia, E.; Mesquita, L.; Quiroga, L. Business Groups in Argentina, 1st ed.; Oxford University Press: New York, NY, USA, 2010.
- Rozada, M.G.; Menendez, A. Public university in Argentina: Subsidizing the rich? *Econ. Educ. Rev.* 2002, *21*, 341–351. [CrossRef]
 Perales Franco, C.; McCowan, T. Rewiring higher education for the Sustainable Development Goals: The case of the In-tercultural University of Veracruz, Mexico. *High. Educ.* 2021, *81*, 69–88. [CrossRef]
- 14. Gill, C.C. Education in a Changing Mexico, 1st ed.; US Office of Education, Institute of International Studies: Washington, DC, USA, 1969.
- Laus, S.P.; Morosini, M.C. Higher Education in Brazil. In *Higher Education in Latin America: The International Dimension*, 1st ed.; De Wit, H., Jarmillo, C., Gacel-Ávila, J., Knight, J., Eds.; The World Bank: Washington, DC, USA, 2005; Volume 638, p. 111.
- 16. Schwartzman, S. Brazil: Opportunity and crisis in higher education. High. Educ. 1988, 17, 99–119. [CrossRef]
- Berchin, I.I.; dos Santos Grando, V.; Marcon, G.A.; Corseuil, L.; de Andrade, J.B.S.O. Strategies to promote sustainability in higher education institutions: A case study of a federal institute of higher education in Brazil. *Int. J. Sustain. High. Educ.* 2017, 18, 1018–1038. [CrossRef]

- 18. Brunner, J.J. Higher Education in Chile from 1980 to 1990. Eur. J. Educ. 1993, 28, 71. [CrossRef]
- 19. Bernasconi, A. Private Higher Education in Chile: The New Exceptionalism. Int. High. Educ. 2003, 32, 18–19. [CrossRef]
- Ramírez, R.I.; Villalobos, J.V.; Lay, N.D.; Herrera, B.A. Medios de comunicación para la apropiación del conocimiento en instituciones educativas. *Inf. Tecnológica* 2021, 32, 27–38. [CrossRef]
- Delors, J. Los cuatro pilares de la educación. In La Educación Encierra un Tesoro. Informe a la UNESCO de la Comisión Internacional Sobre la Educación Para el Siglo XXI; Santillana/UNESCO: Madrid, Spain, 1996; Available online: https://uom.uib.cat/digitalAssets/ 221/221918_9.pdf (accessed on 8 July 2022).
- Prasetvono, H.; Abdillah, A.; Widiyarto, T.; Srivono, H. Character-based economic learning implementation and teacher's reinforcement on student's affective competence in minimizing hoax. *Cakrawala Pendidik*. 2018, 37, 426–435.
- Soubhi, F.Z.; Lima, L.; Talbi, M.; Knouzi, N.; Touri, B. Learning Difficulties Related of Health Status of Moroccan Students. Procedia Soc. Behav. Sci. 2015, 197, 1507–1511. [CrossRef]
- Awan, K. Experimentation and correlates of electronic nicotine delivery system (electronic cigarettes) among university students— A cross sectional study. Saudi Dent. J. 2016, 28, 91–95. [CrossRef]
- de la Fuente-Mella, H.; Umaña-Hermosilla, B.; Fonseca-Fuentes, M.; Elórtegui-Gómez, C. Multinomial Logistic Regression to Estimate the Financial Education and Financial Knowledge of University Students in Chile. *Information* 2021, 12, 379. [CrossRef]
- Torikka, A.; Kaltiala-Heino, R.; Rimpelä, A.; Marttunen, M.; Luukkaala, T.; Rimpelä, M. Self-reported depression is increasing among socio-economically disadvantaged adolescents—Repeated cross-sectional surveys from Finland from 2000 to 2011. BMC Public Health 2014, 14, 408. [CrossRef]
- Trinidad, M.F.; Pascual, J.L.G.; García, M.R. Perception of caring among nursing students: Results from a cross-sectional survey. Nurse Educ. Today 2019, 83, 104196. [CrossRef] [PubMed]
- Webb, J.; Chaffer, C. The expectation performance gap in accounting education: A review of generic skills development in UK accounting degrees. Account. Educ. 2016, 25, 349–367. [CrossRef]
- Douglas, S.; Gammie, E. An investigation into the development of non-technical skills by undergraduate accounting programmes. Account. Educ. 2019, 28, 304–332. [CrossRef]
- Van Vught, F.A. Governmental Strategies and Innovation in Higher Education. Higher Education Policies Series, 7, 1st ed.; Kingsley: London, UK, 1989.
- Kırkgöz, Y. The challenge of developing and maintaining curriculum innovation at higher education. *Procedia Soc. Behav. Sci.* 2009, 1, 73–78. [CrossRef]
- Kemp, K.K.; Goodchild, F.M. Evaluating a major innovation in higher education: The NCGIA core curriculum in GIS. J. Geogr. High. Educ. 1992, 16, 21–35. [CrossRef]
- Leask, B. Internationalisation, globalisation and curriculum innovation. In *Researching International Pedagogies*, 1st ed.; Hellstén, M., Reid, A., Eds.; Springer Dordrecht: Dordrecht, The Netherlands, 2008; pp. 9–26.
- Fuentes, G.Y.; Moreno-Murcia, L.M.; Rincón-Tellez, D.C.; Silva-Garcia, M.B. Evaluación de las habilidades blandas en la educación superior. Form. Univ. 2021, 14, 49–60. [CrossRef]
- Carrión-Martínez, J.J.; Fernández-Martínez, M.D.M.; Pérez-Fuentes, M.D.C.; Gázquez-Linares, J.J. Specific competencies in social work higher education in the framework of the European higher education area: The perception of future professionals in the Spanish context. *Eur. J. Soc. Work* 2018, 23, 43–55. [CrossRef]
- Echeverría-Ramírez, J.A.; Mazzitelli, C. Estudio de la percepción sobre los factores institucionales que influyen en el rendimiento académico de estudiantes de la Universidad Estatal a Distancia de Costa Rica. Rev. Electrónica Educ. 2021, 25, 326–344. [CrossRef]
- Preciado-Serrano, M.D.L.; Ángel-González, M.; Colunga-Rodríguez, C.; Vázquez-Colunga, J.C.; Esparza-Zamora, M.A.; Vázquez-Juárez, C.L.; Obando-Changuán, M.P. Construction and Validation of the University Academic Performance RAU Scale. *Rev. Iberoam. Diagnóstico Evaluación Avaliação Psicológica (RIDEP)* 2021, 60, 1–20. [CrossRef]
- Sagredo, A.V.; Etchepare, G.C.; Mendizábal, E.A.; Wilson, C.P. Academic performance and its relationship with socioemotional variables in chilean students from vulnerable contexts. *Educ. XXI* 2021, 24, 375–398. [CrossRef]
- Jiménez, M.; Pérez, F.; Gómez, P. Análisis de los factores tecnológicos sobre el rendimiento académico en una universidad pública en la Ciudad de México. Form. Univ. 2020, 13, 255–266. [CrossRef]
- 40. Quinapallo, X.P.L.; Yépez, M.M.M.; Corbi, R.G. Clima de aula y rendimiento académico: Apuntes en torno al contexto universitario. *Rev. Venez. Gerenc.* 2021, 26, 140–156. [CrossRef]
- Sukier, H.; Ramírez Molina, R.I.; Parra, M.; Martínez, K.; Fernández, G.; Lay, N. Strategic Management of Human Talent from a Sustainable Approach. *Rev. Opción* 2020, 36, 929–953.
- 42. Matyushok, V.; Krasavina, V.; Berezin, A.; García, J.S. The global economy in technological transformation conditions: A review of modern trends. *Econ. Res. Ekon. Istraživanja* **2021**, *34*, 1471–1497. [CrossRef]
- Glomado, E.; Zagrebaylova, Y. The Overview of Factors Contributing to the Global Economy Development, 1st ed.; Belarusian State Economic University: Minsk, Belarus, 2022.
- Gómez, L.; Oinas, P.; Wall, R.S. Undercurrents in the world economy: Evolving global investment flows in the South. World Econ. 2021, 45, 1830–1855. [CrossRef]
- Velasco, M.; Mumme, S.P. Environmental capacities in Latin America: A comparison of Brazil, Chile, and Mexico. Soc. Sci. J. 2021, 1–20. [CrossRef]
- 46. Fernández, J. Escala de Motivaciones Psicosociales, 1st ed.; Tea Ediciones: Madrid, Spain, 1987.

- 47. Allport, G.W.; Vernon, P.E.; Lindzey, G. Study of Values, 1st ed.; Houghton Mifflin: Boston, FL, USA, 1960.
- 48. Pender, N.J.; Murdaugh, C.L.; Parsons, M.A. Health Promotion in Nursing Practice, 7th ed.; Pearson: London, UK, 2006.
- 49. Wonderlic, E.F. Wonderlic Personnel Test; Wonderlic, Inc.: Liberty, IL, USA, 2004.
- De Moya, M.V.; Hernández, J.A.; Hernández, J.R.; Cózar, R. Análisis de los estilos de aprendizaje y las tic en la formación personal del alumnado universitario a través del cuestionario reatic. *Rev. De Investig. Educ.* 2009, 29, 137–156.
- 51. Coopersmith, S. The Antecedents of Self-Esteem; W. H. Freeman & Co. Behavioral Science: San Francisco, CA, USA, 1967; p. 15. [CrossRef]
- Moren, C.C.; Augant, K.C.; Labrin, B.C.; de Giorgis, R.S.; de la Fuente-Mella, H.; Fritz, P.; Saavedra, M.V.; Monckton, P.H.; Castelli, L. A quantitative analysis of the identification of personality traits in engineering students and their relation to academic performance. *Stud. High. Educ.* 2019, 45, 1323–1334. [CrossRef]
- Mella, H.D.L.F.; Navarro, M.M.; Manero, C.B.; Iglesias, M.P.; Huenuman, C.G. Análisis de los determinantes del rendimiento académico. El caso de Contador Auditor de la Pontificia Universidad Católica de Valparaíso. *Rev. Estud. Pedagógicos* 2021, 47, 469–482. [CrossRef]
- De La Fuente-Mella, H.; Gutiérrez, C.G.; Crawford, K.; Foschino, G.; Crawford, B.; Soto, R.; De La Barra, C.L.; Caneo, F.C.; Monfroy, E.; Becerra-Rozas, M.; et al. Analysis and Prediction of Engineering Student Behavior and Their Relation to Academic Performance Using Data Analytics Techniques. *Appl. Sci.* 2020, *10*, 7114. [CrossRef]
- Lawson, A.E. The Generality of Hypothetico-Deductive Reasoning: Making Scientific Thinking Explicit. Am. Biol. Teach. 2000, 62, 482–495. [CrossRef]
- 56. Greene, W.H. Análisis Econométrico, 3rd ed.; Prentice Hall: Madrid, Spain, 1999.
- Coughenour, C.; Paz, A.; de la Fuente-Mella, H.; Singh, A. Multinomial logistic regression to estimate and predict perceptions of bicycle and transportation infrastructure in a sprawling metropolitan area. J. Public Health 2016, 38, e401–e408. [CrossRef] [PubMed]
- Flores-Mendoza, C.; Ardila, R.; Gallegos, M.; Reategui-Colareta, N. General Intelligence and Socioeconomic Status as Strong Predictors of Student Performance in Latin American Schools: Evidence from PISA Items. Front. Educ. 2021, 6, 1–16. [CrossRef]
- 59. Murillo, F.J.; Román, M. School infrastructure and resources do matter: Analysis of the incidence of school resources on the performance of Latin American students. Sch. Eff. Sch. Improv. 2011, 22, 29–50. [CrossRef]
- Torres-Roman, J.S.; Cruz-Avila, Y.; Suarez-Osorio, K.; Arce-Huamaní, M.; Menez-Sanchez, A.; Robalo, T.R.A.; Mejia, C.R.; Ruiz, E. Motivation towards medical career choice and academic performance in Latin American medical students: A cross-sectional study. *PLoS ONE* 2018, 13, e0205674. [CrossRef]
- Herman, K.C. Engaging Latino-American Students in School: Evaluating a Causal Model for Understanding and Predicting Academic Achievement. Doctoral Thesis, University of Florida, Gainesville, FL, USA, 1997.
- Rodríguez-Hernández, C.F.; Cascallar, E.; Kyndt, E. Socio-economic status and academic performance in higher education: A systematic review. *Educ. Res. Rev.* 2019, 29, 100305. [CrossRef]
- Sackett, P.R.; Kuncel, N.R.; Arneson, J.J.; Cooper, S.R.; Waters, S.D. Does socioeconomic status explain the relationship between admissions tests and post-secondary academic performance? *Psychol. Bull.* 2009, 135, 1–22. [CrossRef]
- Trusty, J.; Robinson, C.R.; Plata, M.; Ng, K.-M. Effects of Gender, Socioeconomic Status, and Early Academic Performance on Postsecondary Educational Choice. J. Couns. Dev. 2000, 78, 463–472. [CrossRef]
- Brodnick, R.J.; Ree, M.J. A structural model of academic performance, socioeconomic status, and Spearman's g. Educ. Psychol. Meas. 1995, 55, 583–594. [CrossRef]
- 66. Muñoz, P.; Redondo, A. Inequality and academic achievement in Chile. CEPAL Rev. 2013, 109, 99–114. [CrossRef]
- Rossi, M. Factors Affecting Academic Performance of University Evening Students. J. Educ. Hum. Dev. 2017, 6, 96–102. [CrossRef]
 Valladares, M.; Ramírez-Tagle, R.; Muñoz, M.A.; Obregón, A.M. Individual differences in chronotypes associated with academic performance among Chilean University students. Chronobiol. Int. 2017, 35, 578–583. [CrossRef]
- Claro, S.; Paunesku, D.; Dweck, C.S. Growth mindset tempers the effects of poverty on academic achievement. Proc. Natl. Acad. Sci. USA 2016, 113, 8664–8668. [CrossRef]
- Gestsdottir, S.; von Suchodoletz, A.; Wanless, S.B.; Hubert, B.; Guimard, P.; Birgisdottir, F.; Gunzenhauser, C.; McClelland, M. Early Behavioral Self-Regulation, Academic Achievement, and Gender: Longitudinal Findings from France, Germany, and Iceland. *Appl. Dev. Sci.* 2014, *18*, 90–109. [CrossRef]
- 71. Zimmerman, B.J. Self-Regulated Learning and Academic Achievement: An Overview. Educ. Psychol. 1990, 25, 3–17. [CrossRef]
- Nota, L.; Soresi, S.; Zimmerman, B.J. Self-regulation and academic achievement and resilience: A longitudinal study. Int. J. Educ. Res. 2004, 41, 198–215. [CrossRef]
- Zimmerman, B.J. Self-Regulated Learning and Academic Achievement: Theoretical Perspectives, 2nd ed.; Routledge: London, UK, 2001.
 Ruban, L.M.; McCoach, D.B.; McGuire, J.M.; Reis, S.M. The Differential Impact of Academic Self-Regulatory Methods on Academic
- Achievement among University Students with and Without Learning Disabilities. J. Learn. Disabil. 2003, 36, 270–286. [CrossRef]
- Matuga, J.M. Self-regulation, goal orientation, and academic achievement of secondary students in online university courses. J. Educ. Technol. Soc. 2009, 12, 4–11.
- Stringer, R.W.; Heath, N. Academic self-perception and its relationship to academic performance. Can. J. Educ./Rev. Can. De L'éducation 2008, 31, 327–345.

- Chevalier, A.; Gibbons, S.; Thorpe, A.; Snell, M.; Hoskins, S. Student's academic self-perception. *Econ. Educ. Rev.* 2009, 28, 716–727. [CrossRef]
- 78. Leung, P.W.; Kwan, K.S. Parenting styles, motivational orientations, and self-perceived academic competence: A mediational model. *Merrill-Palmer Q.* **1998**, 44, 1–19.

MDPI St. Alban-Anlage 66 4052 Basel Switzerland Tel. +41 61 683 77 34 Fax +41 61 302 89 18 www.mdpi.com

Sustainability Editorial Office E-mail: sustainability@mdpi.com www.mdpi.com/journal/sustainability



MDPI St. Alban-Anlage 66 4052 Basel Switzerland Tel: +41 61 683 77 34

www.mdpi.com



ISBN 978-3-0365-5192-0